

September 25, 2019/Calendar No. 26

C 190366 ZSQ

IN THE MATTER OF an application submitted by Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the New York City Charter for the grant of a special permit pursuant to Section 74-743(a)(2) of the Zoning Resolution to modify:

- 1. the rear yard requirements of Section 23-533 (Required rear yard equivalents for Quality Housing buildings) and Section 35-53 (Modification of Rear Yard Requirements);
- 2. the side yard requirements of Section 35-54 (Special Provisions Applying Adjacent to R1 Through R5 Districts); and
- 3. the height and setback requirements of Section 23-664 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors) and Section 35-654 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors);

in connection with a proposed mixed used development, within a large-scale general development, on property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, Beach 52nd Street, Rockaway Beach Boulevard and Beach 53rd Street (Block 15842, Lot 1 & p/o Lot 100, Block 15843, Lot 1, and Block 15857 Lot 1 & p/o Lot 7), in a C4-4 and C4-3A Districts, Borough of Queens, Community District 14.

This application (C 190366 ZSQ) for a special permit was filed by Peninsula Rockaway Limited Partnership on April 16, 2019. The proposed special permit, in conjunction with related applications, would facilitate a new mixed-use development with affordable housing, commercial, and community facility uses within a large-scale general development (LSGD), on property generally bounded by Beach 50th Street, Beach Channel Drive, Beach 53rd Street, and Rockaway Beach Boulevard (Block 15843, Lot 1, Block 15842, Lot 1 and p/o Lot 100, and Block 15857, Lot 1 and p/o Lot 7) in the Edgemere neighborhood of Queens, Community District 14.

RELATED ACTIONS

In addition to the special permit (C 190366 ZSQ) that is the subject of this report, the proposed project also requires action by the City Planning Commission (CPC) on the following applications, which are being considered concurrently with this application:

C 190251 MMQ	City Map amendment to re-establish a portion of former Beach 52^{nd}
	Street to connect from Rockaway Beach Boulevard to Shore Front
	Parkway
C 190325 ZMQ	Zoning map amendment to eliminate from an R5 district a C1-2
	district, change an R5 district to a C4-4 district, and change a C8-1
	district to a C4-3A district
N 190364 ZRQ	Zoning text amendment to Appendix F to establish a Mandatory
	Inclusionary Housing (MIH) Area, as well as allow a Physical
	Cultural Establishment (PCE) as-of-right within the LSGD
C 190375 ZSQ	Special permit to modify surface area requirements for signage

BACKGROUND

This applicant requests a zoning map amendment, special permit, zoning text amendment, and City Map amendment in order to facilitate the redevelopment of an approximately 9.34-acre vacant site formerly occupied by Peninsula Hospital Center in the Edgemere neighborhood of Queens, Community District 14. The proposal would facilitate the construction of 11 mixed-use buildings ranging in height from eight to 19 stories. The total project would encompass approximately 2,007,000 square feet of development, including approximately 1,858,000 square feet of residential floor area (approximately 2,200 dwelling units), approximately 72,000 square feet of retail space, including an 18,000-square foot supermarket, and approximately 77,000 square feet of community facility space. It would also provide approximately 973 parking spaces and approximately 122,000 square feet of area within new streets, and a range of open spaces

totaling 38,000 square feet in area.

The proposal would provide a total of 550 permanently-affordable housing units through the Mandatory Inclusionary Housing (MIH) program, and all remaining units would be 100 percent affordable through a regulatory agreement with the New York City Department of Housing Preservation and Development (HPD) and/or the New York City Housing Development Corporation (HDC). The proposed project would consist of a total of 2,200 affordable housing units, including 201 Affordable Independent Residences for Seniors (AIRS) and 273 moderate-income units. The development would provide approximately 38,000 square feet of open space, including a designated Publicly Accessible Area (PAA) of approximately 21,000 square feet located in the southeastern portion of the project site. This open space would provide seating, planters, and a children's play area.

The project site is located on the eastern portion of the Rockaway Peninsula on the grounds of the former Peninsula Hospital Center, generally bounded by Beach 50th Street to the east, Beach Channel Drive to the north, Beach 53rd Street to the west, and Rockaway Beach Boulevard to the south (Block 15843, Lot 1, Block 15842, Lot 1 and Block 15857, Lot 1). The rezoning area would also include a portion of Block 15842, Lot 100 and a portion of Block 15857, Lot 7.

The northern portion of the project site is a superblock consisting of Block 15843, Lot 1 and Block 15482, Lot 1. It has a total lot area of 392,296 square feet, with approximately 430 feet of frontage on Beach Channel Drive, 277 feet on Beach 50th Street, 702 feet on Rockaway Beach Boulevard, and 794 feet on Beach 53rd Street. These parcels are located within R5 and R5/C1-2 zoning districts. These lots were previously occupied by the Peninsula Hospital Center, founded in 1908. When it closed in 2012, the hospital contained 173 acute care beds and 200 long-term care beds. The applicant acquired the property in 2016, and the site was cleared of all structures in 2018.

The southern portion of the project site is comprised of Block 15857, Lot 1, and has a

total lot area of 14,407 square feet, not including approximately 3,424 square feet of lot area located within the bed of a mapped street, Beach 52nd Street. It has approximately 115 feet of frontage on Rockaway Boulevard, 85 feet on Shore Front Parkway, and 161 feet on Beach 52nd Street. This parcel was previously improved with a warehouse building that was demolished in spring of 2018. It is located within a C8-1 zoning district.

The surrounding area is developed with a mix of uses, including low- and mid-rise residential buildings, community facilities, transportation and light-industrial uses, regional recreation areas and open space. There is a concentration of public and publicly-subsidized housing and long-term care facilities within the surrounding area. The Ocean Bay Apartments, developed by the New York City Housing Authority (NYCHA) and containing 1,806 residential units, are located directly north and west of the project site with building heights ranging from seven to nine stories. Arverne View apartments are located southwest of the project site. These buildings contain a total of 1,093 dwellings, with the tallest structures rising to a maximum of 19 stories. The four-story Peninsula Nursing and Rehabilitation Center is adjacent to the project site to the east, and Public School 105 is located northeast of the project site. Zoning districts within the surrounding area include R4 and R5 lower-density residential districts, an R6 medium-density district, C1-2 and C2-2 commercial overlays, C4-3A and C4-4 medium density mixed residential and commercial districts, and a C8-1 automotive and light-industrial district.

The area is well-served by public transportation. The project site is located between the Beach 44th Street and Beach 54th Street stations of the Metropolitan Transit Authority (MTA) elevated A train, which runs east- west above the Rockaway Freeway. The area is also served by the QM17 bus providing service to midtown Manhattan and the Q22 bus providing east/west service on the Rockaway Peninsula along Beach Channel Drive and Rockaway Beach Boulevard. Additionally, the NYC Ferry Shuttle stops one block west of the site at Beach Channel Drive and Beach 54th Street, providing access to the ferry landing at Beach Channel Drive and Beach 108th Street. Both Beach Channel Drive and Rockaway Beach Boulevard are wide streets, having widths of 85 feet and 80 feet, respectively.

This portion of the Rockaway Peninsula was included within an area-wide planning initiative sponsored by the Department of City Planning (DCP) in 2008 known as Rockaway Neighborhoods Rezoning (C 080371 ZMQ). This effort aimed to reinforce development patterns and support new moderate-density buildings along wide streets and near transit resources in five Rockaway neighborhoods. More recently, the Downtown Far Rockaway Plan (C 170243 ZMQ), adopted by the City Council on July 10, 2017, was a comprehensive planning, zoning, and redevelopment strategy aimed at supporting Downtown Far Rockaway's growth and vitality by fostering a vibrant mix of uses on vacant and underutilized sites along the area's primary corridors.

There are also two Urban Renewal Areas (URA) in the vicinity. To the east of the project site is the Edgemere URA (C 970410 HUQ) that was adopted in 1997 and amended in 2008. Within the Edgemere URA are two recently completed, 100 percent affordable developments, Beach Green Dunes Phases I and II, fronting Rockaway Beach Boulevard east of Beach 47th Street. Phase I is a seven-story mixed-use building completed in December 2017 that provides 101 affordable dwelling units. Phase II was completed in Spring 2019 and is an eight-story mixed-use building providing 127 affordable housing units. Both phases serve households whose incomes do not exceed 100% of Area Median Income (AMI). The median household income for the surrounding area is approximately \$31,104 for a family of three.

The Arverne Urban Renewal Area, originally approved in 1968 then renewed in 2003 (C 030509 HUQ), encompasses extensive oceanfront lots located directly south of the project site, as well as lots to the southwest and southeast. The portion to the southwest, known as Arverne-by-the-Sea, includes affordable and market-rate residences. Breaking ground on its first phase in 2003, it is in the process of completing its final section. There has been no development elsewhere within the Arverne URA.

The project site sits on a superblock that would be divided into six development blocks (Blocks A, B, C, D, E, and F) by the creation of a privately-maintained street network consisting of two segments. Beach 52nd Street would be extended from Beach Channel

Drive at the northern end of the project site to connect to Rockaway Beach Boulevard at the southern end. Peninsula Way, oriented east-west through the development, would connect Beach 50th and Beach 53rd streets. The proposed street network would be publicly-accessible but privately-maintained and owned, would not be mapped, and would be subject to a public easement. Wider sidewalks have been incorporated into the design in order to enhance the pedestrian experience at the street level.

Block A, bounded by Beach 53rd and Beach 52nd streets between Beach Channel Drive and Peninsula Way, would be comprised of Buildings A1 and A2. Building A1, with frontage on Beach Channel Drive, would provide approximately 181 units, 22,058 square feet of retail at the ground floor level, including a grocery store and 141 parking spaces. A surface parking lot, providing 57 parking spaces and having a landscaped, planted perimeter would be accessed from new curb cuts on Beach Channel Drive and Beach 53rd Street, while the remaining parking spaces would be provided in a garage accessed from Beach 53rd Street. Base height portions would range from 30 feet to 100 feet, and the building would have a maximum building height of 190 feet (18 stories). Building A2, with frontage on Peninsula Way and Beach 52nd Street, would provide approximately 205 units, 5,198 square feet of retail use at the ground floor and second story, and 93 enclosed parking spaces accessed from Beach 53rd Street. Base heights for this building would range from 80 feet to 110 feet, and it would rise to a maximum building height of 200 feet (19 stories).

Block B, located east of Beach 52nd Street between Beach Channel Drive and Peninsula Way, would be comprised of Buildings B1 and B2. Building B1 would provide approximately 230 units, 12,564 square feet of retail use at the ground floor level, and 111 parking spaces. Base heights would range from 70 feet to 100 feet, and the building would have a maximum building height of 160 feet (15 stories). Building B2 would provide approximately 246 units, 7,710 square feet of retail use at the ground floor and second story, and 99 parking spaces accessed from Peninsula Way. The base heights would range from 110 feet to 130 feet, and it would rise to a maximum building height of 190 feet (18 stories).

Block C, located south of Peninsula Way between Beach 53rd and Beach 52nd streets, would be comprised of Buildings C1 and C2. Building C1 would provide approximately 269 units and 102 enclosed parking spaces accessed from Beach 53rd Street. Base heights would range from 80 to 110 feet, and there would be a maximum building height of 150 feet (14 stories). Building C2 would provide approximately 320 units, 10,515 square feet of retail use at the ground floor and second story, and 121 enclosed parking spaces accessed from Beach 53rd Street. Base heights would range from 90 feet to 120 feet. The building would have a maximum height of 200 feet (19 stories).

Block D, located east of Beach 52nd Street between Peninsula Way and Rockaway Beach Boulevard, would be comprised of Buildings D1 and D2. Building D1 would provide approximately 139 units and 5,684 square feet of retail use at the ground floor and second story. The base height would range from 90 feet to 120 feet, and it would have a maximum building height of 150 feet (14 stories). Building D2 would provide approximately 201 AIRS units. The base height would range from 80 feet to 110 feet, and this building would rise to a maximum height of 130 feet (12 stories).

Block E, located west of Beach 50th Street between Peninsula Way and Rockaway Beach Boulevard, would be comprised of Buildings E1 and E2. Building E1 would provide approximately 217 units and 39,824 square feet of retail use at the ground floor and second story. Base heights would range from 110 feet to 140 feet, and it would have a maximum building height of 200 feet (19 stories). Building E2 would provide approximately 123 units and 37,176 square feet of community facility space. Base heights would range from 80 feet to 110 feet, and there would be a maximum building height of 150 feet (14 stories). Block E would also provide 163 parking spaces accessed from Peninsula Way.

Block F, located south of Rockaway Beach Boulevard and east of Beach 52nd Street, would be comprised of one building with approximately 69 units and 22 enclosed parking spaces accessed from Beach 52nd Street. It would have a maximum building height of 90 feet (eight stories).

The proposed project would be constructed over five phases with an anticipated 10-year build out. Phase one of the project would include Buildings A1 and B1 and would be completed by 2022. Phase two would include Buildings A2 and B2 and would be completed by 2024. Phase three of the project would include Buildings C1 and D1 and would be complete by 2026. Phase four would include Buildings C2 and D2 and would be completed by 2028. Phase five of the project would consist of Buildings E1, E2, F1 and the PAA and would be completed by 2029.

The proposed project would facilitate the development of affordable housing units using various city programs including HPD's Extremely Low & Low-Income Affordability, Mix & Match, and Senior Affordable Rental Apartments programs, which would provide a range of incomes from 30 percent to 130 percent AMI. The proposed AMI breakdown would include 435 units at 30 percent AMI, 181 units at 40 percent AMI, 194 units at 50 percent AMI, 494 units at 60 percent AMI, 156 units at 70 percent AMI, 467 units at 80 percent AMI, 59 units at 90 percent AMI, 118 units at 100 percent AMI, 32 units at 110 percent AMI, 32 units at 120 percent AMI, and 32 units at 130 percent AMI.

A designated PAA totaling approximately 21,000 square feet would be constructed between Building Blocks D and E. The provided open space would include a children's play area, seating, planters, and bioswales. This open space area would also provide emergency vehicle access to the adjacent nursing home. There is an existing easement on the project site requiring the emergency vehicle access, and this easement need would be integrated into the design of the PAA to ensure that it would remain unobstructed.

The project site is located within the Coastal Zone Boundary and the 1 percent annual chance flood zone on the Federal Emergency Management Administration Preliminary Flood Insurance Rate Maps. The project site has a Base Flood Elevation of 10 feet. The proposed private street network would provide two sloping, publicly-accessible streets intersecting at the center of the project site with a raised elevation of 14 feet. This portion of the project site would be known as Highpoint Plaza and, because of its elevation, would not be within the floodplain. The proposed project would provide stormwater

detention tanks and internal storm drains within the new internal roadway and discharge into the existing Department of Environmental Protection (DEP) sewer within Beach Channel Drive and a proposed DEP sewer within Rockaway Beach Boulevard. The proposed buildings would be constructed to meet all Appendix G building code requirements and would incorporate various flood resiliency measures as necessary, including wet and dry-flood proofing, and removable flood barriers at lobbies, entrances, and retail locations. The proposed buildings would also be able to be further retrofitted to address projected future flood risks. The proposed planting strategy would provide native and coastal salt tolerant plants with larger soil volumes to promote a stronger and more resilient ecosystem. All lighting fixtures, control boxes, and irrigation control boxes are proposed to be elevated above the flood zone. The proposed storm water strategy would include a bio-retention rain garden, street watersheds using bioswale tree pits, continuous tree pits with large soil volume capacities, flow-through planters and planting pockets.

To facilitate the proposed development, the applicant seeks zoning map and zoning text amendments, special permits and an amendment to the City Map.

Zoning map amendment (C 190325 ZMQ)

The proposed zoning map amendment would change the project area's zoning from existing R5, R5/C1-2, and C8-1 districts to C4-3A and C4-4 districts. The proposed C4-4 district would be mapped over the current superblock containing the northern portion of the project (Block 15843, Lot 1 and Block 15842, Lot 1 and part of Lot 100). The C4-4 district has an R7-2 residential equivalent. The proposed C4-3A district would be mapped over Block 15857, Lot 1 and part of Lot 7 within the project area south of Rockaway Beach Boulevard. The proposed C4-3A zoning district would replace an existing C8-1 district. The C4-3A district has an R6A residential equivalent.

R5 districts are lower-density residential districts that allow for a variety of housing types with a maximum FAR of 1.25 (FAR of 2.0 for community facilities). R5 districts typically

produce three- to four-story buildings, with required off-street accessory parking for 85 percent of the dwelling units. C1-2 commercial overlays are mapped within residential districts and allow local retail uses on the ground floors when located below residential uses in R5 zones. The maximum commercial FAR when mapped within an R5 district is 1.0. C8-1 districts are primarily automotive-related commercial and light industrial uses and preclude residential uses. C8-1 districts have a maximum commercial FAR of 1.0 and a maximum community facility FAR of 2.4.

The proposed C4-4 district permits a wide range of medium-density residential, commercial, and community facility uses. The C4-4 district has a maximum residential FAR of 4.6 (5.01 for AIRS), 3.4 FAR for commercial uses, and 6.5 FAR for community facility uses. The proposed C4-3A district also allows a similar range of residential, community facility and commercial uses. The C4-3A district has a maximum residential FAR of 3.6; the maximum commercial or community facility FAR is 3.0. There are existing C4-3A and C4-4 districts mapped southeast of the project area.

Special permit to modify bulk regulations as part of a large-scale general development (Section 74-743) (C 190366 ZSQ)

The applicant is seeking approval of a LSGD special permit (C 190366 ZSQ) to modify height and setback regulations applicable to a C4-4 district. The applicant proposes a waiver of the required side yard adjacent to an R5 district to allow the two-story base on Block B to extend to the zoning lot line. A waiver of the required rear yard equivalent is also requested to allow portions of Buildings A2 and B2 to be located within the rear yard equivalent area.

In the C4-4 district, ZR sections 35-654 and 23-664(c) require a maximum base height of 75 feet and a setback of at least 15 feet along narrow street and 10 feet along a wide street. The street walls throughout the project are proposed to range from 80 to 140 feet, so the applicant requests a waiver of the maximum base height and setback. In the C4-3A district, ZR sections 35-654 and 23-664 require a minimum base height of 40 feet, a maximum base

height of 65 feet, and a front setback of at least 15 feet along a narrow street and 10 feet along a wide street. Building F would rise to a height of 70 feet without providing the required setback. The applicant therefore requests a waiver of the maximum base height and setback. In a C4-4 district, ZR sections 35-654 and 23-664(c) stipulate a maximum building height of 135 feet (13 stories). Because maximum building heights throughout the project would range from 140 feet to 200 feet (19 stories), the applicant requests a waiver of maximum building height and maximum number of stories. ZR sections 35-654 and 23-664 set a maximum building height of 85 feet and a maximum of 8 stories in a C4-3A district. Building F would rise to a maximum building height of 90 feet (eight stories). Thus, a waiver of maximum building height is requested.

Special permit to modify surface area requirements for signage (Section 74-744(c)) – C 190375 ZSQ

Signage regulations for C4 districts generally permit a total surface area of up to five times the street frontage, not to exceed 500 square feet for each retail establishment and a maximum height of 40 feet for illuminated non-flashing signs. Pursuant to ZR Section 32-67, C1 district signage regulations are applicable within 100 feet of the street line of any street that adjoins a residential district. Thus, within 100 feet of Beach Channel Drive, the proposed signage within the rezoning area must conform with C1 signage regulations, which allow a total surface area of three times the street frontage, not to the exceed 50 square feet per retail establishment for illuminated non-flashing signs, at a height of up to 25 feet. Though the proposed signs within 100 feet of Beach Channel Drive would be below 25 feet, they would exceed the maximum allowable total surface area of 50 square feet. Therefore, the applicant proposes a waiver of the signage regulations of ZR Section 32-64 (Surface Area and Illumination Provisions).

Zoning text amendment to modify Appendix F and to allow a PCE as-of-right – N 190364

ZRQ

The applicant proposes a zoning text amendment to designate the project area as an MIH area mapped with Options 1 and 2. MIH Option 1 requires that at least 25 percent of the residential floor area be provided as housing permanently affordable to households with incomes at an average of 60 percent of AMI. Within that 25 percent, at least 10 percent of the square footage must be used for units affordable to residents with household incomes at an average of 40 percent of AMI, with no unit targeted to households with incomes exceeding 130 percent of AMI. MIH Option 2 requires that 30 percent of residential floor area be devoted to housing units affordable to residents with household incomes at an average of 80 percent of AMI. No more than three income bands can be used to average out to 80 percent, and no income band can exceed 130 percent of AMI. The applicant proposes to develop the project using MIH Option 1.

The proposed zoning text amendment would also modify ZR Section 74-744 (a), adding a new paragraph to allow a physical culture establishment (fitness center) as-of-right without obtaining a special permit from the Board of Standards and Appeals, as currently required.

City Map amendment to re-establish a portion of former Beach 52nd Street (C 190251 MMQ)

The proposed City Map change would re-establish a portion of Beach 52nd Street between Rockaway Beach Boulevard and Shore Front Parkway. This area is City-owned and comprises approximately 5,345 square feet. Beach 52nd Street is mapped to a width of 60 feet and currently extends from Rockaway Beach Boulevard south at a distance of approximately 161 feet to a dead end. The proposed action would re-establish the 89-foot by 60-foot portion of Beach 52nd Street and connect to Shore Front Parkway. This portion of street was previously de-mapped in 2003 as part of the Arverne Urban Renewal Plan (C 030433 MMQ).

Following the CPC public hearing on August 14, 2019 and in response to feedback received during the ULURP process, the applicant filed revisions to the application on August 26, 2019. The revised application would permit a flexible range of non-residential floor area,

including 72,000 square feet to 92,000 square feet of commercial floor area and 77,000 square feet to 97,000 square feet of community facility floor area. The combined non-residential floor area would not exceed 169,000 square feet and the total proposed maximum floor area would remain the same. The revised application also included the proposed addition of Use Group 3, allowing for the development of uses such as schools, day care centers, and senior centers.

Additionally, the applicant team revised the phasing plan to provide the designated PAA in an earlier phase of the project. The revised phasing plan would provide Buildings A1 and B1 in phase one, Buildings A2 and B2 in phase two, Buildings D1 and D2 in phase three, Buildings E1, E2, and the PAA in phase four, and Buildings C1, C2, and F1 in phase five.

ENVIRONMENTAL REVIEW

This application (C 190366 ZSQ) and the related applications for a zoning map amendment (C190325 ZMQ), special permit (C 190375 ZSQ), text amendment (N 190364 ZRQ), and City Map amendment (C 190251 MMQ), were reviewed pursuant to the New York State Environmental Quality Review Act (SEQRA), and the SEQRA regulations set forth in Volume 6 of the New York Code of Rules and Regulations, Section 617.00 et seq. and the City Environmental Quality Review (CEQR) Rules of Procedure of 1991 and Executive Order No. 91 of 1977. The designated CEQR number is 18DCP124Q. The lead agency is the City Planning Commission.

It was determined that this application, in conjunction with the related actions, may have a significant effect on the environment, and that an Environmental Impact Statement would be required. A Positive Declaration was issued on March 16, 2018, and subsequently distributed, published and filed. Together with the Positive Declaration, a Draft Scope of Work for the Draft Environmental Impact Statement (DEIS) was issued on March 23, 2018. A public scoping meeting was held on April 26, 2018, and the Final Scope of Work was issued on May 3, 2019.

A DEIS was prepared and a Notice of Completion for the DEIS was issued on May 3, 2019. Pursuant to SEQRA regulations and the CEQR procedures, a joint public hearing was held on the DEIS on August 14, 2019, in in conjunction with the public hearing on the related Uniform Land Use Procedure (ULURP) items (C 190325 ZMQ, C 190366 ZSQ, C 190375 ZSQ, N 190364 ZRQ, and C 190251 MMQ). A Final Environmental Impact Statement (FEIS) reflecting the comments made during the public hearing was completed, and a Notice of Completion of the FEIS was issued on September 13, 2019.

Potential significant adverse impacts related to hazardous materials, stationary air quality and noise would be avoided through the placement of (E) designations (E-532) on the project site as specified in Chapter 9, Chapter 13 and Chapter 15, respectively, of the FEIS.

The application as analyzed in the FEIS contained Project Components Related to the Environment (PCREs), which are set forth in Chapter 18, "Construction.". To ensure the implementation of the PCREs, the applicant will enter into the Restrictive Declaration, attached as Exhibit A, at the time of approval of all land use-related actions required to authorize the proposed project.

The original application as analyzed in the FEIS identified significant adverse impacts with respect to community facilities (public schools and childcare), active open space, transportation (traffic, transit bus and pedestrians), mobile air quality, and construction activities related to noise and traffic. In addition, the FEIS analyzed the revised application (called the Flexibility Alternative), as an alternative to the original application. This alternative was included in Chapter 19 of the FEIS, "Alternatives." The analysis concludes that the modified application would result in the same or similar impacts as the original application.

The identified significant adverse impacts and proposed mitigation measures under the proposed actions and Flexibility Alternative are summarized in Chapter 19, "Alternatives" and Chapter 20, "Mitigation". To ensure the implementation of the mitigation measures identified in the FEIS, the mitigation measures are included in the Restrictive Declaration.

UNIFORM LAND USE REVIEW

This application (C 190366 ZSQ), in conjunction with the related actions for a zoning map amendment (C 190325 ZMQ), special permit (C 190375 ZSQ), zoning text amendment (N 190364 ZRQ), and City Map amendment (C 190251 MMQ), was certified as complete by the Department of City Planning on May 6, 2019 and duly referred to Queens Community Board 14 and the Queens Borough President in accordance with Title 62 of the Rules of the City of New York, Section 2-02(b), along with the related application for a zoning text amendment (N 190364 ZRQ), which was referred for information and review in accordance with the procedures for non-ULURP matters.

Community Board Public Hearing

Community Board 14 held a public hearing on this application (C 190366 ZSQ) and the related applications for a zoning map amendment (C 190325 ZMQ), special permit (C 190375 ZSQ), zoning text amendment (N 190364 ZRQ), and City Map amendment (C 190251 MMQ), on June 25, 2019 and on that date, by a vote of 28 in favor, five opposed and one abstaining, recommended disapproval of the application with the following conditions:

"Density- Total amount of units should be reduced by a minimum of 30% with strong consideration for a 50% reduction on total units.

Parking- residential parking should be increased to a minimum of 75% of total units built.

Schools- That the negative impact on local schools identified in the DEIS shall be addressed by a new K-8 Department of Education school being funded, built and zoned for new residents of the project and existing local students in the immediate area for the purpose of creating a neighborhood school.

Worker Wages- That a prevailing wage agreement be signed which will be applicable for all workers

Transportation/Traffic- That in consultation with the Community Board and community- a holistic transportation/traffic plan be created and funded to mitigate the numerous traffic issues identified in DEIS that will affect the entire Peninsula from the East to the West ends of the District/peninsula.

AMI- That the final AMI mix of affordable units shall be discussed with the Community Board to determine what is needed and appropriate. The board strongly supports the addition of Affordable Senior Units.

Communication- That the developer (Arker) be required to report to the Community Board every five (5) months as to the progress of the project and to discuss and resolve issues that may arise. The developer shall create and maintain a website that shall include updates on project and job opportunities related to the project.

Shadow Impact- That the height of the building shall not exceed 12 stories

Park- That the city owned property (HPD) located north of Beach Channel Drive between approximately Beach 51 to Beach 48 Streets be deeded to NYC Parks and Recreation Department and that the Developer (Arker) provide capital funding for the planning and construction of a community park which shall include Park House, multipurpose field and running track, children's playground with water features. That the developer shall provide funding for maintenance of the park on a yearly basis

That a Community Benefits Agreement (CBA) be signed that guarantees that Community Board #14 residents get construction and other available jobs at development site, including supermarket, etc.

Resident Preference- Residents of Community Board #14 shall get a minimum of a 50% preference of all available units in each phase of development."

Borough President Recommendation

The Queens Borough President held a public hearing on the application (C 190366 ZSQ) and the related applications for a zoning map amendment (C 190325 ZMQ), special permit (C 190375 ZSQ), zoning text amendment (N 190364 ZRQ), and City Map amendment (C 190251 MMQ) on July 11, 2019, and on August 8, 2019 issued a recommendation to disapprove the application with the following conditions:

"There is a need for housing in the area. However, the need for housing in the area should not be met all on one site. CB 14 wanted a 30% to 50% reduction of units. The number of u nits should be reduced from the proposed 2200 units to 1600 units of housing. That reduction would bring the project closer to the density of the nearby Ocean Bay Apartments;

Provision of more moderate-and medium-income units for seniors, working families and individuals who wish to stay in the community. CB 14 has requested further consultation on the AMI mix of the project. Mixed income neighborhoods are important to sustain local businesses that bring jobs, goods and services to the neighborhood;

As proposed the project's tallest buildings would be more than double the height of other building in the immediate area. CB 14 has requested a 12-story cap on building heights. The heights of the tallest buildings should be lowered to be more in character with the heights of area buildings;

CB 14 has requested provision of 75% parking for proposed units. The applicant should stay in contact with CB 14 and community groups and within a year of approval there should be a reassessment to determine the adequacy of the proposed parking and whether there are any opportunities to find more spaces to meet any

unanticipated need. In any case, even with a reduction of units the proposed 973 parking spaces must be maintained;

The developer should meet with CB 14 every five month for project updates and an opportunity to discuss the project as it moves forward;

Continued efforts to bring Article 28 emergency medical facilities into the area. There was testimony at the public hearing that timeliness of treatment is a critical element to survive cardiac emergencies and that there are no emergency cardiac treatment centers on the Rockaway Peninsula. Cardiac and other Article 28 critical emergency treatment facilities must be made available on the Rockaway Peninsula;

Good schools are very important to building communities. Pre-K seats should be provided onsite. CB 14 has requested a new K to 8 school to serve the project and surrounding area. The Department of Education (DOE) must dedicate the resources necessary to improve the local schools immediately to serve the students living in the district in the vicinity of the project. Going forward DOE must account for the forecasted number of students and have funding and plans in place to have seats ready in a new school to accommodate new students."

City Planning Commission Public Hearing

On July 31, 2019 (Calendar No. 18), the Commission scheduled August 14, 2019 for a public hearing on this application (C 190366 ZSQ). The hearing was duly held on August 14, 2019 (Calendar No. 47), in conjunction with the hearings for the related actions. There were 24 speakers in favor of the application and five in opposition.

Speakers in favor included the applicant team, members of Queens Community District 14, a representative from Western Beef, representatives from Mobilizing Preachers and Communities (MPAC), members of 32BJ Service Employees International Union, members of the Rockaway Youth Task Force, a representative from Community Church of Astoria, a representative of First Baptist Church, a member of Woodside Tenants

Association, a representative of Macedonia Baptist Church, a representative of Mount Horeb Baptist Church, and Queens residents.

Those speaking in favor of the project highlighted the need for additional affordable housing in the area for existing residents and families. Several members of the Rockaway Youth Task Force spoke in favor of the proposed project, explaining the long-term need for affordable housing and employment opportunities in their community, particularly if they seek to remain in the neighborhood following graduation. Other speakers in favor of the project noted the various flood resiliency measures provided in the development and acknowledged responsible design. Speakers in support also stated the need for commercial development in the neighborhood, pointing to a lack of services and retail uses in Edgemere, and noting that the proposed project would provide a needed supermarket and local retail absent from the community.

Speakers in opposition included members of Queens Community District 14, Bayswater Civic Association, and Queens residents.

Those speaking in opposition of the project expressed concerns regarding the project's proposed density and building height, a desire for new housing provided at higher AMI levels, concerns related to potential adverse school and traffic impacts, and concerns associated with flood resiliency measures and emergency response following a future storm event.

There were no other speakers, and the hearing was closed.

Waterfront Revitalization Program Consistency Review

This application (C 190366 ZSQ), along with the related applications for a zoning map amendment (C190325 ZMQ), special permit (C 190375 ZSQ), text amendment (N 190364 ZRQ), and the City Map amendment (C 190251 MMQ), was reviewed by the City Coastal Commission for consistency with the policies of the New York City Waterfront

Revitalization Program (WRP), as amended, approved by the New York City Council on October 30, 2013 and by the New York Statement Department of State on February 3, 2016, pursuant to the New York State Waterfront Revitalization and Coastal Resources Act of 1981 (New York State Executive Law, Section 910 et seq.). The designated WRP number is 16-148.

This action was determined to be consistent with the policies of the New York City Waterfront Revitalization Program.

CONSIDERATION

The Commission believes that this application for a special permit (C 190366 ZSQ) and the related applications for a zoning map amendment (C190325 ZMQ), special permit (C 190375 ZSQ), text amendment (N 190364 ZRQ), and City Map amendment (C 190251 MMQ) are appropriate.

The proposed actions will facilitate the redevelopment of the former Peninsula Hospital site with 11 mixed residential and commercial buildings containing 2,200 affordable housing units (including 550 MIH units), community facility uses comprising up to 97,000 square feet of floor area, commercial uses comprising up to 92,000 square feet of floor area, and approximately 38,000 square feet of open space, including approximately 21,000 square feet designated as a PAA. The proposed development is consistent with one of the key initiatives of Housing New York, providing 1,650 affordable apartments in addition to the units required by MIH. Approximately 73 percent of households in the Hammels, Arverne, and Edgemere communities are rent burdened, according to the American Community Survey (ACS) data from 2016, and the proposed development will provide much-need quality of affordable housing to the area.

The proposed C4-4 and C4-3A districts are appropriate for the project area due to its adjacency to two wide streets and proximity to mass transit resources, including the elevated A subway line. Existing C4 districts are located southeast of the project area and are consistent with existing land uses and emerging trends in the area. They allow

commercial uses on the second floor or higher, allowing the development to implement resiliency measures, such as elevating the site and providing retail at a higher grade.

The request for a special permit for a large-scale general development (C 190366 ZSQ) is appropriate. The proposed bulk waivers will facilitate an appropriate project massing while providing a superior site plan and project design. The proposed site plan increases public accessibility to an existing superblock with the establishment of a new privately-maintained street network. A new segment of Beach 52nd Street will connect Rockaway Beach Boulevard and Beach Channel Drive. Peninsula Way will provide an east-west crossing for the site between Beach 50th and Beach 53rd Streets. The establishment of this new street network will provide six appropriately-sized and walkable city blocks. Streetscape features have been incorporated throughout the design to provide wide sidewalks and enhance pedestrian experiences at the street level. The street network will be subject to a public easement, and it will be maintained by the project sponsor in accordance with the terms set forth in the restrictive declaration, attached as Exhibit A.

The proposed site plan will create a range of open space opportunities, including a 21,000 square-foot PAA between building blocks D and E. This public open space will be programmed with a children's play area designed with a rubber mound, timber steppers and balance beams, as well as planters integrated with seating throughout the plaza. Building blocks A and D were designed to provide a wider sidewalk with planters and seating to act as a secondary space to be known as Highpoint Plaza.

The Commission recognizes that the variation of height and bulk facilitated by the waivers creates a superior distribution throughout the site. The variation will result in a phased development outcome that will produce a dynamic mix of building typologies. The Commission believes that the proposed height and bulk modifications will have minimal adverse effects on the structures or open space in the vicinity in terms of scale, location and access to light and air. The project's maximum building height of 19 stories is appropriate for the surrounding context, matching the heights of buildings at Arverne View, a nearby large-scale development located southwest of the project area.

The Commission applauds the proposed flood resiliency measures associated with the project, consistent with the City's goals of creating flood-resilient development. The development site will be sloped to achieve a central point (Highpoint Plaza), having a raised elevation of 14 feet, four feet higher than the Base Flood Elevation. The buildings will also incorporate wet and dry-flood proofing, landscape design providing native and coastal salt tolerant plants for a more resilient ecosystem, and place lighting fixtures and control boxes above the floodplain. The proposal also incorporates best practices in stormwater management, using bioretention rain gardens and streetscape features designed as bioswales to collect, retain, and filter drainage.

There will be a total of 973 parking spaces provided by the project, including 286 spaces more than the minimum amount required by the proposed zoning districts. Because approximately 57 percent of families in the Hammels, Arverne, and Edgemere communities own at least one car, according to ACS data from 2016, the applicant will provide the additional parking to ensure that the project is able to address concerns raised by community members. The surface parking lot, with perimeter landscaping along Beach Channel Drive and 57 spaces to support the proposed supermarket, will enhance the viability of this important retail resource being delivered by the project.

The request for a special permit to modify surface area requirements for signage (C 190375 ZSQ) is appropriate. The sign waivers will permit an increase in total surface area for illuminated and non-illuminated signs to provide additional visibility and flexibility for future tenants without negatively affecting the surrounding area. The proposed signs located beyond 100 feet of Beach Channel Drive will comply with the C1 signage provisions.

The requested City Map change (C 190251 MMQ) will re-establish a portion of Beach 52nd Street between Rockaway Beach Boulevard and Shore Front Parkway, which was demapped in 2003. This remapped street segment will serve the eight-story building

proposed at the southern end of the project site and connect to the proposed Beach 52nd segment of the private street network leading into the center of the project site.

FINDINGS

The City Planning Commission hereby makes the following findings pursuant to Section 74-743(a)(2) of the Zoning Resolution:

- (1) the distribution of floor area, open space, dwelling units, rooming units and the location of buildings, primary business entrances and show windows will result in a better site plan and a better relationship among buildings and open areas to adjacent streets, surrounding development, adjacent open areas and shorelines than would be possible without such distribution and will thus benefit both the occupants of the large-scale general development, the neighborhood and the City as a whole;
- (2) the distribution of floor area and location of buildings will not unduly increase the bulk of buildings in any one block or unduly obstruct access of light and air to the detriment of the occupants or users of buildings in the block or nearby blocks or of people using the public streets;
- (3) Not applicable
- (4) considering the size of the proposed large-scale general development, the public and private streets providing access to such large-scale general development will be adequate to handle traffic resulting therefrom;
- (5) when the Commission has determined that the large-scale general development requires significant addition to existing public facilities serving the area, the applicant has submitted to the Commission a plan and timetable to provide such required additional facilities. Proposed facilities that are incorporated into the City's capital budget may be included as part of such plan and timetable;

- (6) Not applicable
- (7) Not applicable
- (8) Not applicable
- (9) Not applicable
- (10) a declaration with regard to ownership requirements in paragraph (b) of the large-scale general development definition in Section 12-10 (DEFINITIONS) has been filed with the Commission; and
- (11) Not applicable

RESOLUTION

RESOLVED, that having considered the Final Environmental Impact Statement (FEIS), for which a Notice of Completion was issued on September 13, 2019, with respect to this application (CEQR No. 18DCP124Q), the City Planning Commission finds that the requirements of the New York State Environmental Quality Review Act and Regulations have been met and that

- Consistent with social, economic and other essential considerations, from among the reasonable alternatives thereto, the action is one which minimizes or avoids adverse environmental impacts to the maximum extent practicable; and
- 2. The adverse environmental impacts disclosed in the FEIS will be minimized or avoided to the maximum extent practicable by incorporating, as conditions to the approval, pursuant to the restrictive declaration attached as Exhibit A to City Planning Commission report for C 190366 ZSQ, those project components related to environment and mitigation measures that were identified as practicable.

The report of the City Planning Commission, together with the FEIS constitutes the written statement of facts, and of social, economic and other factors and standards, that form the basis of the decision, pursuant to Section 617.11(d) of the SEQRA regulations; and be it further

RESOLVED, that the City Planning Commission, in its capacity as the City Coastal Commission, has reviewed the waterfront aspects of this application and finds that the proposed action is consistent with WRP policies; and be it further

RESOLVED, by the City Planning Commission, pursuant to Sections 197-c and 201 of the New York City Charter, that based on the environmental determination, and the consideration and findings described in this report, the application submitted by Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the New York City Charter, for the grant of special permits pursuant to Section 74-743(a)(2) of the Zoning Resolution to modify the rear yard requirements of Section 23-533 (Required rear yard equivalents for Quality Housing buildings) and Section 35-53 (Modification of Rear Yard Requirements), the side yard requirements of Section 35-54 (Special Provisions Applying Adjacent to R1 Through R5 Districts), and the height and setback requirements of Section 23-664 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors) and Section 35-654 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors), in connection with a proposed mixed used development, within a large-scale general development, on property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, Beach 52nd Street, Rockaway Beach Boulevard and Beach 53rd Street (Block 15842, Lot 1 & p/o Lot 100, Block 15843, Lot 1, and Block 15857 Lot 1 & p/o Lot 7), in a C4-4 and C4-3A Districts, Borough of Queens, Community District 14, is approved, pursuant to Section 74-743(a)(2) of the Zoning Resolution, subject to the following terms and conditions:

1. The property that is the subject of this application (C 190366 ZSQ) shall be developed in size and arrangement substantially in accordance with the dimensions, specification and zoning computation indicated on the following approved plans, prepared by Aufgang Architects and terrain-nyc landscape architecture pc, filed with this application and incorporated in this resolution:

Dwg. No.	<u>Title</u>	Last Date Revised
Z-001.00	Zoning Analysis	09/23/2019
Z-100.00	Zoning Lot Site Plan	05/06/2019
Z-110.00	Waiver Plan	05/06/2019
Z-111.00	Waiver Section	05/06/2019
Z-112.00	Waiver Section	05/06/2019
Z-113.00	Waiver Section	05/06/2019
L-100	Pedestrian Plaza Plan	05/06/2019
L-200	Pedestrian Plaza Planting Plan	05/06/2019
L-201	Pedestrian Plaza Planting Details	05/06/2019
L-300	Pedestrian Plaza Seating Plan	05/06/2019
L-301	Pedestrian Plaza Seating Details 1	05/06/2019
L-302	Pedestrian Plaza Seating Details 2	05/06/2019
L-303	Pedestrian Plaza Seating Parts Plan	05/06/2019
L-400	Pedestrian Plaza Grading and Paving	Plan 05/06/2019
L-401	Pedestrian Plaza Grading and Paving	Details 05/06/2019
L-500	Pedestrian Plaza Photometric Plan	05/06/2019
L-501	Pedestrian Plaza Lighting Cut Sheet	05/06/2019
L-600	Pedestrian Plaza Signs and Site Furnis	shing 05/06/2019
	Details	
L-601	Pedestrian Plaza Play Area Part Plan	05/06/2019
L-602	Pedestrian Plaza Sections	05/06/2019

- 2. Such development shall conform to all applicable provisions of the Zoning Resolution, except for the modifications specifically granted in this resolution and shown on the plans listed above which have been filed with this application. All zoning computations are subject to verification and approval by the New York City Department of Buildings.
- 3. Such development shall conform to all applicable laws and regulation relating to its construction, operation, and maintenance.
- 4. Development pursuant to this resolution shall be allowed only after the restrictive declaration attached hereto as Exhibit A, with such administrative changes as are acceptable to Counsel to the Department of City Planning, has been executed and recorded in the Office of the City Register, New York County. Such restrictive declaration shall be deemed incorporated herein as a condition to this resolution.
- 5. In the event that the property that is the subject of the application is developed as, sold as, or converted to condominium units, a homeowner's association or cooperative ownership, a copy of this report and resolution and any subsequent modification shall be provided to the Attorney General of the State of New York at the time of application for any such condominium, homeowner's or cooperative offering plan and, if the Attorney General so directs, shall be incorporated in full in any offering documents relating to the property.
- 6. All leases, subleases, or other agreements for use or occupancy of space at the subject property shall give actual notice of this special permit to the lessee, sublessee, or occupant.
- 7. Upon failure of any party having any right, title or interest in the property that is the subject of this application, or the failure of any heir, successor, assign, or legal representative of such party, to observe any of the covenants, restrictions, agreements, terms or conditions of this resolution whose provisions shall

constitute conditions of the special permit hereby granted, the City Planning Commission may, without the consent of any other party, revoke any portion of or all of said special permit. Such power of revocation shall be in addition to and not limited to any other powers of the City Planning Commission, or of any agency of government, or any private person or entity. Any such failure as stated above, or any alteration in the development that is the subject of this application that departs from any conditions listed above, is grounds for the City Planning Commission or the City Council, as applicable, to disapprove any application for modification, cancellation or amendment of the special permit hereby granted.

8. Neither the City of New York nor its employees or agents shall have any liability for money damages by reason of the city's or such employee's or agent's failure to act in accordance with the provisions of this special permit.

The above resolution (C 190366 ZSQ), duly adopted by the City Planning Commission on September 25, 2019 (Calendar No. 26) is filed with the Office of the Speaker, City Council, and the Borough President in accordance with the requirements of Section 197-d of the New York City Charter.

MARISA LAGO, Chair
KENNETH J. KNUCKLES, Esq., Vice-Chairman
DAVID J. BURNEY, ALLEN P. CAPPELLI, Esq., ALFRED C. CERULLO, III,
MICHELLE de la UZ, JOSEPH I. DOUEK, HOPE KNIGHT,
ANNA HAYES LEVIN, ORLANDO MARIN, LARISA ORTIZ,
RAJ RAMPERSHAD Commissioners

"Exhibit A"

DECLARATION OF LARGE-SCALE GENERAL DEVELOPMENT

Dated as of _____, 201[]

QUEENS COUNTY

Block 15842, Lot 1 Block 15843, Lot 1 and Block 15857, Lot 1

RECORD AND RETURN TO:

Akerman LLP 666 Fifth Avenue, 20th Floor New York, NY 10103

Attention:

TABLE OF CONTENTS

		<u>Page</u>
ARTICLE I CE	RTAIN DEFINITIONS	3
ARTICLE II DE	EVELOPMENT AND USE OF THE SUBJECT PROPERTY	9
2.1	Designation of the Large-Scale General Development	9
2.2	Development of Large-Scale Development Site	
2.3	Submission of As-Built Drawings	
ARTICLE III P	UBLICLY ACCESSIBLE AREA	11
3.1	Construction of the Publicly Accessible Area	
3.1	PAA Easement	
3.3		
	Hours of Access	
3.4	Maintenance and Repair	
3.5	Signage	
3.6	Private Easement	15
ARTICLE IV P	RIVATE STREET NETWORK	15
4.1	Design and Construction	15
4.2	Street Easement	16
4.3	Liens	16
4.4	Use and Floor Area	16
4.5	Operational Requirements	16
4.6	Street Maintenance Obligation	
4.7	Closure of Private Street Network	
4.8	Maintenance and Access Security	
4.9	Right of Inspection.	
4.10	Right of Access	
4.11	Right of Access for Repairs	
4.12	Assumption of Liability	
4.13	Waiver	
4.13	w aivei	,17
ARTICLE V EN	IVIRONMENTAL PROTECTION MEASURES	19
5.1	Project Component Related to the Environment	
5.2	Environmental Mitigation Relating to Open Space	
5.3	Environmental Mitigation Relating to Community Facilities	22
5.4	Environmental Mitigation Relating to Operational and Construction Transportation	23
5.5	Environmental Mitigation Relating to Operational Air Quality	
5.6	Environmental Mitigation Relating to Construction Noise	
5.7	Force Majeure Involving a PCRE or Mitigation Measure	
5.8	Incorporation of FEIS Mitigation Measures	
5.9	Innovation; Alternatives; Modifications Based on Further Assessments	

ARTICLE VI	CERTIFICATES OF OCCUPANCY	25
6.1	Temporary Certificates of Occupancy	25
6.2	Notice of Substantial Completion	
6.3	Force Majeure.	26
6.4	Permanent Certificates of Occupancy	27
ARTICLE VII	ENFORCEMENT	28
7.1	Enforcement by the City	
7.2	Additional Remedies	
7.3	Enforcement by Declarant	28
7.4	Notice and Cure	28
7.5	Default	29
ARTICLE VII	I MISCELLANEOUS	29
8.1	Representation	29
8.2	Binding Nature; Successors and Assigns	29
8.3	Parties in Interest	30
8.4	Counterparts	30
8.5	Recordation	30
8.6	Effective Date.	30
8.7	Notice	30
8.8	Applications	31
8.9	Severability	
8.10	Applicable Law	
8.11	Exhibits	
ARTICLE IX	CONDOMINIUMS AND COOPERATIVE CORPORATIONS	32
9.1	Filing Requirements	32
9.2	Offering Plans	
9.3	Common Elements	
9.4	Estoppel	
ARTICLE X P	ROPERTY OWNER'S ASSOCIATION	34
10.1	Property Owner's Association	34
10.2	Obligations	
10.3	Members	
10.4	Powers	35
10.5	Assessments	
10.6	Deed References	
ADTICI E VI	AMENDMENT MODIFICATION & CANCELLATION	27

SCHEDULE OF EXHIBITS

EXHIBIT A	DESCRIPTION OF SUBJECT PROPERTY
EXHIBIT B	CERTIFICATION OF PARTIES IN INTEREST
EXHIBIT C	SITE PLAN
EXHIBIT D	FORM OF NOTICE OF SUBSTANTIAL COMPLETION
EXHIBIT E	FORM OF NOTICE OF FINAL COMPLETION
EXHIBIT F-1	PROJECT COMPONENTS RELATED TO THE ENVIRONMENT
	 CONSTRUCTION AIR QUALITY
	 CONSTRUCTION NOISE
EXHIBIT F-2	MITIGATION MEASURES
	 CONSTRUCTION & OPERATIONAL TRANSPORTATION
	 OPERATIONAL AIR QUALITY
	 CONSTRUCTION NOISE
EXHIBIT G	PROJECT PHASING DIAGRAM
EXHIBIT H	PRIVATE STREET NETWORK TEMPORARY AND PERMANENT DESIGN DRAWINGS

DECLARATION OF LARGE-SCALE GENERAL DEVELOPMENT

THIS DECLARATION (the "<u>Declaration</u>"), dated as of ______, 2019, by **PENINSULA ROCKAWAY LIMITED PARTNERSHIP**, a New York State limited partnership having an address c/o The Arker Companies, 1044 Northern Boulevard, Roslyn, New York 11576 (the "<u>Declarant</u>").

WITNESSETH:

WHEREAS, Peninsula Rockaway Housing Development Fund Corp., a New York not-for-profit corporation, (the "<u>Nominee</u>") is the nominal fee owner of the premises located in the Borough of Queens, County of Queens, City of New York, and State of New York, designated for real property tax purposes as (a) Block 15842, Lot 1 and Block 15843, Lot 1 ("<u>Parcel 1</u>") and (b) Block 15857, Lot 1 ("<u>Parcel 2</u>") on the Tax Map of the City of New York, which real property is more particularly described in attached Exhibit A (Parcel 1 and Parcel 2 are collectively referred to as the "<u>Subject Property</u>");

WHEREAS, pursuant to that certain Declaration of Interest and Nominee Agreement dated as of May 5, 2016, and recorded as CRFN 2016000172286 by and between Declarant and Nominee, Nominee has granted to Declarant all equitable and beneficial interests in the Subject Property;

WHEREAS, Declarant desires to improve the Subject Property as a "large-scale general development" pursuant to the requirements of Section 12-10 of the Zoning Resolution definition of a "large scale general development" (such proposed improvement of the Subject Property hereinafter referred to as the ("**Large-Scale Development Project**");

WHEREAS, Declarant intends that the Large-Scale Development Project will contain eleven (11) new residential, community facility, and commercial buildings that will contain affordable and moderate-income housing that incorporate sustainability and resiliency features on the Subject Property;

WHEREAS, Declarant intends to develop the Large-Scale Development Project in five phases, as set forth in Exhibit G;

WHEREAS, the Site Plan, as defined herein, for the Large-Scale Development Project includes a Private Street Network, attached as Exhibit H to provide for a better site plan and facilitate improved circulation within the superblock and in the neighborhood (the "<u>Private Street Network</u>");

WHEREAS, Declarant intends to construct the Private Street Network in five phases, and all interim and final segments of the Private Street Network will be publicly accessible and privately maintained as set forth in this Declaration;

WHEREAS, the Site Plan for the Large-Scale Development Project, attached as Exhibit C, designates a portion of the Subject Property as Publicly Accessible Area that is to be constructed

and maintained by Declarant and accessible to the public as set forth in this Declaration (the "Publicly Accessible Area" or "PAA");

WHEREAS, in connection with the Large-Scale Development Project, Declarant filed an application with the New York City Department of City Planning ("DCP") for approval by New York City Planning Commission (the "Commission" or "CPC") of: (1) a zoning map amendment to rezone Parcel 1 from an R5 and R5/C1-2 zoning districts to a C4-4 zoning district and Parcel 2 from a C8-1 to a C4-3A zoning district under Application No. 190325 ZMQ; (2) zoning text amendments under Application No. 190364 ZRQ to designate the Subject Property a Mandatory Inclusionary Housing area and permit a physical culture establishment as-of-right; (3) a special permit, pursuant to Zoning Resolution Section 74-743 to waive applicable yard requirements and height and setback regulations under Application No. 190366 ZSQ (the "Large-Scale Special Permit"); (4) a special permit, pursuant to Zoning Resolution Section 74-744(c)(1) to allow signage that exceeds the surface area requirements under Application No. 190375 ZSQ; and (5) a City Map amendment to map a portion of Beach 52nd Street under Application No. 190251 MMQ (collectively referred to as the "Land Use Applications");

WHEREAS, the Commission acting as lead agency for the City Environmental Quality Review Application No. 18DCP124Q conducted an environmental review of the Land Use Applications pursuant to Executive Order No. 91 of 1977, as amended, and the regulations promulgated thereunder at 62 RCNY§5-01 et seq. ("CEQR") and the State Environmental Quality Review Act, New York State Environmental Conservation Law § 8-0101 et seq. and the regulations promulgated thereunder at 6 NYCRR Part 617 ("SEQRA"), and issued a Notice of Completion for the Final Environmental Impact Statement (the "FEIS") prepared the Land Use Applications on September 13, 2019;

WHEREAS, at the time of the Commission's Approval of the Land Use Applications the Commission found, as required pursuant to SEQRA, that the action is one that avoids or minimizes adverse environmental impacts to the maximum extent practicable and that the adverse impacts will be avoided or minimized to the maximum extent practicable by incorporating as conditions of the decision certain project components related to the environment ("PCREs") and those Mitigation Measures, as defined herein, that were identified in the FEIS as practicable;

WHEREAS, to ensure that the development of the Subject Property pursuant to the Final Approval, as defined herein, is consistent with the analysis in the FEIS upon which the Commission has made its findings and that the development of the Subject Property incorporates certain PCREs and Mitigation Measures for potential significant adverse environmental impacts as conditions of the Commission's decision on the Land Use Applications, Declarant has agreed to restrict the development, operation, use, and maintenance of the Subject Property in certain respects, which restrictions are set forth in this Declaration;

WHEREAS, Section 74-743(b)(10) of the Zoning Resolution requires that a declaration with regard to ownership requirements in paragraph (b) of the large-scale general development definition in Section 12-10 of the Zoning Resolution be filed with the Commission;

WHEREAS, pursuant to the certificate attached as Exhibit B:1 (the "<u>Title Company</u>") has certified that, as of the date of this Declaration, Declarant, Nominee and Wells Fargo Bank,

N.A. are the sole parties-in-interest (the "<u>Parties-in-Interest</u>") in the Subject Property, as such term is defined in the definition of "zoning lot" in Section 12-10 of the Zoning Resolution;

WHEREAS, all Parties in Interest have either executed this Declaration or waived their rights to execute this Declaration by written instrument attached as Exhibit B:2, which instrument is intended to be recorded simultaneously with this Declaration;

WHEREAS, Declarant desires, on the terms and conditions herein, to restrict the manner in which the Subject Property may be developed, redeveloped, maintained, and operated now and in the future and intends these restrictions to benefit the Subject Property and all the land lying within a one-half mile radius of the Subject Property; and

NOW, THEREFORE, Declarant hereby declares, covenants, and agrees as follows:

ARTICLE I

CERTAIN DEFINITIONS

For purposes of this Declaration, the following terms shall have the following meanings.

- "Adjustments" shall have the meaning set forth in Section 5.4(a).
- "AG" shall mean the Attorney General of the State of New York.
- "<u>Applicable Law</u>" shall mean all presently existing or hereafter enacted laws, orders, ordinances, rules, regulations and requirements of all federal, state, municipal, and local governments and their departments, agencies, commissions, boards, and officers, and any other governmental agency applicable to or having appropriate jurisdiction over the Subject Property or any portion thereof.
- "<u>Approval</u>" shall mean approval of the Land Use Applications by the Commission relating to the Large-Scale Development Project or otherwise with respect to the Subject Property.
 - "Association" shall have the meaning set forth in Section 10.1 of this Declaration.
 - "Assessment Property" shall have the meaning set forth in Section 10.5.
 - "Association Members" shall have the meaning set forth in Section 10.3.
 - "Association Obligation Date" shall have the meaning set forth in Section 10.1.
 - "Board" shall have the meaning set forth in Section 9.1.
- "<u>Building</u>" or "<u>Buildings</u>" shall mean any one of the buildings shown on the Site Plan or all such buildings, as the context shall require.
- "<u>Building Permit</u>" shall mean the issuance of any permit by DOB whether in the form of (i) an excavation permit, authorizing excavations, including those made for the purposes of

removing earth, sand, gravel, or other material from the Subject Property; (ii) a foundation permit, authorizing foundation work at the Subject Property; (iii) a demolition permit, authorizing the dismantling, razing or removal of a building or structure, including the removal of structural members, floors, interior bearing walls and/or exterior walls or portions thereof; (iv) a New Building Permit, as defined herein; or (v) any other permit normally associated with the development of a building.

"<u>Business Day</u>" shall mean any day other than a Saturday, Sunday, or other day on which banks in the State of New York are authorized or required by Applicable Law to be closed.

"CEQR" shall have the meaning given in the Recitals to this Declaration.

"<u>Chairperson</u>" shall mean the Chair of the Commission from time to time or any successor to the jurisdiction thereof.

"City" shall mean the City of New York.

"<u>City Council</u>" shall mean the City Council of the City of New York or any successor to the jurisdiction thereof.

"Commission" or "CPC" shall have the meaning given in the Recitals to this Declaration.

"Construction Commencement" shall mean the issuance of the first Building Permit by DOB to Declarant for the commencement of work to develop the Subject Property, in whole or in part, regarding the Large-Scale Development Project or any portion thereof.

"Coop/Condominium" shall have the meaning set forth in Section 9.1(a).

"Condominium Obligation Date" shall have the meaning set forth in Section 9.1(a).

"<u>DCP</u>" shall mean the New York City Department of City Planning or any successor to the jurisdiction thereof.

"<u>Declarant</u>" shall mean the named Declarant, the heirs, successors and assigns of the named Declarant, the Coops/Condominiums as set forth in Article IX, and the Association as set forth in Article X, except that: (i) Declarant shall not include the holder of a mortgage or deed of trust on all or any portion of the Subject Property unless and until it succeeds to the interest or obligation of Declarant by purchase, assignment, foreclosure or otherwise; and (ii) Declarant shall only include a Coop/Condominium from and after the Coop/Condominium Obligation Date and the Association from and after the Association Date.

"<u>Declaration</u>" shall have the meaning given in the Preamble to this Declaration.

"Development" shall mean the construction of the Large-Scale Development Project.

"Delay Notice" shall have the meaning set forth in Section 6.3 of this Declaration.

- "<u>DOB</u>" shall mean the New York City Department of Buildings or any successor to its jurisdiction.
- "<u>DOE</u>" shall mean the New York City Department of Education or any successor to its jurisdiction.
- "<u>DOT</u>" shall mean the New York City Department of Transportation or any successor to its jurisdiction.
- "<u>DPR</u>" shall mean the New York City Department of Parks and Recreation or any successor to its jurisdiction.
- "<u>Effective Date</u>" shall mean the date on which this Declaration is recorded in the Register's Office following New York City Council's adoption of the Land Use Applications.
- "Entity" shall mean any general partnership, limited partnership, limited liability company, corporation, joint venture, trust, business trust, cooperative, or association.
- "FDNY" shall mean the New York City Fire Department or any successor to its jurisdiction.
 - "FEIS" shall have the meaning set forth in the Recitals to this Declaration.
- "FEIS Requirement" shall mean any measure set forth in the FEIS that is required in order for the Commission to find in the Commission's Approval of the Land Use Applications that the action is one that avoids or minimizes adverse environmental impacts to the maximum extent practicable, and that the adverse impacts will be avoided or minimized to the maximum extent practicable by incorporating as conditions of the Approval PCREs and those Mitigation Measures that were identified in the FEIS as practicable.
- "Final Approval" shall mean approval of the Land Use Applications by the Commission pursuant to New York City Charter Section 197-c, which shall be effective on the date that the City Council's period of review has expired, unless: (a) pursuant to New York City Charter Section 197-d(b), the City Council reviews the decision of the Commission approving the Land Use Applications and takes final action pursuant to New York City Charter Section 197-d approving or approving with modifications the Land Use Applications, in which event "Final Approval" shall mean such approval of the Land Use Applications by the City Council; or (b) the City Council disapproves the decision of the Commission and the Office of the Mayor files a written disapproval of the City Council's action pursuant to New York City Charter Section 197-d(e), and the City Council does not override the Office of the Mayor's disapproval, in which event "Final Approval" shall mean the Office of the Mayor's written disapproval pursuant to such New York City Charter Section 197-d(e). Notwithstanding anything to the contrary contained in this Declaration, "Final Approval" shall not be deemed to have occurred for any purpose of this Declaration if the final action taken pursuant to New York City Charter Section 197-d is disapproval of the Land Use Applications.

"<u>Final Completion</u>" or "<u>Finally Complete</u>" shall mean the completion of all relevant items of work that remain to be completed after Substantial Completion.

"<u>Final Street Network Plan</u>" shall mean plans of the complete Private Street Network and all Street Network Phases, which conform to the standards set by DOB for Builders Pavement Plans.

"Force Majeure Event" shall include, but not be limited to (i) governmental restrictions, regulations or controls; (ii) enemy or hostile government action, civil commotion, insurrection, revolution, terrorism or sabotage; (iii) fire or other casualty; (iv) inclement weather substantially delaying construction of any relevant portion of the Subject Property; (v) failure or inability of a public utility to provide power, heat or light or any other utility service; (vi) strikes, lockouts or labor disputes; (vii) inability to obtain labor or materials or reasonable substitutes therefor (unless due to any act or omission of Declarant); (viii) acts of God; (ix) a taking of the whole or a portion of the Subject Property by condemnation or eminent domain; (x) denial to Declarant by any party of a right of access to any adjoining real property which right is vested in Declarant by contract or pursuant to applicable law, if such access is required to accomplish the obligations of Declarant pursuant to this Declaration; (xi) any undue material delay in the issuance of approvals by any department or agency of the City, the State of New York or the United States that is not caused by any act or omission of Declarant; (xii) underground or soil conditions that were not and could not reasonably have been foreseen by Declarant prior to their discovery or occurrence; (xiii) the pendency of any litigation relating to the Application or to the underlying sections of the Zoning Resolution; or (xiv) any other condition similar to the foregoing which are beyond Declarant's control.

"Governmental Authority" shall mean any governmental authority (including any Federal, State, City, or County governmental authority or quasi-governmental authority, or any political subdivision of any thereof, or any agency, department, commission, board or instrumentality of any thereof) having jurisdiction over the matter in question.

"HPD" shall mean the New York City Department of Housing Preservation and Development, or any successor to its jurisdiction.

"<u>Land Use Applications</u>" shall have the meaning set forth in the Recitals to this Declaration.

"Large-Scale Development Project" shall have the meaning set forth in the Recitals.

"Large-Scale Special Permit" shall have the meaning set forth in the Recitals.

"Maintenance Obligations" shall refer to the totality of Declarant's obligations and consist of both of the PAA Maintenance Obligation and the Street Maintenance Obligation.

"Maintenance and Protection of Traffic Plan" or "MPT" shall have the meaning set forth in Section 5.1(c).

"Mayor" shall mean the Mayor of the City of New York.

"Modification of Mitigation Measure" shall have the meaning set forth in Section 5.9.

- "Monitor" shall mean the environmental monitor hired to ensure compliance with the PCREs set forth in Article V.
- "Monitor Agreement" shall mean the agreement between the Declarant and Monitor as set forth in Section 6.1(c).
- "<u>Mitigation Measure</u>" shall mean the measures set forth in Article V and Exhibit F1-F2 that are required to mitigate the significant adverse impacts identified in the FEIS.
- "Mortgage" shall mean a mortgage given as security for a loan in respect of all or any portion of the Subject Property, other than a mortgage secured by any condominium unit or other individual residential unit located within the Subject Property.
 - "Mortgagee" shall mean the holder of a Mortgage.
- "New Building Permit" shall mean the issuance of a permit by DOB in the form a "New Building" Job type authorizing construction of a new Building.
- "New York City Charter" shall mean the Charter of the City of New York, effective as of January 1, 1990, as the same may be amended from time to time.
- "<u>Notice of Final Completion</u>" shall have the meaning set forth in Section 6.4 of this Declaration.
- "<u>Notice of Substantial Completion</u>" shall have the meaning set forth in Section 6.2 of this Declaration.
 - "Offering Plan" shall have the meaning set forth in Section 9.2.
- "<u>Open Space Mitigation Measure</u>" shall mean the measures set forth in Article V of this Declaration that are required to partially mitigate the significant adverse impacts on active open space identified in the FEIS.
 - "Open Space Mitigation Notice" shall have the meaning set forth in Section 5.2(c).
 - "Open Space Mitigation Interested Parties" shall mean DPR and DCP.
- "<u>Operational Air Quality Mitigation Measure</u>" shall mean the measures set forth in Article V of this Declaration that are required to mitigate the significant adverse impacts on air quality identified in the FEIS.
- "<u>Operational Specifications</u>" shall have the meaning set forth in Section 4.5 of this Declaration.
- "Party-in-Interest" shall have the meaning set forth in subdivision (d) of the definition of the term "zoning lot" in Section 12-10 of the Zoning Resolution.
 - "PCO" shall mean a Permanent Certificate of Occupancy issued by DOB.

- "Person" shall mean any individual or Entity, and the heirs, executors, administrators, legal representatives, successors, and assigns of such Person as the context may require.
 - "Private Street Network" shall have the meaning set forth in the Recitals.
- "<u>Project Components Related to the Environment</u>" or "<u>PCRE</u>" shall mean the measures set forth in Article V of this Declaration that are identified in the FEIS as components of the Large-Scale Development Project that are required to avoid impacts to the environment.
 - "Parcel 1" shall have the meaning set forth in the Recitals to this Declaration.
 - "Parcel 2" shall have the meaning set forth in the Recitals to this Declaration.
 - "Possessory Interest" shall have the meaning set forth in Section 10.5(d).
 - "Publicly Accessible Area" or "PAA" shall have the meaning set forth in the Recitals.
 - "PAA Easement" shall have the meaning set forth in Section 3.2 of this Declaration.
- "PAA Maintenance Obligation" shall have the meaning set forth in Section 3.4 of this Declaration.
- "Phase" shall mean any of the phases shown in the Phasing Diagram attached as Exhibit G.
- "Register's Office" shall mean the Register's Office of the City of New York, Kings County.
- "<u>School Mitigation Measure</u>" shall mean the measures set forth in Article V of this Declaration that are required to partially mitigate the significant adverse impacts on public elementary and intermediate schools identified in the FEIS.
 - "School Mitigation Notice" shall have the meaning set forth in Section 5.3(d).
 - "School Mitigation Interested Parties" shall mean SCA and DOE.
 - "Scope of Work" shall have the meaning set forth in Section 5.4(b).
- "<u>Site Plan</u>" shall mean the plans and drawings set forth in Exhibit C and referred to in Article 2.2 of this Declaration.
 - "SEQRA" shall have the meaning set forth in the Recitals to this Declaration.
 - "State" shall mean the State of New York, its agencies, and instrumentalities.
 - "Subject Property" shall have the meaning set forth in the Recitals to this Declaration.
- "<u>Substantial Completion</u>" or "<u>Substantially Complete</u>", with respect to the Publicly Accessible Area, shall mean that such Publicly Accessible Area has been constructed substantially

in accordance with the Site Plan, as same may be amended from time to time in accordance herewith, and has been completed to such an extent that all portions of the improvement may be operated and made available for public use. An improvement may be deemed Substantially Complete notwithstanding that (a) minor or insubstantial items of construction, decoration or mechanical adjustment remain to be performed or (b) Declarant has not completed any relevant planting or vegetation or tasks that must occur seasonally. With respect to segments of the Private Street Network, Substantial Completion or Substantially Complete shall mean that the street segments associated with a Street Network Phase, as defined herein, have been constructed substantially in accordance with the Street Final Plan and have been completed to such an extent that the street segments may be operated and made available for public use, notwithstanding minor or insubstantial items of construction, planting, or street furniture.

"Street Easement" shall have the meaning set forth in Article IV of this Declaration.

"<u>Street Maintenance Obligation</u>" shall have the meaning set forth in Article IV of this Declaration.

"<u>Street Network Phase</u>" shall mean any of the phases shown in the Private Street Network Phasing Diagram attached as Exhibit H.

"Successor Declarant" shall have the meaning set forth in Section 9.1(b).

"TCO" shall mean a Temporary Certificate of Occupancy issued by DOB.

"<u>Title Company</u>" shall mean a title insurance company licensed to do business by the State of New York.

"<u>Transportation Mitigation Measure</u>" shall mean the measures set forth in Article V of this Declaration that are required to partially mitigate the significant adverse impacts on transportation identified in the FEIS.

"Unit Interested Party" shall have the meaning set forth in Section 9.1.

"Zoning Resolution" shall mean the Zoning Resolution of the City of New York, effective December 15, 1961, as amended from time to time.

ARTICLE II

DEVELOPMENT AND USE OF THE SUBJECT PROPERTY

- 2.1 <u>Designation of the Large-Scale General Development</u>. Declarant hereby declares and agrees that, following the Effective Date, the Subject Property, if developed pursuant to the Large-Scale Special Permit, shall be treated as a large-scale general development site and shall be developed and enlarged as a single unit pursuant to the Zoning Resolution.
- 2.2 <u>Development of Large-Scale Development Site.</u> If the Subject Property is developed in whole or part in accordance with the Large-Scale Special Permit, Declarant covenants that the Subject Property shall be developed in substantial conformity with the following

plans prepared by Aufgang Architects, Terrain-NYC Landscape, and PHT Lighting Design Inc., approved as part of the Large-Scale Special Permit and attached in Exhibit "C" and made a part hereof (the "<u>Site Plan</u>"):

C 190366 ZSQ

Drawing	No. <u>Title</u>	Last Date Revised
Z-001.00	Zoning Analysis	09/23/2019
Z-100.00	Zoning Lot Site Plan	05/06/2019
Z-110.00	Waiver Plan	05/06/2019
Z-111.00	Waiver Section	05/06/2019
Z-112.00	Waiver Section	05/06/2019
Z-113.00	Waiver Section	05/06/2019
L-100	Pedestrian Plaza Plan	05/06/2019
L-200	Pedestrian Plaza Planting Plan	05/06/2019
L-201	Pedestrian Plaza Planting Details	05/06/2019
L-300	Pedestrian Plaza Seating Plan	05/06/2019
L-301	Pedestrian Plaza Seating Details 1	05/06/2019
L-302	Pedestrian Plaza Seating Details 2	05/06/2019
L-303	Pedestrian Plaza Seating Parts Plan	05/06/2019
L-400	Pedestrian Plaza Grading and Paving Plan	05/06/2019
L-401	Pedestrian Plaza Grading and Paving Details	05/06/2019
L-500	Pedestrian Plaza Photometric Plan	05/06/2019
L-501	Pedestrian Plaza Lighting Cut Sheet	05/06/2019
L-600	Pedestrian Plaza Signs and Site Furnishing Details	05/06/2019
L-601	Pedestrian Plaza Play Area Part Plan	05/06/2019
L-602	Pedestrian Plaza Sections	05/06/2019

C 190375 ZSQ

<u>Drawing No.</u> <u>Title</u>	Last Date Revised
Z-001.00 Zoning Analysis	09/23/2019
Z-100.00 Zoning Lot Site Plan	05/06/2019
Z-120.00 Signage Details	05/06/2019

2.3 <u>Submission of As-Built Drawings</u>. Following the issuance of a Permanent Certificate of Occupancy for a Building, as-built drawings for the Building, associated segments of the Private Street Network, and the PAA, if applicable, shall be submitted to DCP. The asbuilt drawings shall include the following: Declarant's company name, address and telephone number, Subject Property's address, drawing date, the word "As-Built", and Declarant's signature.

ARTICLE III

PUBLICLY ACCESSIBLE AREA

- 3.1 Construction of the Publicly Accessible Area.
- (a) Declarant shall construct the Publicly Accessible Area substantially in accordance with the specifications in Drawing No. L-100-602 of the Site Plan attached in Exhibit C.
- (b) Declarant, at its sole cost and expense, shall diligently apply for and prosecute the applications for all City, State, and Federal permits and approvals to fully construct the Public Accessible Area.
- (c) Declarant shall not accept a Temporary or Permanent Certificate of Occupancy, as set forth in Article VI, for Building E2 in Phase 4, as shown in Exhibit G, until the Chairperson has determined that the PAA is Substantially Complete.

3.2 PAA Easement.

- (a) Immediately upon the certification of Substantial Completion, Declarant grants the City and the general public a permanent, perpetual access easement over the entirety of the Publicly Accessible Area, subject to the terms of this Declaration, unobstructed from the surface of the Publicly Accessible Area to the sky, for the purposes of (i) passive recreational use by the general public and (ii) pedestrian access (the "PAA Easement").
- (b) All liens, including but not limited to judgment liens, mortgage liens, mechanics' liens and vendees' liens, and all burdens, covenants, encumbrances, leases, licensees, easements, profits, security interests in personal property or fixtures, and all other interests subsequent thereto, excepting governmental tax liens and assessments and public utilities and easements, shall be subject and subordinate to the rights, claims, entitlements, interests, and priorities created by the PAA Easement as herein defined in Article 3.2(a).

3.3 Hours of Access.

- (a) The Publicly Accessible Area shall be open and accessible to the public each day of the week between 6:00 am to 10:00 pm all year.
- (b) Declarant may close the Publicly Accessible Area in order to: (i) perform required maintenance, repairs, or replacements of the Publicly Accessible Area, or portions thereof, and shall notify the Chairperson of such closure no less than seven (7) days in advance and such notice shall set forth the area and duration of closure as well as confirm the posting of signs providing prior notice to the public at appropriate locations and entrances of the Publicly Accessible Area; (ii) perform required repair, restoration, rehabilitation, renovation, or replacement of pipes, utility lines or conduits or other equipment on or under the Publicly Accessible Area and shall notify the Chairperson of such closure no less than ten (10) days in advance and such notice shall set forth the area and duration of closure as well as confirm the posting of signs providing prior notice to the public at appropriate locations and entrances of the Publicly Accessible Area; or (iii) to make emergency repairs to mitigate hazardous site conditions or address other emergency conditions as specified in this Article, III.
- (c) In the event that the closure of the Publicly Accessible Area is required due to an emergency condition specified herein, Declarant shall notify the Chairperson of such closing and its expected duration as soon as practicable but in no event more than two (2) business days after such closure. The notice to the Chairperson shall further specify which portion has been closed and describe the nature of the emergency or hazardous condition causing the closure. Emergency conditions for which the Publicly Accessible Area may be closed, pursuant to Article 3.3(b), shall be limited to actual or imminent emergency situations, including security alerts, riots, casualties, disasters, or other events endangering public safety or property, provided that no such emergency closure shall continue for more than forty-eight (48) consecutive hours without Declarant having consulted with DOB or other agency and such agency confirming the continued closure of the Publicly Accessible Area.
- (d) In the event of a closure, pursuant to this Article, III, Declarant shall only close those portions of such areas which must or should reasonably be closed to effectuate the repairs or remediation, shall exercise due diligence in the performance of such repairs or remediation so that it is completed expeditiously and the temporarily closed areas are re-opened to the public promptly, and shall, wherever reasonably possible, perform the needed work in such a manner that the public will continue to have access to the Publicly Accessible Area.
- 3.4 <u>Maintenance and Repair</u>. Declarant shall be solely responsible for the maintenance and repair of the PAA, and all such maintenance shall be performed in a good and worker-like manner (the "<u>PAA Maintenance Obligation</u>"). The PAA Maintenance Obligation shall include but not limited to the following:

(a) <u>Cleaning</u>.

(i) Dirt, litter, and obstructions shall be removed as needed and leaves collected and removed as needed to maintain the Publicly Accessible Area in clean, neat, and good condition.

- (ii) All walkways, lighting, and all other improvements and facilities installed in the Publicly Accessible Area shall be routinely cleaned and maintained so as to keep such improvements and facilities in a clean, neat, and good condition.
- (iii) Graffiti shall be regularly painted over or removed, as appropriate to the nature of the surface, promptly, with reasonable dispatch.
- (iv) Drains, sewers, and catch basins shall be cleaned regularly to prevent clogging.
- (v) Branches and trees damaged or felled by winds, ice, vandalism, or by any other reason whatsoever, shall be promptly removed.
- (vi) Snow and ice shall be promptly removed from all walkways so as not to interfere with safe passage and from all other paved surfaces no more than 24 hours after each snowfall or accumulation of ice.
- (b) <u>Landscape Maintenance</u>. A maintenance program for the planted portions of the PAA shall be established, consisting of a "Spring Start-up Period" program, a "Season Closing Period" program, and a continuing maintenance program through the "Growing Season."
- (i) Spring Start-Up Period: The Spring Start-up Period shall commence on March 1st and terminate not later than the end of the second week of April of each calendar year. The following work shall be undertaken and carried out annually during the Spring Start-up Period:
 - (aa) Remove any winter protectives from trees, shrubs and other planting materials;
 - (bb) Remove all landscaping debris including leaves and dead branches:
 - (cc) Prune and trim trees that have overextended, dead or otherwise unsightly branches to maintain natural form;
 - (dd) Remove or destroy any weeds growing between paving blocks, pavement, and concrete areas;
 - (ee) Apply commercially available nitrogen rich fertilizer to trees, shrubs, planting materials and other lawn areas as appropriate;
 - (ff) Remove any sand deposited as a result of winter sandings;
 - (gg) Replace any plant material or trees that are dead, diseased and/or otherwise unhealthy with specimens of substantially equal type and reasonable size; and
 - (hh) Reseed grassed areas as needed.

- (ii) Season Closing Period: The Season Closing Period shall begin on October 1st and shall terminate not later than November 1st of each calendar year. The following work shall be undertaken and carried out during the Season Closing Period:
 - (aa) Rake and collect leaves;
 - (bb) Wrap trees, shrubs, and other plant materials as necessary to ensure adequate winter protection;
 - (cc) Apply commercially available nitrogen rich fertilizer to all lawn areas; and
 - (dd) Reseed grassed areas as needed.
- (iii) Growing Season: The Growing Season shall commence at the end of the Spring Start-up period and shall terminate at the end of the Season Closing Period. The following work shall be undertaken and carried out during the Growing Season:
 - (aa) Inspect trees on a regular basis and spray when necessary;
 - (aa) Water all trees, shrubs, plantings and grass areas as necessary to maintain in a healthy condition. In extended periods of drought (i.e., little precipitation/high temperatures for more than one week) ground cover, trees, shrubs and other plantings shall be thoroughly watered, subject to City or State regulations governing water usage;
 - (bb) Mow grassed areas on a bi-weekly basis; during periods of excessive growth, mowing shall occur on a weekly basis; reseed grassed areas as needed; and
 - (cc) Weed as needed, no less than on a bi-weekly basis.
- (c) Repairs and Replacements. Declarant shall perform repairs and replacements as needed to maintain the Publicly Accessible Area in state of good repair and in compliance with the specifications set forth in the Site Plan. Declarant shall exercise due diligence in commencing the repair or replacement of same as promptly as possible and completing the same within a reasonably expeditious time after commencement. All repairs and replacements shall be performed in substantial compliance with the specifications set forth in the Site Plan and replacement materials shall match existing materials to the extent feasible. Repairs shall include, but not be limited to, the following:
- (i) Benches or Other Seating: Maintenance, including replacement of any broken or missing slats and painting, as necessary;
- (ii) Walls or Other Barriers: Any broken or materially cracked walls, or barriers shall be repaired or removed and replaced;

- (iii) Paving: All paved surfaces shall be maintained so as to be safe and attractive;
- (iv) Signage: All signs and graphics shall be maintained in good condition and all vandalized or damaged signage shall be promptly cleaned or replaced with new signage or graphics;
- (v) Painting: All items with painted surfaces shall be painted on an "as needed basis." Surfaces shall be scraped free of rust or other extraneous matter and painted to match the installed color;
- (vi) Plant Materials and Trees: Plant materials and trees that are dead, diseased and/or otherwise unhealthy shall be replaced with healthy specimens of substantially equal type and reasonable size; and
- (vii) Construction Defects and Hazardous Conditions: Declarant shall periodically inspect the Publicly Accessible Area for construction defects and hazardous conditions and shall promptly repair and remediate any construction defects or hazardous conditions, as well as implement any safety measures required on an interim basis to protect public safety.
- 3.5 <u>Signage</u>. Pursuant to Local Law 116 of 2018, the Publicly Accessible Area qualifies as a Privately Owned Public Space ("POPS"), and Declarant shall provide signage in accordance with the Site Plan.
- 3.6 <u>Private Easement</u>. Notwithstanding the provisions of this Article, III, the private easement, for the benefit of Peninsula General Nursing Home Corp. dated 3/12/1971 and recorded 3/16/1971 in Reel 465 Page 647, located within the Large-Scale Development Project shall be maintained open and unobstructed as a right of way for vehicular access throughout Development and upon the completion of the Large-Scale Development Project in accordance with the terms of the recorded easement.

ARTICLE IV

PRIVATE STREET NETWORK

4.1 Design and Construction.

- (a) Declarant shall, at its sole cost and expense, construct the Private Street Network, for the purpose of providing public pedestrian and general vehicular access, in accordance with an approved Final Street Network Plan that is consistent with the plans set forth in Exhibit H and which conforms to DOT and FDNY standards.
- (b) Declarant may construct the Private Street Network in five phases as set forth in Exhibit H. If Declarant elects to build the Private Street Network in phases, Declarant shall construct the street segments within each Phase in accordance with the approved Final Street Network Plan for such Phase. As a condition precedent to applying for Building Permits for any

Phase, the Chairperson must certify to DOB that a Final Street Network Plan has been submitted for the segment of the Private Street Network within such Phase (the "<u>Street Phasing Certification</u>").

- (c) Declarant, at its sole cost and expense, shall diligently apply for and prosecute the applications for all City, State, and Federal permits and approvals to fully construct the Private Street Network.
- (d) Declarant shall not accept a Temporary or Permanent Certificate of Occupancy, as set forth in Article VI, for a Building within any Phase until the Chairperson has determined that the segment of the Private Street Network within such Phase is Substantially Complete.
- 4.2 <u>Street Easement</u>. The Private Street Network and any Substantially Completed Street Network Phase shall be accessible to the public at all times, except when required to be closed for repairs and in order to preserve private ownership in accordance with Section 4.7. Declarant grants to the City and general public a permanent, perpetual, and non-exclusive public access easement overall and any portion of the roadbed and sidewalks of the Private Street Network for public pedestrian and general vehicular use, subject to the terms and conditions of this Declaration, which shall be effective from and after Substantial Completion of all or any segment of the Private Street Network (the "<u>Street Easement</u>"). Notice of the of the Street Easement, which form must be approved by DCP, shall be separately recorded against the Subject Property to facilitate awareness of the public's right of way.
- 4.3 <u>Liens</u>. Declarant agrees that liens, including but not limited to judgment liens, mortgage liens, mechanics liens, and vendees' liens, shall be subject and subordinate to the rights, claims, entitlements, interests, and priorities created by the Street Easement.
- 4.4 <u>Use and Floor Area.</u> The Private Street Network, and each segment thereof, shall only be used for the purpose set forth in Section 4.1(a), unless otherwise approved by the City. In no event shall any structure or building be constructed on or above any segment of the Private Street Network, as otherwise shown in the Site Plan. Neither shall any excavation below the Private Street Network be authorized as otherwise shown in the Site Plan and approved by the City, except for the purposes of installing, repairing, or maintaining utilities, water, sewer, and drainage infrastructure. However, the Private Street Network shall not be deemed to reduce the floor area (as defined in the Zoning Resolution) that may be developed on the Subject Property.
- 4.5 Operational Requirements. Declarant shall maintain the Private Street Network in full compliance with approved operational specifications, which shall include but not be limited to traffic control devices and parking regulations (the "Operational Specifications"). As part of the Street Phasing Certification, the Chairperson, in consultation with DOT, shall approve the Operational Specifications for the subject Street Network Phase. To ensure compliance with the Operational Specifications, Declarant's management company shall monitor and inspect daily all improved segments of the Private Street Network, for unlawful parking and cleanliness. The Declarant shall also retain a towing company to promptly remove any illegally parked vehicles. As a condition precedent to the issuance and acceptance of a TCO for any Building, Declarant shall execute agreements with a management company and a towing company, and the services

provided in the agreements shall commence no later than the date on which the first TCO for the Large-Scale Development Project is issued.

- 4.6 <u>Street Maintenance Obligation</u>. Declarant shall be solely responsible for monitoring the use of the Private Street Network and maintaining all Substantially Completed segments of the Private Street Network in a state of good repair and cleanliness (the "<u>Street Maintenance Obligation</u>"). The Street Maintenance Obligation shall include but not be limited to the following:
 - (a) Maintaining the paved surfaces in good repair;
 - (b) Monitoring street usage and compliance with the Operational Specifications;
 - (c) Maintaining streetlights;
- (d) Clearing snow from the street at such times as the accumulated snowfall in any 12-hour period exceeds two inches;
- (e) Maintaining any required storm and sanitary drainage systems in a clear, workable and efficient manner;
- (f) Maintaining signage indicating that the Private Street Network is privately owned and maintained; and
- (g) Maintaining the traffic control devices and road markings as specified in the Operational Specifications.
- Closure of Private Street Network. Notwithstanding Section 4.2, Declarant may close all or any portion of the Private Street Network either: (i) for the repair, restoration, rehabilitation, renovation or replacement of pipes, utility lines or conduits, or other equipment on or under Private Street Network or for the repair thereof; (ii) as may be approved by DOB or DOT in connection with work on any of the buildings in the Large-Scale Development Project; or (iii) in the event of an emergency or hazardous condition. Except in cases of emergency, Declarant shall provide seven (7) Business Days prior notice to the public, DCP, and DOT of the temporary closure of any portion of the Private Street Network, by posting signs at appropriate locations. In cases of emergency, Declarant shall provide such public notice as soon as practicable and shall promptly, but in no event more than two (2) Business Days after such closure, give notice to DOT and DCP that such portion has been closed, which notice shall describe the nature of the emergency or hazardous condition causing the closure, the portion to be closed and the anticipated duration thereof. Emergency conditions, for which the Private Street Network may be closed pursuant to (iii) above, shall be limited to actual or imminent emergency situations, including but not limited to, security alerts, riots, casualties, disasters, or other events engendering public health, safety or property, provided that no such closure shall continue for more than twelve (12) consecutive hours without Declarant having consulted with DOT, and any direction by the New York City Police Department ("NYPD") or DOB regarding the emergency situation. Declarant will close or permit to be closed only those portions of the Private Street Network which must or should reasonably be closed to effect the repairs, replacements or mitigation of hazardous site conditions to be undertaken, pursuant to (i), (ii), and (iii), above and will exercise due diligence in the performance

of such repairs, replacements, or mitigation such that they are completed expeditiously and the temporarily closed areas (or any portions thereof) are re-opened to the public promptly. Notwithstanding other limitations contained in the Declaration, Declarant shall have the right to close all or any portions of the Private Street Network to the City and the general public one (1) Business Day in each year to preserve its ownership interest in such portions of the Private Street Network.

- Maintenance and Access Security. Declarant shall, as a condition precedent to 4.8 accepting a Temporary Certificate of Occupancy, as set forth in Article VI, either: (i) post or cause the posting with DOT a performance bond for the benefit of the City, in a form reasonably satisfactory to DOT and issued by a surety company licensed to do business in the State of New York; (ii) deposit with DOT one or more clean, irrevocable letters of credit, naming the City as beneficiary, in a form reasonably satisfactory to the City; or (iii) deliver to DOT other security reasonably acceptable to the City (the "Maintenance Security"). The Maintenance Security shall be in an amount set at one hundred and twenty-five percent (125%) of an amount equal to the assessed cost, over a two-year period, of maintaining and performing the needed capital repairs for the full width and length of the street segments of the Private Street Network within Substantially Completed segment of the Private Street Network (the "Assessed Maintenance Cost"). The Assessed Maintenance Cost shall be adjusted every five years to reflect changes in the CPI. The Assessed Maintenance Cost, which includes but is not limited to street cleaning, pothole filling, drainage repair, sidewalk and signage repair, tree care, curb replacements, sidewalk reconstruction, roadway repaving, and replacement of both posted and painted signage, shall be submitted to DCP for approval. If the Maintenance Security is not renewed at least thirty (30) calendar days prior to the expiration date of such Maintenance Security or if the City is informed in any way that the instrument of the Maintenance Security will be cancelled or will not be renewed, the City shall be entitled to draw down on the full amount of the Maintenance Security for the purpose of maintaining the Private Street Network. Declarant shall promptly replenish any sums drawn down and used by the City for the maintenance of the Private Street Network so that the available Maintenance Security remains at the amount set herein.
- 4.9 <u>Right of Inspection</u>. During construction of the Private Street Network, the City upon one (1) Business Day's prior notice, shall have the right to inspect and determine whether the Private Street Network is being constructed in substantial conformance with the approved Final Street Network Plan.
- 4.10 <u>Right of Access</u>. The City, acting through DOT, in addition to any rights or remedies contained in Article VII of the Declaration, shall have the right to access the Private Street Network at all times the Private Street Network is open and at other times upon reasonable notice, for the purpose of inspecting the Private Street Network and determining Declarant's compliance with the terms of this Declaration or for any lawful purpose.
- 4.11 <u>Right of Access for Repairs</u>. The Declarant hereby grants the City, its agents, or its contractors, the right, upon five (5) Business Day's prior written notice, to enter the Private Street Network upon a Maintenance Default to perform Declarant's Street Maintenance Obligation until such time as Declarant resumes the performance of such obligations. A "Maintenance Default" shall occur if Declarant has failed to perform any of its obligations relating to the Private Street Network under the Declaration, and if upon notice, in accordance with this Declaration,

Declarant has not commenced and diligently prosecuted efforts to effect a cure as provided in Article VII.

- 4.12 <u>Assumption of Liability</u>. Declarant shall assume all costs and liabilities arising from the construction, improvement, maintenance, and repair of the Private Street Network including, without limitation, all liabilities arising from the size, location, grade, design and configuration of the Private Street Network. Actions taken by the City pursuant to this Article, IV, shall not constitute an assumption of responsibility or be construed as an implied or express acceptance by the City of the Private Street Network or any segment thereof under either the General City Law or any other law; upon the City exercising its Right to Repair set forth in Section 4.11, Declarant shall promptly resume its Street Maintenance Obligations.
- 4.13 <u>Waiver</u>. Declarant waives any right, cause of action or claim for damages against the City or any of its subdivisions, whether such right, cause of action or claim arises from the use of the Private Street Network in accordance with the terms of this Declaration or otherwise.

ARTICLE V

ENVIRONMENTAL PROTECTION MEASURES

Declarant shall implement the following Project Components Related to the Environment ("<u>PCREs</u>") and Mitigation Measures in accordance with the FEIS and as further set forth in this Article, V; such PCREs and Mitigation Measures may be modified in accordance with the provisions of Section 5.7.

- 5.1 Project Component Related to the Environment.
- (a) <u>Construction Noise</u>. To avoid or minimize increases in noise levels at sensitive receptors in relation to construction of the Large-Scale Development Project, the Declarant shall implement the PCREs related to noise, as indicated on **Exhibit F-1**.
- (b) <u>Construction Air Quality</u>. To avoid or minimize pollutant emissions in relation to construction of the Large-Scale Development Project, the Declarant shall implement the PCREs related to air quality, as indicated on **Exhibit F-1**.
- (c) <u>Construction Transportation</u>. To avoid or minimize congestion and maintain pedestrian and vehicle safety in relation to construction of the Large-Scale Development Project, the Declarant shall prepare a plan of the proposed staging areas, temporary lane and sidewalk alterations on public streets, the duration such alterations will be implemented, the width and length of affected street segments, sidewalk protection measures for pedestrians on public streets, the location of flaggers to control trucks entering the Subject Property, and temporary fencing consistent with the FEIS (the "Maintenance and Protection of Traffic Plan" or "MPT"). Declarant shall submit a MPT to DOT for review and approval, provided, however, that completion and submission of the MPT shall not be necessary for preliminary site work, unless DOT advises Declarant that a MPT is required.

(d) <u>Timing of the PCREs</u>. The implementation of the construction PCREs must be confirmed by the Monitor, as defined herein, prior to Declarant accepting Building Permits for a Phase. The Monitor shall separately confirm the implementation of the MPT prior to the start of construction for any Phase.

(e) Appointment and Role of Independent Environmental Monitor.

- (i) Declarant shall, with the consent of DCP, retain a third party (the "Monitor") to oversee, on behalf of DCP, the implementation and performance by Declarant of the construction period PCREs and Mitigation Measures set forth in this Article, V, and Exhibits F1-F2. The Monitor shall be a licensed engineer, architect, general contractor or environmental consultant with significant experience in environmental management and construction management (or multiple persons or a firm employing such persons), including familiarity with the means and methods for implementing of the PCREs and Mitigation Measures.
- The "Scope of Services" described in any agreement between Declarant and (ii) the Monitor pursuant to which the Monitor is retained (the "Monitor Agreement") shall be subject to prior review by and approval of DCP, such approval not to be unreasonably withheld, conditioned, or delayed. Such agreement shall include provisions in a form acceptable to DCP that, among others, shall: (i) ensure that the Monitor is independent of Declarant in all respects relating to the Monitor's responsibilities under this Declaration (provided that the Monitor shall be responsible to Declarant with regard to practices generally applicable to or expected of consultants and independent contractors of Declarant) and has a duty of loyalty to DCP; (ii) provide for appropriate DCP management and control of the performance of services by the Monitor; (iii) authorize DCP to direct the termination of services by the Monitor for unsatisfactory performance of its responsibilities under the Monitor Agreement; (iv) allow for the retention by the Monitor of sub-consultants with expertise appropriate to assisting the Monitor in its performance of its obligations to the extent reasonably necessary to perform its obligations under this Declaration and the Monitor Agreement; and (v) allow for termination by Declarant for cause, but only with the express written concurrence of DCP, which concurrence shall not be unreasonably withheld or delayed. If DCP fails to act upon a proposed Monitor Agreement within thirty (30) days after submission of a draft form of Monitor Agreement, the form of Monitor Agreement submitted shall be deemed acceptable by DCP and may be executed by Declarant and the Monitor. The Monitor Agreement shall provide for the commencement of services by the Monitor prior to any demolition or excavation on the Subject Property and shall continue in effect at all times that construction activities are occurring on the Subject Property with respect to any Phase, until the issuance of temporary or permanent certificate of occupancies.
- (iii) The Monitor shall: (i) assist and advise DCP with regard to the review of plans and measures proposed by Declarant for purposes of satisfying PCREs and Mitigation Measures; and (ii) provide reports of Declarant's compliance with the PCREs and Mitigation Measures during any period of construction. The Monitor may at any time also provide Declarant and DCP with notice of a determination that a PCRE or Mitigation Measure has not been implemented, accompanied by supporting documentation establishing the basis for such determination, provided that any such notice shall be delivered to both parties. The Monitor shall: (x) have full access to the portion of the Subject Property then being developed (as referenced in the Monitor Agreement), subject to compliance with all generally applicable site safety

requirements imposed by law, pursuant to construction contracts, or imposed as part of the site safety protocol in effect for the Subject Property; (y) on reasonable notice and during normal business hours, be provided with access to all books and records of Declarant pertaining to the applicable portion of the Subject Property either on or outside the Subject Property which it reasonably deems necessary to carry out its duties, including the preparation of periodic reports; and (z) be entitled to conduct any tests on the Subject Property that the Monitor reasonably deems necessary to verify Declarant's implementation and performance of the PCREs and Mitigation Measures, subject to compliance with all generally applicable site safety requirements imposed by law, site operations, or pursuant to construction contracts in effect for the Subject Property and provided further that any such additional testing shall be coordinated with Declarant's construction activities and use of the Subject Property by the occupants of and visitors to any of the Buildings and conducted in a manner that will minimize any interference with the Development. The Monitor Agreement shall provide that Declarant shall have the right to require Monitor to secure insurance customary for such activity and may hold the Monitor liable for any damage or harm resulting from such testing activities.

(iv) Declarant shall be responsible for payment of all fees and expenses due to the Monitor (including fees and expenses paid to sub-consultants engaged pursuant to this Section, 5.1(e), in accordance with the terms of the Monitoring Agreement.

5.2 Environmental Mitigation Relating to Open Space.

- (a) The FEIS has identified a significant adverse impact to active open space that occurs upon occupancy of more than 1,244 residential units and that in order to partially mitigate the impact the Declarant shall provide the Open Space Mitigation Measure, as defined herein, in order to increase the active open space in the study area.
- (b) The Open Space Mitigation Measure shall consist either: (i) under direction and with approval from DPR providing active recreation improvements to 1.67 acres of Rockaway Community Park, consistent with the 2014 Rockaway Parks Conceptual Plan; such active recreation improvements will consist of, but are not limited to, tennis courts, basketball courts, handball courts, and/or ballfields; or (ii) under direction and with approval from DPR, providing 1.67 acres of new publicly accessible active open space, in consultation with DCP, within the open space study area to serve the proposed population and offset the proposed project's impact on existing active open space (the "Open Space Mitigation Measure").
- (c) Declarant shall give DPR at least two (2) months' notice prior to applying for Building Permits for any Building within Phase 4 of the Large-Scale Development Project as designated on Exhibit G, which together with all Development that has previously received a TCO or PCO, would result in the occupancy of more than 1,244 residential units on the Subject Property (the "Open Space Mitigation Notice"). The Open Space Mitigation Notice shall include a proposed plan, developed by the Declarant in consultation with DPR, which demonstrates the proposed measure to mitigate the impact substantially in accordance with the FEIS and Section 5.2(b). Within thirty (30) days of receiving the Open Space Mitigation Notice, DPR shall certify whether the plan submitted in connection with the Open Space Mitigation Notice achieves the partial mitigation of active open space substantially in accordance with the FEIS and shall notify the Declarant whether DPR accepts or reject the Open Space Mitigation Measure.

- (d) Declarant shall not accept and DOB shall not issue a TCO or PCO, pursuant to Article VI, for any portion of the Large-Scale Development Project, which together would result in the occupancy of more than 1,244 new residential units on the Subject Property, unless and until DPR certifies to DOB that the Declarant has satisfied the Open Space Mitigation Measure. If funding is the selected mitigation, such funds must be provided thirty (30) days prior to the Declarant's acceptance of a TCO or PCO for more than 1,244 dwelling units.
- (e) Notwithstanding anything to the contrary, if Declarant demonstrates to the satisfaction of the Chairperson that DPR failed to respond to the Open Space Mitigation Notice or certify the Open Space Mitigation Measure selected in accordance with the time periods set forth in Section 5.2(c), then Declarant shall have no further obligations under this Declaration or the FEIS relating to the Open Space Mitigation Measure.

5.3 Environmental Mitigation Relating to Community Facilities.

- (a) The FEIS identified a significant adverse impact to public schools that occurs upon occupancy of more than 910 new non-senior residential units and that in order to fully mitigate the impact the Declarant shall, subject to the provisions set forth herein, increase the elementary school capacity in the study area by 162 seats and intermediate school capacity in the study area by 57 seats (the "School Mitigation Measure").
- (b) The School Mitigation Measures shall consist of accommodating an increase in school capacity by up to 162 public elementary and 57 public intermediate school seats at school(s) in the school study area where such capacity increase is warranted (as determined by DOE/SCA) either by: (i) providing funding to the DOE/SCA; (ii) performing work in accordance with SCA specifications and procurement processes; or (iii) in accordance with DOE/SCA approval, provide off-site land and/or fit-out annex space (up to core and shell).
- (c) Prior to implementation of the School Mitigation Measure, the Declarant may conduct an additional analysis, in accordance with the CEQR Technical Manual guidelines, to determine whether, based on the data available at the time of the additional analysis, the extent of the impacts and/or timing of when the impacts on public elementary and intermediate schools are projected to occur varies from that which had been identified in the FEIS. Such additional analysis shall be submitted by Declarant to the SCA, DOE, and DCP for review prior to DOB issuing a TCO or PCO for the first Building constructed within Phase 2, as set forth in Exhibit G. If the SCA and DOE, in consultation with DCP, as lead agency, determine, based on the additional analysis submitted by Declarant, that the extent of the impacts and/or timing of when the impacts are projected to occur varies from that set forth in the FEIS, the School Mitigation Measure shall be adjusted, up or down, accordingly.
- (d) Declarant shall give DOE and the SCA (the DOE and the SCA shall collectively be referred to as the "<u>School Mitigation Interested Parties</u>") at least two (2) months' advance notice of Declarant's intention to apply for the first Building Permit for Phase 3 of the Large-Scale Development Project as designated on Exhibit G, which, together with all Development that has previously receive a TCO or PCO, would result in the occupancy of more than 910 non-senior residential units on the Subject Property (the "<u>School Mitigation Notice</u>"). The School Mitigation Notice shall include a proposed plan, developed by the Declarant in consultation with the School

Mitigation Interested Parties, which demonstrates the proposed measure to mitigate the impact substantially in accordance with the FEIS and Section 5.3(d). Within thirty (30) days of receiving the School Mitigation Notice, subject to DCP concurrence, the School Mitigation Interested Parties shall certify whether the plan submitted in connection with the School Mitigation Notice achieves the full mitigation of school seats substantially in accordance with the FEIS and shall notify the Declarant whether they accept or reject the School Mitigation Measure.

- (e) Declarant shall not accept and DOB shall not issue a TCO or PCO, pursuant to Article VI, for any portion of the Large-Scale Development Project, which together with all Development that has previously received a TCO or PCO, would result in the occupancy of more than 910 new residential units for purposes of the required the elementary school mitigation and 1,030 new residential units for purposes of the required the intermediary school mitigation, unless and until the School Mitigation Interested Parties certify to DOB that the Declarant has satisfied the School Mitigation Measure. If funding is the selected mitigation, such funds must be provided prior to the Declarant's acceptance of a TCO or PCO for more than 910 dwelling units.
- (f) Notwithstanding anything to the contrary, if Declarant demonstrates to the satisfaction of the Chairperson that the School Mitigation Parties failed to respond to the School Mitigation Notice or certify the School Mitigation Measure selected in accordance with the time periods set forth in Section 5.3(c), then Declarant shall have no further obligations under this Declaration or the FEIS relating to the School Mitigation Measure.

5.4 <u>Environmental Mitigation Relating to Operational and Construction Transportation.</u>

- (a) Chapter 20 of the FEIS identifies mitigation measures to operational transportation impacts, including traffic and pedestrians, and construction-period traffic impacts, which measures are anticipated to be implemented in connection with each Phase of the Large-Scale Development Project, as more particularly set forth in Exhibit F2. To implement the Mitigation Measures, it will be necessary to determine: (i) to what extent the FEIS-identified traffic, pedestrian, and construction-traffic mitigation measures (collectively referred as the "Transportation Mitigation Measures") will be needed and implemented as the Buildings within each Phase are constructed and occupied; and (ii) the need for any DOT-approved adjustments to the FEIS-identified traffic and pedestrian mitigation measures based on current conditions ("Adjustments"), provided that any such Adjustments shall be commensurate with the intended effects of the Transportation Mitigation Measures and shall be the most cost-effective measures available to reasonably achieve such intended effects.
- (b) Declarant shall not apply for and shall not accept a Building Permit regarding a Phase, unless and until DOT has approved a Scope Work for the Transportation Mitigation Measures, including any Adjustments, regarding: (i) the installation of new signals at the intersections identified in connection with the Phase, including the installation of required conduits; (ii) restriping and other changes to lane geometries; and (iii) signal timing changes, in connection with completed Phases as well as during construction of each Phase, all as set forth in Exhibit F2 (the "Scope of Work"). Upon approval by DOT of a Scope of Work, the Chairperson shall certify to DOB that the Scope of Work has been approved by DOT.

- (c) DOT shall make reasonable efforts to approve the Scope of Work within 90 days of receipt of proposed Scope of Work. In reviewing the Scope of Work, to the extent that DOT deems unnecessary one or more specific Transportation Mitigation Measures in a particular Phase, Declarant shall continue to be obligated to include such measures in the Scope of Work for each subsequent Phase, unless otherwise determined by DOT.
- (d) Within thirty (30) days of the completion and seventy percent (80%) occupancy of all the Buildings within a Phase, Declarant shall notify DOT of Declarant's intent to perform the Transportation Mitigation Measures set forth in the Scope of Work. Upon receipt of Declarant's notice to proceed with the implementation of the Scope of Work, DOT shall determine the needed warrant analysis to confirm the implementation of the Transportation Mitigation Measures set forth in the Scope of Work. To the extent that a traffic control device is specified in a Builders Pavement Plan ("BPP"), DOT shall certify to DOB that Declarant cannot install such a device until a warrant analysis is performed and that DOB may issue a TCO, pending the completion of the needed warrant analysis if Declarant has complied with all other aspects of the BPP.
- (e) Declarant shall either pay DOT or the City for the ordinary and customary costs, if any, of improvements associated with the needed Transportation Mitigation Measures and/or Adjustments (including but not limited to the costs associated with the design, installation of the traffic signal(s), proposed geometric modifications, traffic signs and pavement markings removals/installations) or, if so directed by DOT, shall in good order implement such measures itself under DOT's direction and in compliance with DOT requirements for implementation. Declarant shall submit all the required drawings/designs as per AASHTO and DOT specifications to DOT for its review and approval and to other City agencies, as appropriate. Any and all signal work shall be performed by an electrical contractor approved by DOT.
- 5.5 <u>Environmental Mitigation Relating to Operational Air Quality</u>. Declarant shall comply with the Operational Air Quality Mitigation Measures and timing of those measures set forth in Exhibit F-2 (the "**Operational Air Quality Mitigation Measure**").
- 5.6 <u>Environmental Mitigation Relating to Construction Noise</u>. Declarant shall comply with the Construction Noise Mitigation Measures and timing of those measures set forth in Exhibit F-2 (the "<u>Construction Noise Mitigation Measure</u>").
- 5.7 Force Majeure Involving a PCRE or Mitigation Measure. Notwithstanding any provision of this Declaration to the contrary, if Declarant is unable to perform a PCRE or Mitigation Measure required by the FEIS by reason of the occurrence of a Force Majeure Event, as determined by the Chairperson, pursuant to the procedures set forth in Section 6.3, then Declarant shall not be excused from performing such PCRE or Mitigation Measure that is affected by Force Majeure Event unless and until the Chairperson has made a determination in his or her reasonable discretion that the failure to implement the mitigation measure during the period of the Force Majeure Event, or implementing an alternative proposed by Declarant, would not result in any new or different significant environmental impact not addressed in the FEIS.
- 5.8 <u>Incorporation of FEIS Mitigation Measures</u>. If this Declaration inadvertently fails to incorporate a PCRE or Mitigation Measure set forth in the FEIS, such PCRE or Mitigation Measure shall be deemed incorporated herein by reference. If there is any inconsistency between

the PCRE or Mitigation Measure set forth in the FEIS and in this Declaration, the PCRE or Mitigation Measure set forth in the FEIS shall be applicable.

- 5.9 <u>Innovation; Alternatives; Modifications Based on Further Assessments.</u>
- (a) <u>Innovation and Alternatives</u>. In complying with any PCRE or Mitigation Measure set forth in this Article, V, or in the FEIS, Declarant may, at its election, implement innovations, technologies, or alternatives that are or hereafter become available, which Declarant demonstrates to the reasonable satisfaction of DCP would result in equal or better methods of achieving the relevant PCRE or Mitigation Measure than those set forth in this Declaration (the "<u>Alternative Mitigation Measures</u>").
- Modifications Based on Further Assessments. Where Declarant believes, in good (b) faith, based on changed conditions, that a PCRE or Mitigation Measure required under this Declaration should not apply or be modified without diminishment of the environmental standards that would be achieved by implementation of the PCRE or Mitigation Measure ("Modification of Mitigation Measure"), it shall set forth the basis for such belief in an analysis submitted to DCP. In the event that DCP determines that the relevant PCRE or Mitigation Measure should not apply or could be modified, Declarant may eliminate or modify the PCRE or Mitigation Measure consistent with the DCP determination, provided that Declarant records a notice of such change against the Subject Property in the Registers Office. Following the delivery of a notice to DCP requesting an Alternative Mitigation Measure or a Modification of a Mitigation Measure pursuant to this Section 5.9 (the "Section 5.9 Request"), Declarant shall meet with DCP (and at DCP's option, the Monitor) to respond to any questions or comments on such request and accompanying materials and shall provide additional information as may reasonably be requested by DCP or the Monitor in writing in order to allow DCP to determine, acting in consultation with the Monitor and City agency personnel as necessary in relation to the subject matter of the Section 5.9 Request.

ARTICLE VI

CERTIFICATES OF OCCUPANCY

- 6.1 Temporary Certificates of Occupancy.
- (a) Declarant shall not accept a TCO for any Building until the following conditions are met:
- (i) DCP has issued a Notice of Substantial Completion for the segments of the Private Street Network within the Phase that the Building is located and Declarant has provided the Maintenance Security and proof of the executed management and towing contracts in accordance with Article IV: and
- (ii) The applicable Mitigation Measures set forth in Article V are determined to have been completed by the Chairperson, except if the implementation of such Mitigation Measures requires further analysis upon occupancy of such Building, in which case Declarant must have fulfilled its obligations to request the implementation of the Mitigation Measure.

- (b) Declarant shall not accept a TCO for Building E2 in Phase 4, as shown in Exhibit G, until DCP has issued a Notice of Substantial Completion for the PAA. The TCO for Building E2 shall reflect the PAA space for the Large-Scale Development Project
- (c) The TCO for any Building shall reflect the segments of the Private Street Network that must be provided within the Phase that the Building is located.
- (d) In the event that due to a Force Majeure Event Declarant has obtained a TCO, prior to completion of the conditions set forth in Section 6.1(a), upon cessation of the Force Majeure Event Declarant shall, as promptly as possible, satisfy the conditions of Section 6.1(a)

6.2 <u>Notice of Substantial Completion</u>.

- (a) <u>Notification</u>. Declarant shall notify the Chairperson at such time as it believes that the PAA, if applicable, and segments of the Private Street Network for a Phase is Substantially Complete and shall request that the Chairperson certify to DOB the Substantial Completion of the PAA and segments of the Private Street Network.
- (b) <u>Initial Review</u>. No later than thirty (30) days after the receipt of the notification set forth in Section 6.2(a), the Chairperson shall either: (i) issue a Notice of Substantial Completion; or (ii) deliver to Declarant written notice setting forth the reasons why the PAA, if applicable, and segments of the Private Street Network within the Phase are not Substantially Complete and the items that need to be completed in order to determine that the PAA and segments of the Private Street Network within the Phase are Substantially Complete.
- (c) <u>Subsequent Review</u>. Upon completing the outstanding work specified by the Chairperson based on the Initial Review, Declarant shall notify the Chairperson of such completion. No later than twenty (20) calendar days of the receipt of such notice, the Chairperson shall either: (i) issue a Notice of Substantial Completion; or (ii) notify Declarant in writing of items that have not been completed or satisfactorily performed. This process shall continue until the Chairperson has issued a Notice of Substantial Completion.
- 6.3 Force Majeure. In the event that Declarant is unable to Substantially Complete the PAA or segment of the Private Street Network by the time a Building is ready for a TCO, as a result of a Force Majeure Event, then Declarant shall notify DCP as soon as Declarant learns of such circumstances. Declarant's written notice (the "Delay Notice") shall include a description of the condition or event, its cause and probable duration (if known to Declarant), and in Declarant's reasonable judgment, the impact it is reasonably anticipated to have on the completion of the item of work. The Chairperson shall, within ten (10) calendar days of its receipt of the Delay Notice, either: (i) certify in writing that a Force Majeure Event has occurred; or (ii) notify Declarant that it does not reasonably believe a Force Majeure Event has occurred, in which case the Chairperson shall state with particularity the reasons it believes Force Majeure has not occurred. Such certification or notice shall constitute a final determination. Upon a determination that a Force Majeure Event has occurred, the Chairperson shall grant Declarant appropriate relief for such delay, including certifying in writing to the DOB that the Chairperson has no objection to the issuance of a TCO for all or part of the building(s) in the Large-Scale Development Project. Any delay caused as the result of a Force Majeure Event shall be deemed to continue only as long as

the Force Majeure Event is continuing. Upon cessation of the events causing such delay, the Declarant shall promptly recommence the PAA Work. As a condition of granting such relief, DCP may require that Declarant post a bond or other security in a form and amount reasonably acceptable to DCP in order to ensure that the PAA Work is Substantially Completed and that all other requirements of Section 6.1 or 7.1, as applicable, are satisfied. Such security shall be in a sum equal to 150% of the cost of the remaining work in order to Finally Complete such Publicly Accessible Area. Such estimated cost is subject to the reasonable approval of DCP. Declarant shall be obligated to Substantially Complete or Finally Complete construction within the period of time specified in the Delay Notice, or such lesser period of time as DCP reasonably determined in the Delay Notice; provided, however, that if the Force Majeure Event has a longer duration than as set forth in the Delay Notice or as reasonably determined by DCP, DCP may grant additional time for Substantial Completion or Final Completion, as the case may be.

- 6.4 <u>Permanent Certificates of Occupancy</u>. Declarant shall not accept a Permanent Certificate of Occupancy from DOB for any building on the Subject Property until the Chairperson certifies to Declarant and DOB that the PAA, if applicable, and segments of the Private Street Network, within the Phase corresponding to building Declarant is seeking a Permanent Certificate of Occupancy is Finally Complete, subject to the following provisions:
- (a) <u>Notification</u>. Declarant shall notify the Chairperson at such time as it believes that the PAA, if applicable, and segments of the Private Street Network within a Phase are Finally Complete and shall request that the Chairperson certify to DOB the Final Completion of the PAA and segments of the Private Street Network within such Phase.
- (b) <u>Initial Review</u>. No later than thirty (30) days after the receipt of the notification set forth in Section 6.4, the Chairperson shall either: (A) issue a Notice of Final Completion; or (B) deliver to Declarant written notice setting forth the reasons why the PAA and segments of the Private Street Network are not Finally Complete and the items that need to be completed in order to determine that the PAA and segments of the Private Street Network within the Phase are Finally Complete.
- (c) <u>Subsequent Review</u>. Upon completing the outstanding work specified by the Chairperson to achieve Final Completion, Declarant shall notify the Chairperson of such completion. No later than twenty (20) calendar days of receipt of such notice, the Chairperson shall either: (A) issue a Notice of Final Completion; or (B) notify Declarant in writing of items that have not been completed or satisfactorily performed. This process shall continue until the Chairperson has issued a Notice of Final Completion.
- (d) The applicable Mitigation Measures set forth in Article V are determined to have been completed by the Chairperson, except if the implementation of such Mitigation Measures requires further analysis upon occupancy of such Building, in which case Declarant must have fulfilled its obligations to request the implementation of the Mitigation Measure.

ARTICLE VII

ENFORCEMENT

7.1 Enforcement by the City.

- (a) Declarant acknowledges that the City is an interested party to this Declaration, and consents to enforcement by the City, administratively or at law or equity, of the restrictions, covenants, easements, obligations and agreements contained herein.
- (b) No person or entity other than the Declarant, its successors and assigns, or the City shall have any right to enforce the provisions of this Declaration. Nothing contained herein shall be deemed to allow any other person or entity, public or private, any interest or right of enforcement of any provision of this Declaration or in any attached exhibits, including any claim by any public or private landowner to be the beneficiary of an easement appurtenant to lands adjoining the Subject Property which could or might be affected by the enforcement of the provisions of this Declaration with respect to the Subject Property.
- (c) In the event of a failure to cure, upon notice, a breach of the restrictions, covenants, easements, obligations, and agreements herein, the City shall have the right to pursue enforcement of this Declaration and seek any remedy herein provided against any party that constitutes a Declarant at the time in question, including any Association Members as defined in Section 10.3, subject to the limitations set forth in Section 10.5(f).
- 7.2 <u>Additional Remedies</u>. Declarant also acknowledges that the remedies set forth in this Declaration are not exclusive, and that the City and any agency thereof may pursue other remedies not specifically set forth herein.
- 7.3 <u>Enforcement by Declarant</u>. Declarant shall have the right to pursue any remedy available to Declarant, both administrative and judicial and at law and equity, to enforce its rights under this Declaration.

7.4 Notice and Cure.

- (a) If Declarant fails to perform any of its obligations under this Declaration, the City shall give Declarant and Mortgagee thirty (30) days written notice of such alleged default, during which period Declarant shall have the opportunity to cure the noncompliance. If the City finds that Declarant has diligently commenced and prosecuted efforts to cure the noncompliance during such thirty (30) day period, then the aforesaid thirty (30) day period shall be extended for so long as Declarant continues to proceed diligently with curing the noncompliance.
- (b) Notwithstanding the foregoing, in the event of a denial of public access of an ongoing nature or interference on a continuing basis with the City's rights pursuant to this Declaration, the City may immediately exercise any and all of its rights hereunder, including but not limited to the issuance of violations and seeking a mandatory injunction; in such an event, the notice and cure provisions of this Section, 7.4(a), shall not apply.

7.5 Default.

- (a) In the event that any alleged default has not been cured or there is a denial of public access as set forth in this Article, the City shall be entitled to draw down on the Maintenance Security and to apply such monies to cure such defaults and shall be entitled to any other remedy available in law or at equity. If the City has drawn down the Maintenance Security, Declarant shall within five (5) business days post a new Maintenance Security replacing the Maintenance Security drawn down by the City. Nothing set forth herein shall prevent the City from curing a default, in accordance with this Article, prior to drawing down on the Maintenance Security and then drawing down on the Maintenance Security to the extent necessary to reimburse or otherwise pay for such expenses incurred.
- (b) If, upon or after Substantial Completion of the Private Street Network, Owner fails to timely post and/or maintain the Maintenance Security and such failure has not been cured as set forth in this Article, then, in addition to any other remedy provided herein, Declarant shall be liable to the City for liquidated damages in the cash amount of \$1000 per day for every calendar day or part thereof that a failure to provide the Maintenance Security continues. Declarant acknowledges that: (i) said cash amount is reasonable in proportion to the probable damages likely to be sustained by the City if Declarant fails to comply with the terms of this Declaration; (ii) the amount of actual damages to be sustained by the City in the event of such failure is incapable of precise estimation; and (iii) such payment does not constitute a penalty or punitive damages for any purposes.

ARTICLE VIII MISCELLANEOUS

8.1 <u>Representation</u>. Declarant hereby represents and warrants that there is no restriction of record on the development, enlargement, or use of the Subject Property, nor any present or presently existing estate or interest in the Subject Property, nor any existing lien, obligation, covenant, easement, limitation, or encumbrance of any kind that shall preclude the restriction and obligation to develop and enlarge the Subject Property as a large-scale general development as set forth herein.

8.2 Binding Nature; Successors and Assigns.

- (a) The provisions of this Declaration shall be covenants running with the land and shall inure to the benefit of and be binding upon the respective heirs, successors, legal representatives and assigns of Declarant to the Subject Property, including Mortgagee, and references to Declarant shall be deemed to include such heirs, successors, legal representatives and assigns as well as the successors to their interests in the Subject Property, subject to the further provisions of this Section, 8.2. Reference in this Declaration to agencies or instrumentalities of the City shall be deemed to include agencies or instrumentalities succeeding to jurisdiction thereof pursuant to the laws of the State of New York and the New York City Charter.
- (b) Notwithstanding anything to the contrary contained in this Declaration, in the event that any building in the Large-Scale Development Project is converted to condominium or cooperative corporation forms of ownership, the Coop/Condominium, as defined herein, and any Unit Interested Party (except that where the Declarant or any successor in interest to Declarant is

also a Unit Interested Party, it shall remain obligated as Declarant pursuant to the provisions of this Declaration) shall not have any obligations under this Declaration to construct the Public Access Area and Private Street Network.

- (c) Notwithstanding the provisions of Section 8.2(b), in the event that a temporary or permanent certificate of occupancy has been issued for any portion of the Large-Scale Development Project prior to the receipt of a Notice of Substantial or Final Completion due to Force Majeure, the Declarant that developed such portion of the Large-Scale Development Project that was allowed to proceed due to the Force Majeure event shall remain obligated as Declarant until a Notice of Final Completion has been issued.
- 8.3 <u>Parties in Interest.</u> As of the date hereof, the Title Company has determined that there has been no change in the certification attached as Exhibit B and Declarant represents and warrants that the Parties in Interest listed in Exhibit B are the only known Parties in Interest in the Subject Property as of the date hereof.
- 8.4 <u>Counterparts</u>. This Declaration may be executed in one or more counterparts, each of which shall be an original and all of which, together, shall constitute one agreement.
- 8.5 <u>Recordation</u>. Declarant shall File and record this Declaration in the Office of the City Register of the City of New York (the "<u>Register's Office</u>"), indexing it against the Subject Property within five (5) business days of the New York City Council's approval or approval with modifications of the Land Use Applications by an affirmative vote or by operation of law as set forth in New York City Charter Section 197-d (such date hereinafter referred to as the "<u>Recording Date</u>"). Declarant shall promptly provide to the Chairperson of the CPC a copy of the Declaration as recorded, so certified by the City Register. If Declarant fails to so record this Declaration by the Recording Date, CPC may record a duplicate original of this Declaration, but all costs of recording, whether undertaken by Declarant or by CPC, shall be borne by Declarant.
- 8.6 <u>Effective Date.</u> This Declaration and the provisions and covenants hereof shall become effective as of the date of recordation of this Declaration in accordance with Section 8.5 above.

8.7 Notice.

- (a) All notices, demands, requests, consents, approvals, and other communications (collectively referred to as "<u>Notice</u>") which may be or are permitted, desirable, or required to be given under this Declaration shall be in writing and shall be sent or delivered as follows:
 - (i) if to Declarant: to the address at the commencement of this Declaration

Attention: Daniel Moritz c/o The Arker Companies 1044 Northern Boulevard Roslyn, New York with a copy to: Akerman LLP 666 Fifth Avenue New York, NY 10103

Attention: Steven Sinacori, Esq.

(ii) if to CPC:

New York City Planning Commission 120 Broadway, 31st Floor New York, New York 10271

Attention: Chairperson with a copy to: the General Counsel of CPC at the same address

(iii) if to a Mortgagee:

[mortgagee] at the address provided in writing to CPC in accordance with this Declaration

(b) Declarant, CPC, any Party in Interest, and any Mortgagee may, by notice provided in accordance with this Section, 8.8, change any name or address for purposes of this Declaration. In order to be deemed effective any Notice shall be sent or delivered in at least one of the following manners: (A) sent by registered or certified mail, postage pre-paid, return receipt requested, in which case the Notice shall he deemed delivered for all purposes hereunder five days after being actually mailed; (B) sent by overnight courier service, in which case the Notice shall be deemed delivered for all purposes on the date that the Notice was received or was refused; or (C) delivered by hand, in which case the Notice will be deemed delivered for all purposes on the date that the Notice was received. All Notices from CPC to Declarant shall also be sent to every Mortgagee of whom CPC has notice, and no Notice shall be deemed properly given to Declarant without such notice to such Mortgagee(s). In the event that there is more than one Declarant at any time, any Notice from the City or the CPC shall he provided to all Declarants of whom CPC has notice.

8.8 Applications.

- (a) Declarant shall include a copy of this Declaration with any application for a Building Permit for any portion of the Large-Scale Development Project subject to the Large-Scale Special Permit. Nothing in this Declaration herein shall be construed to prohibit or preclude Declarant from filing for, or DOB from issuing, any permit for all or any portion of the Large-Scale Development Project, in such phase or order as Declarant sees fit in Declarant's sole discretion.
- (b) Nothing in this Declaration shall be construed to prevent Declarant or any of Declarant's successors or assigns from making any application of any sort to any governmental agency or department (each an "Agency") in connection with the development of the Subject

Property; provided, that Declarant shall include a copy of this Declaration in connection with any application for any such discretionary approval, and provided that nothing in this Section, 8.9(b), shall be construed as superseding the requirements, restrictions, or approvals that may be required under agreements with any other Agency or the City.

- 8.9 <u>Severability</u>. In the event that any of the provisions of the Declaration shall be deemed, decreed, adjudged, or determined to be invalid or unlawful by a court of competent jurisdiction, such provision shall be severable and the remainder of this Declaration shall continue to be in full force and effect.
- 8.10 <u>Applicable Law</u>. This Declaration shall be governed and construed by the laws of the State of New York, without regard to principles of conflicts of law.
- 8.11 <u>Exhibits</u>. Any and all exhibits, appendices, or attachments referred to herein are hereby incorporated fully and made an integral part of this Declaration by reference.

ARTICLE IX

CONDOMINIUMS AND COOPERATIVE CORPORATIONS

9.1 Filing Requirements.

- In the event that any part of the Large-Scale Development Project shall be subject to a condominium declaration, or if any part of the Large-Scale Development Project shall be owned or otherwise held by a cooperative corporation in accordance with the provisions of the laws of the State of New York (in either instance, referred to herein as the "Coop/Condominium," and such term shall refer to either organizational form), from and after the date the declaration of condominium has been recorded in the Office of the City Register, or the date that the Large-Scale Development Project (or some portion thereof) shall be in title to the cooperative corporation (the "Coop/Condominium Obligation Date"), under the directorship of a duly elected or appointed Board of Directors or Board of Managers, as the case may be (the "Board"), the Coop/Condominium shall thereafter be deemed to be a Declarant under this Declaration. The owners of the shares of stock of the cooperative corporation, the holder of a lien encumbering any such shares, the holder of a proprietary lease or of any other right to occupancy or other interest therein, the owner of any residential or commercial unit in the condominium, or the holder of a lien encumbering any such condominium unit and the holder of any lease, right of occupancy or any other interest in such condominium unit, or the holder of any Affordable Housing Unit (each of the foregoing, hereinafter, a "Unit Interested Party") shall not be deemed to be a Declarant or a Party in Interest, except to the extent that such Unit Interested Party was a Declarant and retains the right under Applicable Law to control the Board. In the event that a Declarant becomes a Unit Interested Party and the PAA and Private Street Network are not Finally Complete, such party shall continue to be deemed a Declarant irrespective of Board control.
- (b) The initial Declarant or any successor entity to the balance and entirety of Declarant's fee interest in the Subject Property (the "<u>Successor Declarant</u>"), together with the Coop/Condominium, shall be considered Declarants until all Phases of the PAA and Private Street Network are Finally Complete. At such time as Declarant or the Successor Declarant no longer

holds a fee interest in the Subject Property (unless such interest is solely that of a Unit Interested Party and Declarant no longer retains the right to control the Board) and provided that all Phases of the PAA and Private Street Network are Finally Complete, the initial Declarant or the Successor Declarant shall no longer be deemed a Declarant, and the Coop/Condominium shall be deemed to be the sole Declarant and Party in Interest under this Declaration, subject to the provisions of Article XI. In such event, the Board of the Coop/Condominium shall be the sole party with any right to amend, modify, cancel, revise or otherwise change the Declaration, or make any application therefor, subject to the provisions of Article XI.

- (c) Each and every Unit Interested Party hereby (x) irrevocably consents to any amendment, modification, cancellation, revision or other change in this Declaration by the Board; (y) waives and subordinates any rights it may have to enter into an amended Declaration or other instrument amending, modifying, canceling, revising or otherwise changing this Declaration; and (z) nominates, constitutes and appoints the Board its true and lawful attorney-in-fact, coupled with an interest, to execute any documents or instruments that may be required in order to amend, modify, cancel, revise or otherwise change this Declaration and exhibits hereto.
- (d) From and after the Association Obligation Date (hereinafter defined): (i) the Board of any Coop/Condominium shall be responsible for satisfying its obligations as an Association Member (hereinafter defined), and (ii) any Coop/Condominium shall continue to be considered a Declarant, provided that: (x) prior to Final Completion of all Phases of the PAA and Private Street Network, any party that constitutes a Declarant under this Article or Article X shall be jointly and severally responsible for the construction and Final Completion of the PAA and Private Street Network, and (y) upon Final Completion of all Phases of the PAA and Private Street Network, the liability of any Coop/Condominium that is a Declarant shall be limited to the extent of its proportionate share of the Maintenance Obligations assessed by the Association pursuant to Article X.
- Offering Plans. Upon the marketing and sale of securities appurtenant to units in a Coop/Condominium constructed or otherwise included in the Large-Scale Development Project as is contemplated in this Article, a summary of the terms of this Declaration shall be included in any offering plan or "red herring" issued in connection therewith (the "Offering Plan"). Such Offering Plan shall clearly identify the rights and obligations of the unit owners or the owners of shares of stock in the cooperative corporation, as the case may be, under this Declaration, as well as the obligations of the Association Members (hereafter defined) with respect to the assessment of maintenance fees for the ongoing operation and maintenance of the PAA and Private Street Network. The cost of maintenance of the Public Access Area and the obligations of the Coop/Condominium under this Declaration are essential elements of the City actions permitting the development of the Large-Scale Development Project in accordance with the provisions of this Declaration, and in accordance with any other approvals granted by the City, shall be described in any Offering Plan and copies of the Declaration shall be provided as exhibits thereto.
- 9.3 <u>Common Elements</u>. Any condominium declaration shall, upon filing, contain provisions describing all areas covered in the PAA and Private Street Network as "common elements," as that term is constructed under RPL 339-I, or, alternatively, as elements owned jointly by multiple Coops/Condominiums on the zoning lot in their capacity as members of a Property

Owners' Association, for which Unit Interested Parties are obligated to pay assessments, as provided in Article X.

9.4 <u>Estoppel</u>. Declarant shall certify in writing to the Chairperson, or any individual succeeding to its position, that all governing documents of the Coop/Condominium are in full compliance with the requirements of this Declaration and shall provide the Chairperson with copies of such governing documents within ten (10) days of the AG's acceptance for filing of the Offering Plan for such Coop/Condominium. If Declarant fails to comply with the provisions of this Section 9.5, the City may proceed with any available enforcement measures.

ARTICLE X

PROPERTY OWNER'S ASSOCIATION

- 10.1 <u>Property Owner's Association</u>. If the Subject Property is subdivided and conveyed to multiple fee owners and/or ground lessees, Declarant shall cause one property owner's association (the "<u>Association</u>") to be organized, pursuant to the provisions of 13 NYCRR Article 22, to assume all Maintenance Obligations. The obligations of the Associations under this Declaration shall commence on the date that a Declaration of Covenants, Restrictions and Easements establishing the Association is recorded in the Register's Office (the "<u>DCRE</u>") (the "<u>Association Obligation Date</u>"), at which time the Associations shall be deemed a Declarant.
- 10.2 <u>Obligations</u>. The Association shall be established for, among other things, the purposes of assuming the Declarant's performance of the Maintenance Obligations as set forth in this Declaration. In the event that all Phases of the PAA and Private Street Network are not Finally Complete, it shall remain the responsibility of individual Association Members to Finally Complete the PAA and Private Street Network.

10.3 Members.

- (a) The members of the Association (the "<u>Association Members</u>") shall consist of (a) the fee owners and/or ground lessees of any portion of the Subject Property that has not been developed as or converted to a Coop/Condominium, and (b) the Boards of any portion of the Subject Property which is developed as or converted to a Coop/Condominium.
- (b) Each Member hereby (x) irrevocably consents to any amendment, modification, cancellation, revision or other change in this Declaration, relating to the PAA and Private Street Network, by the Association; (y) waives and subordinates any rights it may have to enter into an amended Declaration or other instrument amending, modifying, canceling, revising or otherwise changing this Declaration; and (z) nominates, constitutes and appoints the Association its true and lawful attorney-in-fact, coupled with an interest, to execute any documents or instruments that may be required in order to amend, modify, cancel, revise or otherwise change this Declaration and Exhibits hereto relating to the PAA and Private Street Network.
- (c) Each Member, by acceptance of a deed or lease for a portion of the Subject Property, shall, whether or not the covenant is expressed in such deed or lease, be deemed to have

consented to pay all assessments which may be imposed by the Association on the parcel owned or leased by such Member, as set forth herein.

- 10.4 <u>Powers</u>. Declarant shall cause the Association to be established with the power and authority to:
- (a) Maintain, repair and operate the PAA and Private Street Network to the extent required by this Declaration, and the terms of any license, easement or other agreement that may hereafter be executed between or among DCP, DOT, and Declarant with respect to the PAA and Private Street Network;
- (b) impose fees or assessments against the Association Members, for the purpose of collecting funds necessary to satisfy the obligations of the Association pursuant to this Declaration;
 - (c) collect, receive, administer, protect, invest and dispose of funds;
- (d) bring and defend actions and negotiate and settle claims to recover fees or assessments owed to the Association pursuant to this Article X;
- (e) exercise any of its duties or obligations pursuant to this Declaration without seeking the consent of any Unit Interested Parties; and
- (f) exercise any and all of such powers as may be necessary or appropriate for purposes of this Declaration and as may be granted to the Association in furtherance of the Association's purposes pursuant to the New York Not-for-Profit Corporation Law or New York State Business Corporation Law, or the rules and regulations promulgated by the Department of Law for the regulation of homeowners' (property owner's) Associations, as the case may be.

10.5 Assessments.

- (a) The Association shall assess real property constituting each Coop/Condominium or fee parcel within the Subject Property (the "Assessment Property") for its proportionate share of the Maintenance Obligations as provided in the DCRE, in order to obtain funds for the performance of the obligations of Declarant pursuant to this Declaration. The Assessment Property shall be assessed on a reasonable prorated basis as determined by Declarant, in compliance with Applicable Law. For Association Members who are Boards, a reasonable basis for such proration shall be conclusively established if the AG accepts for filing an offering plan for the sale of interests in such Association and the Coop/Condominium, as applicable, which plan describes such proration. The Boards of each Coop/Condominium shall collect such assessments from the owners of individual residential or commercial units for delivery to the Association in accordance with the DCRE and the governing documents of each Coop/Condominium.
- (b) Each periodic assessment by the Association, together with such interest, costs and reasonable attorney's fees as may be assessed in accordance with the provisions of this Declaration, shall be the obligation of the Association Members against whom the assessment is charged at the time such assessment falls due and may not be waived by such Association Member.

- The periodic assessments shall be a charge on the land and a continuing lien upon the property owned by the Association Member against which each such assessment is made, except that if the Association Member is a Board, such lien shall be subordinate to the lien of any prior recorded mortgage in respect of such property given to a bank or other institutional lender (including but not limited to a governmental agency), the lien of any real property taxes, and the lien of the Board for unpaid maintenance or common charges. The periodic assessments charged to an Association Member which is a Board shall be included within the common charges of the Condominium. The Association may bring an action to foreclose the Association's lien against the property owned by such Association Member, or a Unit Interested Party (other than the owner of an Affordable Housing Unit), as the case may be, to recover such delinquent assessment(s), including interest and costs and reasonable attorneys' fees of any such action. Any Unit Interested Party by acceptance of a deed or a lease to a portion of the Subject Property, thereby agrees to the provisions of this Section, 10.5. Any unit owner may eliminate the Association's lien described above on his or her unit by payment to the Condominium of such unit owner's prorated share of the periodic assessment by the Association to the Condominium in which such Unit is located. No Association Member may waive or otherwise escape liability for the assessments provided for herein by non-use of the Public Access Area or abandonment of the Association's property, or by renunciation of membership in the Association, provided, however, that an Association Member's liability with respect to future assessments shall end upon the valid sale or transfer of such Association Member's interest in the Subject Property.
- (d) Association Members who may be assessed for the Maintenance Obligations shall not include the holder of a mortgage or other lien encumbering (i) the fee estate in the Subject Property or any portion thereof, (ii) the lessee's estate in a ground lease of all or substantially all of the Subject Property or all or substantially all of any parcel or portion thereof, or (iii) any single Building, unless and until any such mortgagee succeeds to either (x) a fee interest in the Subject Property or any portion thereof or (y) the lessee's estate in a ground Lease of all or substantially all the Subject Property or all or substantially all of any Parcel or portion thereof (the interests described in sub-clauses (x) or (y) immediately preceding being each referred to as a ("Possessory Interest") by foreclosure of the lien of the mortgage or other lien or acceptance of a deed or other transfer in lieu of foreclosure or exercise of an option to convert an interest as mortgagee into a Possessory Interest in any such fee or ground leasehold estate in the Subject Property or by other means permitted under Applicable Law from time to time and no such mortgagee or lien holder shall be liable for any assessment imposed by the Association pursuant to this Article 14 until the mortgagee or lien holder succeeds to such Possessory Interest.
- (e) Upon acceptance of an Offering Plan for the sale of membership interests in the Association by the AG, the Association shall certify in writing to the Chairperson or any individual succeeding its position, that all governing documents of the Association are in full compliance with the requirements of this Declaration and shall provide the Chairperson of such governing documents. If Declarant fails to comply with the provision of this Section, 10.5, the City may proceed with any available enforcement measures.
- (f) Association Members shall be considered Declarants. Notwithstanding the foregoing, upon Final Completion of all Phases of the PAA and Private Street Network, the liability of any Association Member shall be limited to such Association Member's proportionate share of the Maintenance Obligations assessed by the Association pursuant to Section 10.5(a). In

the event of a default by the Association, the City shall, prior to enforcing any rights against any Association Members, seek enforcement against the Association and, if applicable, the initial Declarant or a Successor Declarant, pursuant to the notice and cure provisions of Section 7.4.

10.6 <u>Deed References</u>. Every deed conveying title to, or a partial interest in, all or a portion of the Subject Property, every lease of all or any portion of the Subject Property, each condominium declaration, or bylaws relating to any portion of the Subject Property shall contain a recital or other provision that (a) the Association Member is liable for its pro rata share of the assessment by the Association and the obligation to make such payments is binding on such Association Member, and (b) the construction and maintenance of the PAA and Private Street Network and all other obligations of the Association under this Declaration are essential elements of the City actions permitting the development of the Subject Property.

ARTICLE XI

AMENDMENT, MODIFICATION & CANCELLATION

- 11.1 This Declaration may be modified, amended or canceled only upon application by Declarant and subject to the approval and upon express written consent of the Commission or an agency succeeding to the Commission's jurisdiction, and no other approval or consent by any other public body shall be required for such modification, amendment or cancellation.
- 11.2 Notwithstanding anything to the contrary contained in Section 11.1 hereof, any change to this Declaration proposed by Declarant and submitted to the Chairperson, which the Chairperson deems to be a minor modification of this Declaration, may, by express written consent, be approved administratively by the Chairperson and no other approval or consent shall be required from the Commission, any public body, private person or legal entity of any kind.
- 11.3 A modification to a PCRE or Mitigation Measure pursuant to Section 5.9 of this Declaration shall not require a modification of this Declaration.
- 11.4 Notwithstanding anything to the contrary contained in this Declaration, for so long as Declarant (including any successor to its interest as fee owner of all or any portion of the Subject Property, other than a Unit Interested Party) shall hold any fee interest in the Subject Property, or any portion thereof, (i) all Unit Interested Parties, (ii) all boards of managers of any condominium or cooperative association, and (iii) the Association, hereby (x) irrevocably consent to any amendment, modification, cancellation, revision or other change in this Declaration by Declarant; (y) waive and subordinate any rights they may have to enter into an amended Declaration or other instrument amending, modifying, canceling, revising or otherwise changing this Declaration, and (z) nominate, constitute and appoint Declarant, their true and lawful attorney-in-fact, coupled with an interest, to execute any document or instruments that may be required in order to amend, modify, cancel, revise or otherwise change this Declaration.

Approvals given in connection with the Land Use Applications are declared invalid or otherwise voided by a final judgment of any court of competent jurisdiction from which no appeal can be taken or for which no appeal has been taken within the applicable statutory period provided for such appeal, then, upon entry of said judgment or the expiration of the applicable statutory period for such appeal, this Declaration shall be cancelled and shall be of no further force or effect and an instrument discharging it may be recorded. Prior to the recordation of such instrument, Declarant shall notify the Chairperson of Declarant's intent to discharge this Declaration and request the Chairperson's approval, which approval shall be limited to insuring that such discharge and termination is in proper form and provides that the proper provisions which are not discharged survive such termination. Upon recordation of such instrument, Declarant shall provide a copy thereof to the Chairperson so certified by the Register's Office. If some of the Final Approvals given in connection with the Land Use Applications are declared invalid, then Declarant may apply for modification, amendment or cancellation of this Declaration.

IN WITNESS WHEREOF, Declarant has executed this Declaration as of the date first above written.

PARTNERSHIP, a

New	York limited partnership
By:	
J	Name:
	Title:

PENINSULA ROCKAWAY LIMITED

STATE OF NEW YORK)) SS.: COUNTY OF NEW YORK)	
Public in and for said State, personally appearance known to me or proved to me on the basis of is subscribed to the within instrument and a	2019, before me, the undersigned, a Notary ared, personally satisfactory evidence to be the individual whose name acknowledged to me that he executed the same in his strument, the individual, or the person upon behalf of trument.
	Notary Public

EXHIBIT A

EXHIBIT B

PARTIES-IN-INTEREST CERTIFICATION

(attached)

EXHIBIT C

PUBLICLY ACCESSIBLE OPEN SPACE PLAN

(attached)

EXHIBIT D

FORM OF NOTICE OF SUBSTANTIAL COMPLETION

[Letterhead of the Chair of the New York City Planning Commission]

[Date]

NOTICE OF SUBSTANTIAL COMPLETION

Re: Block, Lot, Queens, New York, New York
Dear:
This letter constitutes the Notice of Substantial Completion of the pursuant to the Restrictive Declaration made by PENINSULA ROCKAWAY LIMITED PARTNERSHIP dated as of, (the "Declaration").
By this notice, the undersigned, for the New York City Planning Commission, confirms that the [Street Network Phase or Publicly Accessible Open Space] (as defined in the Declaration) has been Substantially Completed (as defined in the Declaration) in accordance with all requirements of the Declaration.
Yours very truly,

[THIS LETTER SHALL BE MODIFIED AS APPROPRIATE TO THE CERTIFICATION BEING ISSUED]

EXHIBIT E

FORM OF NOTICE OF FINAL COMPLETION

[Letterhead of the Chair of the New York City Planning Commission]

[Date]

NOTICE OF FINAL COMPLETION

Re: Block, Lot, Queens, New Yor	k, New York
Dear:	
This letter constitutes the Notice of Final Com the Restrictive Declaration made by PENINSULA R dated as of, (the "Declaration")	ROCKAWAY LIMITED PARTNERSHIP
By this notice, the undersigned, for the New Yes that the [Street Network Phase or Publicly Accessible Chas been Substantially Completed (as defined in requirements of the Declaration.	Open Space] (as defined in the Declaration)
Yours ver	ry truly,

[THIS LETTER SHALL BE MODIFIED AS APPROPRIATE TO THE CERTIFICATION BEING ISSUED]

Exhibit F-1 Project Components Related to the Environment

Construction Air Quality

- Measures to reduce pollutant emissions during construction of the Large-Scale Development Project will be in accordance with all applicable laws, regulations, and building codes. The proposed emission reduction program will significantly reduce DPM emissions consistent with the goals of the currently best available control technologies under NYC Local Law 77, which are required only for publicly funded City projects. To minimize pollutant emissions and ensure that construction of the Large-Scale Development Project, Declarant has committed to the following measures to reduce air pollutant emissions.
 - Diesel Equipment Reduction. Minimize the use of diesel engines and maximize the use of electric engines where practical.
 - Utilization of Newer Equipment. All nonroad construction equipment will meet EPA's Tier 4 or newer emissions standards for nonroad engines.
 - Best Available Tailpipe Reduction Technologies. In the event nonroad construction equipment for the Proposed Project does not meet EPA Tier 4 or newer emissions standards, then best available tailpipe reduction technologies are to be utilized.
 - a. Nonroad diesel engines with a power rating of 50 hp or greater and controlled truck fleets (i.e., truck fleets under long-term contract, such as concrete mixing and pumping trucks) will utilize the best available tailpipe reduction technology for reducing diesel particulate matter (DPM) emissions, such as diesel particle filters (DPFs).
 - b. Construction contracts will specify that all diesel nonroad engines rated at 50 hp or greater will utilize DPFs, either installed on the engine by the original equipment manufacturer or a retrofit DPF verified by the EPA or the California Air Resources Board, and may include active DPFs, if necessary or other technology proven to achieve equivalent emissions reduction. Stationary equipment will be fitted with devices to reduce NO₂.
 - Dust Control. Fugitive dust control plans will be required as part of contract specifications.
 - a. Truck exit areas will be established for washing off the wheels of all trucks that exit the construction site.
 - b. Tracking pads will be established at construction exits to prevent dirt from being tracked onto roadways.
 - c. Truck routes within the construction site will be either watered or, in cases where such routes will remain in the same place for an extended duration, the routes will be stabilized, covered with gravel, or temporarily paved to avoid the re-suspension of dust.
 - d. All trucks hauling loose material will be equipped with tight fitting tailgates and their loads securely covered prior to leaving the sites. T

- e. Vehicles on-site will be limited to a speed of five mph.
- f. Water sprays and or misting systems will be used for all excavation, and transfer of spoils to ensure that materials are dampened as necessary to avoid the suspension of dust into the air. Loose materials will be watered, stabilized with a biodegradable suppressing agent, or covered. In addition, all necessary measures will be implemented to comply with the NYC Air Pollution Control Code regulating construction-related dust emissions.
- g. Construction areas will be surrounded by perimeter fencing to contain fugitive dust emissions.
- h. Fugitive dust during truck loading will be minimized through the use of screens to block the wind or by carrying out the loading when winds are less than 10 mph.
- o **Idle Times.** On-site vehicle idle time will be restricted to three minutes for all equipment and vehicles that are not using their engines to operate a loading, unloading, or processing device (e.g., concrete mixing trucks) or otherwise required for the proper operation of the engine.
- Source Location. Stationary equipment will be located at least 50 feet away from nearby sensitive receptors (i.e., residential buildings and publicly-accessible open spaces) and at least 30 feet away from sidewalks, to the extent practicable and feasible.
- Ultra-Low Sulfur Fuel. All diesel engines used in construction will use ultra-low sulfur fuel (ULSD).
- Construction Fencing. Install a construction fence at least 8 feet high per the NYC
 Construction Code. The fence between the Building B1 and B2 sites and the Peninsula Nursing Home will be at least 16 feet high.

Construction Noise

- Path controls are placed between the equipment and the sensitive receptors to block noise. The following path controls will be implemented to the extent feasible and practicable.
 - Noise barriers shall be erected around the perimeter of the construction areas where construction activities are taking place to minimize construction noise consistent with reasonable construction procedures. Noise barriers shall be a solid fence with a minimum height of 8 feet. The fence between the Building B1 and B2 sites and the Peninsula Nursing Home will be at least 16 feet high incorporating a sound barrier matting attached to one side of it.
 - Noisy equipment, such as cranes, concrete pumps, concrete trucks, and delivery trucks, will be located away from and shielded from sensitive receptors.
 - o Where feasible, truck deliveries will take place behind noise barriers.
- Source controls reduce noise levels at the source of the noise. The following general source controls will be implemented to the extent feasible and practicable.
 - Equipment shall be properly installed, and quality mufflers must be installed and maintained.

- Construction sites will be configured to minimize back-up alarm noise. In addition, all trucks will not be allowed to idle more than three minutes per Title 24, Chapter 1, Subchapter 7, Section 24-163 of the NYC Administrative Code.
- o Equipment with noise levels quieter than typical noise levels generated by construction equipment will be used.
- Table 1 below summarizes the construction equipment anticipated for use and the developer-committed source and path controls feasible for construction noise as well as construction plans and phasing that will reduce the potential impacts of noisy equipment. These measures are in addition to the construction fencing described above under path controls.

Table 1 PCREs for Construction Activities

Equipment	DEP & FHWA Typical L _{max} Noise Levels at 50 feet (dBA)	Developer- Committed Noise Levels at 50 feet (dBA) (Source Controls)
Stationary Equipment		
Concrete Pump	N.L.	82
Electric Hoist	N.L.	74.1
Manlift	85	75
Pile Rig	95	85
Mobile Equipment		
Bobcat	N.L.	75
Delivery Truck	N.L.	84
Excavators	85	77

Notes:

N.L. - not listed by DEP or in RCNM

Source: NYC CEQR Technical Manual (2014) and FHWA Roadway Construction Noise Model User's Guide (January 2006); NYC Noise Code (2005).

EXHIBIT F-2: MITIGATION MEASURES

Transportation

Mitigation Measures Triggered at the End of Phase 1:

Traffic

- Intersection 3 (Rockaway Beach Boulevard & Beach 116th Street): Reallocate 1 second from NB/SB phase to EB/WB phase in Weekday AM, Weekday MD, Weekday PM peak hours.
- Intersection 4 (Beach Channel Drive & Rockaway Freeway): Reallocate 4 seconds from NB phase to EB/WB phase in Weekday AM (intersection remains unmitigated), Weekday PM (intersection remains unmitigated), and Saturday MD peak hours, 2 seconds from NB phase to EB/WB phase in Weekday MD peak hour. Note: the intersection will be partially mitigated during the Weekday AM and PM peak hours, as identified in Chapter 20, "Mitigation."
- Intersection 5 (Beach Channel Drive & Beach 108th Street): Reallocate 3 seconds from NB phase to EB/WB phase in Weekday AM peak hour.
- Intersection 26 (Beach Channel Drive and Beach 53rd Street): Signalize the intersection.
- Intersection 42 (Rockaway Freeway and Seagirt Boulevard): Change signal offset in Weekday MD peak hour by 6 seconds.
- Intersection 49 (Beach Channel Drive and Nameoke Avenue): Reallocate 3 seconds from EB phase to NB/SB phase in Weekday MD and Saturday MD peak hours, 4 seconds from EB phase to NB/SB phase in Weekday PM peak hour.

Mitigation Measures Triggered at the End of Phase 2:

Traffic

- Intersection 7 (Rockaway Beach Boulevard and Beach 108th Street): Reallocate 4 seconds from NB/SB phase to EB/WB phase in Weekday AM peak hour.
- Intersection 8 (Beach Channel Drive & Beach 92nd Street/Beach 94th Street): Reallocate 1 second from EB/WB phase to NEB (Cross Bay Bridge & Beach 94th Street)/WB phase in Weekday AM and Weekday MD peak hours, 2 seconds from EB/WB phase to NEB (Cross Bay Bridge & Beach 94th Street)/WB phase in Weekday PM and Saturday MD peak hours.
- Intersection 23 (Arverne Boulevard and Beach 54th Street): Install "No Standing Anytime" parking regulation along the north curb of the WB approach between Beach 54th Street and Beach 53rd Street (intersection remains unmitigated during the Weekday MD and Weekday PM peak hours). *Note: the intersection will be partially mitigated during the Weekday MD and PM peak hours, as identified in Chapter 20, "Mitigation."*
- Intersection 27 (Rockaway Beach Boulevard and Beach 53rd Street): Signalize the intersection. Restripe the EB approach as one left-turn lane and one through lane. Restripe the WB approach to align the EB and WB approaches and eliminate on-street parking along the north curb of the WB receiving lanes.

Mitigation Measures Triggered at the End of Phase 3:

Traffic

- Intersection 13 (Beach Channel Drive & Beach 73rd Street): Restripe the EB approach to provide one left-turn/through lane and one through lane and eliminate on-street parking to provide one additional eastbound receiving lane. Restripe the WB approach and eliminate on-street parking to provide one left-turn/through lane, and one through/right-turn lane and eliminate on-street parking to provide one additional westbound receiving lane. Install "No Standing Anytime" parking regulations along the north curb of the WB approach (for approximately 170') and along the north curb of the WB receiving lanes between Beach 73rd and Beach 74th streets and between Beach 74th and Beach 75th streets (for approximately 120'). Install "No Standing Anytime" parking regulations along the south curb of the EB receiving lanes (for approximately 125').
- Intersection 19 (Arverne Boulevard and Beach 59th Street): Restripe the WB approach to provide one left-turn turn bay, one through lane, and one parking lane. Restripe the EB approach to provide a median to better align the EB and WB approaches. Reallocate 1 second from SB phase to EB/WB phase in Weekday PM peak hour.

Pedestrians

Crosswalks:

South crosswalk at Beach 54th Street and Beach Channel Drive Crosswalk: widen the crosswalk by six feet and relocate stop bar 10 feet from crosswalk.

Mitigation Measures Triggered at the End of Phase 4:

Traffic

- Intersection 16 (Rockaway Beach Boulevard and Beach 62nd Street): Reallocate 4 seconds from EB phase to EB/WB phase in Weekday AM peak hour.
- Intersection 20 (Rockaway Freeway and Beach 59th Street): Reallocate 2 seconds from SB phase to WB phase in Weekday MD peak hour, 1 second from SB phase to WB phase in Saturday MD peak hour.
- Intersection 21 (Rockaway Beach Boulevard and Beach 59th Street): Reallocate 2 seconds from NB/SB phase to SB phase in Weekday MD peak hour to mitigate intersection 20, and 1 second from NB/SB phase to SB phase in Saturday MD peak hour to mitigate intersection 20. *Note: intersection 21 will be unmitigable during all the peak hours, as identified in Chapter 20, "Mitigation."*
- Intersection 28 (Rockaway Beach Boulevard and Beach 52nd Street): Signalize the intersection. Restripe the EB approach to one wider left-turn/through/right-turn lanes and one narrower parking lane.

Traffic (Additional Intersections for Peak Construction in Phase 4)

- Intersection 15 (Beach Channel Drive and Beach 62nd Street): Reallocate 1 second from NB /SB phase to WB (Arverne Boulevard) phase in Weekday PM Saturday PM peak hours.
- Intersection 24 (Rockaway Freeway and Beach 54th Street): Reallocate 2 seconds from NB/SB phase to EB/WB phase and 1 second from NB/SB phase to WB phase during the

Weekday PM peak hour to mitigate intersection 25. Reallocate 1 second from NB/SB phase to EB/WB phase and 1 second from NB/SB phase to WB phase during the Saturday PM peak hour to mitigate intersection 25.

- Intersection 25 (Edgemere Avenue and Beach 54th Street): Reallocate 2 seconds from NB/SB phase to EB/WB phase and 1 second from NB/SB phase to SB phase during the Weekday PM peak hour to mitigate intersection 25. Reallocate 1 second from NB/SB phase to EB/WB phase and 1 second from NB/SB phase to SB phase during the Saturday PM peak hour to mitigate intersection 25.
- Intersection 46 (Beach Channel Drive and Mott Avenue): Reallocate 1 second from EB/WB phase to NB/SB phase in Saturday PM peak hour.
- Intersection 47 (Beach Channel Drive and Dix Avenue): Reallocate 1 second from EB/WB phase to NB/SB phase in Weekday PM and Saturday PM peak hours.

Mitigation Measures Triggered at the Full Build-Out of the Proposed Project:

Traffic

- Intersection 44 (Rockaway Freeway and Cornaga Avenue): Reallocate 1 second from NB/SB phase to EB/WB phase in Weekday AM peak hour.
- Intersection 48 (Beach Channel Drive and Birdsall Avenue): Reallocate 1 second from EB/WB phase to NB/SB phase in Weekday AM peak hour, 3 seconds from EB/WB phase to NB/SB phase in Weekday PM peak hour.
- Intersection 50 (Beach Channel Drive and Hassock Street): Reallocate 1 second from EB/WB phase to NB/SB phase in Weekday AM peak hour.

Pedestrians

Sidewalks:

North sidewalk on the west leg of Beach 56th Street and Arverne Boulevard: pave one section of unpaved sidewalk (grass) with concrete.

Operational Air Quality

The maximum predicted PM_{2.5} concentrations at the Rockaway Beach Boulevard/Beach 54th Street/Beach 53rd Street is predicted to exceed NAAQS and result in a significant adverse air quality impact. Therefore, a traffic mitigation measure to signalize Rockaway Beach Boulevard/Beach 53rd Street intersection location will be required. Signalization of this intersection is necessary to provide gaps for vehicles traveling southbound on Beach 53rd Street due to the high pedestrian volumes expected on the north crosswalk.

Construction Noise

• The following additional path controls will be implemented as mitigation to the extent feasible and practicable: portable noise barriers, panels, curtains, enclosures, and acoustical tents. The required construction and materials for these measures are as shown in Chapter 28 of NYCDEP's Citywide Construction Noise Mitigation. Where practicable and feasible, noise curtains and equipment enclosures will be placed around the noisy equipment shown in Table 1 to provide an STC of 30 (a 5 dBA insertion loss) or greater of noise shielding to sensitive receptor locations.

Table 1: Construction Path Controls

Equipment	DEP & FHWA Typical L _{max} Noise Levels at 50 feet (dBA)	Developer- Committed Noise Levels at 50 feet (dBA) (Source Controls)	Developer- Committed Path Controls and Construction Practices
Air Compressor (< 350 cfm)	75-80	67	Portable enclosure; position so noise levels at sensitive receptors < 75 dBA
Bobcat	N.L.	75	Source control
Chain Saws	85	75	Portable enclosure
Circular Saws	76	76	Portable enclosure
Crane: Manitowoc 999	85	77	Portable enclosure
Cut-Off Saw	N.L.	76	Source control
Electric Hoist	N.L.	74.1	Source control
Excavators	85	77	Source control
Drum Mixer	80	76	Portable enclosure
Generators	70-82	68	Portable enclosure
Manlift	85	75	Source control
Jackhammer	85	85	Portable enclosure
Mortar Mixer	N.L.	85	Portable enclosure
Pile Rig	95	85	Source control
Pumps (Grout)	77	77	Portable enclosure
Rebar Bending Machine	80	80	Portable enclosure

Notes:

 $N.L.-not\ listed\ by\ DEP\ or\ in\ RCNM$

Source: NYC CEQR Technical Manual (2014) and FHWA Roadway Construction Noise Model User's Guide (January 2006); NYC Noise Code (2005).

EXHIBIT G

Project Phasing Diagram

[Attached]

EXHIBIT H

Private Street Network Temporary and Permanent Design Drawings

[Attached]





Community/Borough Board Recommendation

Pursuant to the Uniform Land Use Review Procedure

Application #: C 190375 ZSQ

Project Name: Peninsula Redevelopment

CEQR Number: 18DCP124Q

Borough(s): Queens

Community District Number(s): 14

Please use the above application number on all correspondence concerning this application

SUBMISSION INSTRUCTIONS

- 1. Complete this form and return to the Department of City Planning by one of the following options:
 - EMAIL (recommended): Send email to CalendarOffice@planning.nyc.gov and include the following subject line:

 - (CB or BP) Recommendation + (6-digit application number), e.g., "CB Recommendation #C100000ZSQ"

 MAIL: Calendar Information Office, City Planning Commission, 120 Broadway, 31st Floor, New York, NY 10271
 - FAX: to (212) 720-3488 and note "Attention of the Calendar Office"
- Send one copy of the completed form with any attachments to the applicant's representative at the address listed below, one copy to the Borough President, and one copy to the Borough Board, when applicable

Docket Description:

IN THE MATTER OF an application submitted by Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the New York City Charter for the grant of a special permit pursuant to Section 74-744(c)(1) of the Zoning Resolution to modify the surface area requirements of Section 32-64 (Surface Area and Illumination Provisions), in connection with a proposed mixed used development, within a large-scale general development, on property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, Beach 52nd Street, Rockaway Beach Boulevard and Beach 53rd Street (Block 15842, Lot 1 & p/o Lot 100, Block 15843, Lot 1, and Block 15857 Lot 1 & p/o Lot 7), in a C4-4* and C4-3A* Districts, Borough of Queens, Community District 14.

* Note: The site is proposed to be rezoned by eliminating a C1-2 District within an existing R5 District and by changing an existing R5 and C8-1 Districts to C4-4 and C4-3A Districts under a concurrent related application for a Zoning Map change (C 190352 ZMQ).

Plans for this proposal are on file with the City Planning Commission and may be seen at 120 Broadway, 31st Floor, New York, N.Y. 10271-0001.

Applicant(s):		Applicant's Rep	Applicant's Representative:		
Peninsula Rockaway Limited Partnership		1 *	Jaclyn C. Scarinci, Esq.		
1044 Northern Boulevard, 2 nd Floor			Akerman, LLP		
Roslyn, NY 11256		666 Fifth Avenue, 20th Floor New York, NY 10103			
Recommendation submitted by: Queens Community Board 14					
Date of public hearing: June 25 80	内 Location: Kni.	ents of column	bus Hail 10 Street		
Was a quorum present? YES NO		res a quorum of 20% of the ap han seven such members.	pointed members of the board,		
Date of Vote: JUNE 25 70 RECOMMENDATION 25 TOTAL	19 Location: 35	33 Beach	90 street		
· _ 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 - 100 -		그렇게 되었다. 그렇게 그는 얼마를 받는 그 살아보다 그	교회 전문을 하면 있다. 급하게 되었다는 그 사이를 가라지는 이렇다		
Approve		n Modifications/Condition			
Disapprove	Disapprove With Modifications/Conditions *				
Please attach any further explanation of the	ie recommendation on a	idditional sheets, as n	ecessary.		
Voting #In Favor: 건승 #Against: 与 #Abs	staining: Total	members appointed to	othe board:		
Name of CB/BB officer completing this fol	in Tie	e	Date JUN 3 2014		
Jonathan Gaska		orst mar	1-013-014		

On Tuesday – June 25, 2019 a Special meeting of Community Board #14Q was held at 333 Beach 90 St. Rockaway Beach, NY. A quorum being present a Public Hearing was held regarding: ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ- Peninsula Rockaway Limited Partnership- redevelopment of Peninsula Hospital site

After a Public Hearing with 60 speakers, a presentation by the Peninsula Rockaway Limited Partnership, a statement from Councilmember Donovan Richards read by Chief of Staff Manuel Silva, the following motion carried by a vote of 28-yes; 5- no; 1 –abstention.

Motion to Disapprove with Modifications/Conditions ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ.

- 1. <u>Density</u>-Total amount of units should be reduced by a minimum of 30% with strong consideration for a 50% reduction in total units.
- 2. Parking residential parking should be increased to a minimum of 75% of total units built.
- 3. <u>Schools</u> That the negative impact on local schools identified in the DEIS shall be addressed by a new K-8 Department of Education school being funded, built and zoned for new residents of the project and existing local students in the immediate area for the purpose of creating a neighborhood school.
- 4. Worker Wages That a prevailing wage agreement be signed which will be applicable for all workers.
- 5. <u>Transportation/Traffic</u> -That in consultation with the Community Board and community- a holistic transportation/traffic plan be created and funded to mitigate the numerous traffic issues identified in DEIS that will affect the entire Peninsula from the East to the West ends of the District/ peninsula.
- 6. <u>AMI</u> -That the final AMI mix of affordable units shall be discussed with the Community Board to determine what is needed and appropriate. The board strongly supports the addition of Affordable Senior Units.
- 7. <u>Communication</u> That the developer (ARKER) be required to report to the Community Board every five (5) months as to the progress of the project and to discuss and resolve issues that may arise. The developer shall create and maintain a website that shall include updates on project and job opportunities related to the project.
- 8. Shadow impact That the height of the buildings shall not exceed 12 stories
- 9. <u>Park</u> That the city owned property (HPD) located north of Beach Channel Drive between approximately Beach 51 to Beach 48 Streets be deeded to NYC Parks and Recreation Department and that the Developer (Arker) provide capital funding for the planning and construction of a community park which shall include Park House, multipurpose field and running track, children's playground with water features. That the developer shall provide funding for maintenance of the park on a yearly basis
- 10 That a <u>Community Benefits Agreement</u> (CBA) be signed that guarantees that Community Board # 14 residents get construction and other available jobs at development site, including supermarket, etc.
- 11. Resident Preference Residents of Community Board #14 shall get a minimum of a 50% preference of all available units in each phase of development.





Community/Borough Board Recommendation

Pursuant to the Uniform Land Use Review Procedure

Application #: C 190366 ZSQ

Project Name: Peninsula Redevelopment

CEQR Number: 18DCP124Q

Borough(s): Queens

Community District Number(s): 14

Please use the above application number on all correspondence concerning this application

SUBMISSION INSTRUCTIONS

- 1. Complete this form and return to the Department of City Planning by one of the following options:
 - EMAIL (recommended): Send email to <u>CalendarOffice@planning.nyc.gov</u> and include the following subject line: (CB or BP) Recommendation + (6-digit application number), e.g., "CB Recommendation #C100000ZSQ"
 - MAIL: Calendar Information Office, City Planning Commission, 120 Broadway, 31st Floor, New York, NY 10271
 - FAX: to (212) 720-3488 and note "Attention of the Calendar Office"
- Send one copy of the completed form with any attachments to the <u>applicant's representative</u> at the address listed below, one copy to the Borough President, and one copy to the Borough Board, when applicable.

Docket Description:

IN THE MATTER OF an application submitted by Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the New York City Charter for the grant of a special permit pursuant to Section 74-743(a)(2) of the Zoning Resolution to modify:

- the rear yard requirements of Section 23-533 (Required rear yard equivalents for Quality Housing buildings) and Section 35-53 (Modification of Rear Yard Requirements):
- the side yard requirements of Section 35-54 (Special Provisions Applying Adjacent to R1 Through R5 Districts); and
- 3. the height and setback requirements of Section 23-664 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors) and Section 35-654 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors);

in connection with a proposed mixed used development, within a large-scale general development, on property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, Beach 52nd Street, Rockaway Beach Boulevard and Beach 53nd Street (Block 15842, Lot 1 & p/o Lot 100, Block 15843, Lot 1, and Block 15857 Lot 1 & p/o Lot 7), in a C4-4* and C4-3A* Districts, Borough of Queens, Community District 14.

* Note: The site is proposed to be rezoned by eliminating a C1-2 District within an existing R5 District and by changing an existing R5 and C8-1 Districts to C4-4 and C4-3A Districts under a concurrent related application for a Zoning Map change (C 190352 ZMQ).

Plans for this proposal are on file with the City Planning Commission and may be seen at 120 Broadway. 31st Floor. New York, N.Y. 10271-0001.

rialis for this proposal are on the with the City Fianning Commission a	na may be seen at	120 Broadway, 51 Troor, 110 W Term, 11			
Applicant(s): Applicant's Representative:		presentative:			
Peninsula Rockaway Limited Partnership		•	Jaclyn C. Scarinci, Esq.		
1044 Northern Boulevard, 2nd Floor	044 Northern Boulevard, 2nd Floor		Akerman, LLP		
Roslyn, NY 11256		666 Fifth Aven	,		
		New York, NY	10103		
Recommendation submitted by:					
Queens Community Board 14					
Date of public hearing: JUNE 25 2017	Location:	Knishts of 333 Beac	columbus Hall n 90 Street		
Was a quorum present? YES NO		g requires a quorum of 20% of the a rewer than seven such members.			
Date of Vote: JUNE 25 2019	Location:	333 Blach	90 Street		
RECOMMENDATION					
Approve *	Approv	e With Modifications/Condition	ns " ø		
		rove With Modifications/Cond	- Security		
Disapprove	Pag Disapp	rove vintri Midelingalions/Gond	IIIOUS SHAPE		
Please attach any further explanation of the reo	ommendation	ron augilional sweets, as r	arazzaik.		
Voting Board Infloren Wasto Dis.	*bbubh	m was condu	HAND		
#In Favor: 28 #Against: 5 #Abstainir	ig: / .	Total members appointed t	o the board:		
Name of CB/8B officer completing this form	200 mm (100 mm) 200 mm	Title	Date		
		h til	120143,2019		
Jonathan GASKA		DOL WELL			

On Tuesday – June 25, 2019 a Special meeting of Community Board #14Q was held at 333 Beach 90 St. Rockaway Beach, NY. A quorum being present a Public Hearing was held regarding: ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ- Peninsula Rockaway Limited Partnership- redevelopment of Peninsula Hospital site

After a Public Hearing with 60 speakers, a presentation by the Peninsula Rockaway Limited Partnership, a statement from Councilmember Donovan Richards read by Chief of Staff Manuel Silva, the following motion carried by a vote of 28-yes; 5-no; 1—abstention.

Motion to Disapprove with Modifications/Conditions ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ.

- 1. <u>Density</u>-Total amount of units should be reduced by a minimum of 30% with strong consideration for a 50% reduction in total units.
- 2. <u>Parking</u> residential parking should be increased to a minimum of 75% of total units built.
- 3. <u>Schools</u> That the negative impact on local schools identified in the DEIS shall be addressed by a new K-8 Department of Education school being funded, built and zoned for new residents of the project and existing local students in the immediate area for the purpose of creating a neighborhood school.
- 4. Worker Wages That a prevailing wage agreement be signed which will be applicable for all workers.
- 5. <u>Transportation/Traffic</u> -That in consultation with the Community Board and community- a holistic transportation/traffic plan be created and funded to mitigate the numerous traffic issues identified in DEIS that will affect the entire Peninsula from the East to the West ends of the District/ peninsula.
- 6. <u>AMI</u> -That the final AMI mix of affordable units shall be discussed with the Community Board to determine what is needed and appropriate. The board strongly supports the addition of Affordable Senior Units .
- 7. <u>Communication</u> That the developer (ARKER) be required to report to the Community Board every five (5) months as to the progress of the project and to discuss and resolve issues that may arise. The developer shall create and maintain a website that shall include updates on project and job opportunities related to the project.
- 8. Shadow impact- That the height of the buildings shall not exceed 12 stories
- 9. <u>Park</u> That the city owned property (HPD) located north of Beach Channel Drive between approximately Beach 51 to Beach 48 Streets be deeded to NYC Parks and Recreation Department and that the Developer (Arker) provide capital funding for the planning and construction of a community park which shall include Park House, multipurpose field and running track, children's playground with water features. That the developer shall provide funding for maintenance of the park on a yearly basis
- 10 That a <u>Community Benefits Agreement</u> (CBA) be signed that guarantees that Community Board # 14 residents get construction and other available jobs at development site, including supermarket, etc.
- 11. Resident Preference Residents of Community Board #14 shall get a minimum of a 50% preference of all available units in each phase of development.





Community/Borough Board Recommendation

Pursuant to the Uniform Land Use Review Procedure

Application #:

C 190251 MMQ

Project Name: Peninsula Hospital Redevelopment Plan

CEQR Number: 18DCP124Q

Borough(s): Queens

Community District Number(s): 14

Please use the above application number on all correspondence concerning this application

SUBMISSION INSTRUCTIONS

1. Complete this form and return to the Department of City Planning by one of the following options:

- EMAIL (recommended): Send email to CalendarOffice@planning.nyc.gov and include the following subject line:
- (CB or BP) Recommendation + (6-digit application number), e.g., "CB Recommendation #C100000ZSQ"

 MAIL: Calendar Information Office, City Planning Commission, Room 2E, 22 Reade Street, New York, NY 10007
- FAX: (212) 720-3356 and note "Attention of the Calendar Office"
- Send one copy of the completed form with any attachments to the applicant's representative at the address listed below, one copy to the Borough President, and one copy to the Borough Board, when applicable.

Docket Description.

IN THE MATTER OF an application submitted by Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 199 of the New York City Charter for an amendment to the City Map involving:

- the establishment of a portion of Beach 52nd Street between Rockaway Beach Boulevard and Shorefront Parkway;
- the adjustment of grades and block dimensions necessitated thereby;

including authorization for any acquisition or disposition of real property related thereto, in Community District 14, Borough of Queens, in accordance with Map No. 5033 dated April 17, 2019 and signed by the Borough President.

Applicant(s):	Applicant's Representative:	
Peninsula Rockaway Limited Partnership 1044 Northern Boulevard, 2nd Floor Roslyn, NY 11256 516.277.9300	Eleanore C. Martin, Esq. Akerman LLP 666 Fifth Avenue, 20th Floor New York, NY 10103 212.259.6456	
Recommendation submitted by: Queens Community Board 14		
Date of public hearing: JUNR 25 20 19 Location:	anishts of columbus 333 Beach 90 Stru	Hall el
	g requires a quorum of 20% of the appointed membe lewer than seven such members.	rs of the board,
	333 Beach 90 stro	হা
<u> 호텔하는</u> 이 등 전한 시간을 통하는 시간 하는 경험 시간 하는 모든 회에 가장 이 사람들이 되는 경험하는 데 되는 것은	approve w modicondu	- Sel
	e With Modifications/Conditions	િંહ
Disapprove Disapp	rove With Modifications/Conditions	
Please attach any further explanation of the recommendation	on additional sheets, as necessary.	
Voting # In Favor: 28 # Against: 5 # Abstaining:	Total members appointed to the board:	
Name of CB/BB officer completing this form	Title Date	
Jonathan GASKA	Net mai win	3 2019

On Tuesday – June 25, 2019 a Special meeting of Community Board #14Q was held at 333 Beach 90 St. Rockaway Beach, NY. A quorum being present a Public Hearing was held regarding: ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ- Peninsula Rockaway Limited Partnership- redevelopment of Peninsula Hospital site

After a Public Hearing with 60 speakers, a presentation by the Peninsula Rockaway Limited Partnership, a statement from Councilmember Donovan Richards read by Chief of Staff Manuel Silva, the following motion carried by a vote of 28-yes; 5- no; 1 –abstention.

Motion to Disapprove with Modifications/Conditions ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ.

- 1. <u>Density</u>-Total amount of units should be reduced by a minimum of 30% with strong consideration for a 50% reduction in total units.
- 2. Parking residential parking should be increased to a minimum of 75% of total units built.
- 3. <u>Schools</u> That the negative impact on local schools identified in the DEIS shall be addressed by a new K-8 Department of Education school being funded, built and zoned for new residents of the project and existing local students in the immediate area for the purpose of creating a neighborhood school.
- 4. Worker Wages That a prevailing wage agreement be signed which will be applicable for all workers.
- 5. <u>Transportation/Traffic</u> -That in consultation with the Community Board and community- a holistic transportation/traffic plan be created and funded to mitigate the numerous traffic issues identified in DEIS that will affect the entire Peninsula from the East to the West ends of the District/ peninsula.
- 6. <u>AMI</u> -That the final AMI mix of affordable units shall be discussed with the Community Board to determine what is needed and appropriate. The board strongly supports the addition of Affordable Senior Units .
- 7. <u>Communication</u> That the developer (ARKER) be required to report to the Community Board every five (5) months as to the progress of the project and to discuss and resolve issues that may arise. The developer shall create and maintain a website that shall include updates on project and job opportunities related to the project.
- 8. Shadow impact- That the height of the buildings shall not exceed 12 stories
- 9. <u>Park</u> That the city owned property (HPD) located north of Beach Channel Drive between approximately Beach 51 to Beach 48 Streets be deeded to NYC Parks and Recreation Department and that the Developer (Arker) provide capital funding for the planning and construction of a community park which shall include Park House, multipurpose field and running track, children's playground with water features. That the developer shall provide funding for maintenance of the park on a yearly basis
- 10 That a <u>Community Benefits Agreement</u> (CBA) be signed that guarantees that Community Board # 14 residents get construction and other available jobs at development site, including supermarket, etc.
- 11. Resident Preference Residents of Community Board #14 shall get a minimum of a 50% preference of all available units in each phase of development.



Community/Borough Board Recommendation

Pursuant to the Uniform Land Use Review Procedure

Application #: C 190325 ZMQ

Project Name Peninsula Redevelopment

CEGR Number 18DCP124Q

Boroughts): Queens

Community District Number(s), 14

Please use the above application number on all correspondence concerning this application

SUBMISSION INSTRUCTIONS

- Complete this form and return to the Department of City Planning by one of the following options:
 - EMAIL (recommended): Send email to CalendarOffice@planning.nyc.gov and include the following subject line: (CB or BP) Recommendation + (6-digit application number), e.g., "CB Recommendation #C100000ZSQ"

 MAIL: Calendar Information Office, City Planning Commission, 120 Broadway, 31st Floor, New York, NY 10271

 - FAX: to (212) 720-3488 and note "Attention of the Calendar Office"
- Send one copy of the completed form with any attachments to the applicant's representative at the address listed below, one copy to the Borough President, and one copy to the Borough Board, when applicable.

Docket Description

IN THE MATTER OF an application submitted by Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the New York City Charter for the amendment of the Zoning Map, Section No. 30c:

- eliminating from within an existing R5 District a C1-2 District bounded by a line 420 feet southerly of Beach Channel Drive. Beach 50th Street, Rockaway Beach Boulevard, and the easterly street line of former Beach 51st Street;
- changing from an R5 District to a C4-4 District property bounded by Beach Channel Drive, the westerly street line of former 2. Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, and Beach 53rd Street; and
- changing from a C8-1 District to a C4-3A District property bounded by Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, and Beach 52nd Street;

Borough of Queens. Community District 14, as shown on a diagram (for illustrative purposes only) dated May 6, 2019, and subject to the conditions of CEQR Declaration E-532.

Applicant(s):		Applicant's Representative:	
Peninsula Rockaway Limited Partnership 1044 Northern Boulevard. 2 nd Floor Roslyn. NY 11256	·	Jaclyn C. Scarinci, Esq. Akerman, LLP 666 Fifth Avenue, 20 th Floor New York, NY 10103	
Recommendation submitted by:			
Queens Community Board 14			
Date of public hearing: June 25. 2019		nto of columb 133 Beach 90 &	
Was a quorum present? YES NO	A public hearing requires a qu but in no event fewer than seve	orum of 20% of the appointed members en such members.	of the board,
Date of Vote: June 25, 2019	Location: 33	3 Beach 90st	
RECOMMENDATION Bd motion was	edow - Disapp	more m/condin	NO.
Approve	Approve With Modi	•	~ Jones
Disapprove	Disapprove With M	odifications/Conditions	- See to mope
Please attach any further explanation of the rec	ommendation on addition	nal sheets, as necessary.	* ***********************************
Voting			
#In Favor: 🗸 🧗 # Agamst: 🖒 # Abstainin	rg: Total memi	pers appointed to the board:	
Name of CB/BB officer completing this form	Title	t mr r Date-Ju	14 ,3 20Kg

On Tuesday – June 25, 2019 a Special meeting of Community Board #14Q was held at 333 Beach 90 St. Rockaway Beach, NY. A quorum being present a Public Hearing was held regarding: ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ- Peninsula Rockaway Limited Partnership- redevelopment of Peninsula Hospital site

After a Public Hearing with 60 speakers, a presentation by the Peninsula Rockaway Limited Partnership, a statement from Councilmember Donovan Richards read by Chief of Staff Manuel Silva, the following motion carried by a vote of 28-yes; 5- no; 1 – abstention.

Motion to Disapprove with Modifications/Conditions ULURP#'s C-190375-ZSQ, C-190366 ZSQ, C-190251 MMQ, 190325 ZMQ.

- 1. **<u>Density</u>**-Total amount of units should be reduced by a minimum of 30% with strong consideration for a 50% reduction in total units.
- 2. Parking residential parking should be increased to a minimum of 75% of total units built.
- 3. <u>Schools</u> That the negative impact on local schools identified in the DEIS shall be addressed by a new K-8 Department of Education school being funded, built and zoned for new residents of the project and existing local students in the immediate area for the purpose of creating a neighborhood school.
- 4. Worker Wages That a prevailing wage agreement be signed which will be applicable for all workers.
- 5. <u>Transportation/Traffic</u> -That in consultation with the Community Board and community- a holistic transportation/traffic plan be created and funded to mitigate the numerous traffic issues identified in DEIS that will affect the entire Peninsula from the East to the West ends of the District/ peninsula.
- 6. <u>AMI</u> -That the final AMI mix of affordable units shall be discussed with the Community Board to determine what is needed and appropriate. The board strongly supports the addition of Affordable Senior Units.
- 7. <u>Communication</u> That the developer (ARKER) be required to report to the Community Board every five (5) months as to the progress of the project and to discuss and resolve issues that may arise. The developer shall create and maintain a website that shall include updates on project and job opportunities related to the project.
- 8. Shadow impact That the height of the buildings shall not exceed 12 stories
- 9. <u>Park</u> That the city owned property (HPD) located north of Beach Channel Drive between approximately Beach 51 to Beach 48 Streets be deeded to NYC Parks and Recreation Department and that the Developer (Arker) provide capital funding for the planning and construction of a community park which shall include Park House, multipurpose field and running track, children's playground with water features. That the developer shall provide funding for maintenance of the park on a yearly basis
- 10 That a <u>Community Benefits Agreement</u> (CBA) be signed that guarantees that Community Board # 14 residents get construction and other available jobs at development site, including supermarket, etc.
- 11. Resident Preference Residents of Community Board #14 shall get a minimum of a 50% preference of all available units in each phase of development.

Queens Borough President Recommendation

APPLICATION: ULURP #190251 MMQ

COMMUNITY BOARD: Q14

DOCKET DESCRIPTION

IN THE MATTER OF an application submitted by Akerman LLP on behalf of the Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 199 of the NYC Charter for an amendment of the City Map involving:

- the establishment of a portion of Beach 52nd Street between Rockaway Beach Boulevard and Shorefront Parkway;
- the adjustment of grades and block dimensions necessitated thereby:

including authorization for any acquisition or disposition of real property related thereto, in Community District 14, Borough of Queens, in accordance with Map No. 5033 dated April 17, 2019 and signed by the Borough President

(Related: ULURPs #190325 ZMQ, N190364 ZRQ, 190366 ZSQ, 190375 ZSQ)

PUBLIC HEARING

A Public Hearing was held in the Borough President's Conference Room at 120-55 Queens Boulevard on Thursday, July 11, 2019 at 10:30 A.M. pursuant to Section 82(5) of the New York City Charter and was duly advertised in the manner specified in Section 197-c (i) of the New York City Charter. The applicant made a presentation. There were ten (10) speakers in favor with five (5) speakers against. The hearing was closed.

CONSIDERATION

Subsequent to a review of the application and consideration of testimony received at the public hearing, the following issues and impacts have been identified:

- The applicant is proposing to re-establish and re-open a portion of Beach 52nd Street between Rockaway Beach Boulevard and the Rockaway Freeway. The reopening of this portion of Beach 52nd Street would improve traffic circulation in the area. This application was filed as part of the proposed redevelopment of the former Peninsula Hospital site as a Large Scale General Development (LSGD);
- The applicant has also concurrently filed other applications to facilitate development of the proposed LSGD. The other applications are for a rezoning to C4-4 and C4-3A, zoning text amendments to Appendix F of the NYC Zoning Resolution establishing and designating the area to be rezoned as a Mandatory Inclusionary Housing Area and to allow a Physical Culture Establishment within the LSGD, a Special Permit allowing modification of yard, height and setback requirements, and a Special Permit to modify signage regulations for potential commercial uses in the LSGD;
- The proposed LSGD would total 2,371,000 gross square feet built in 11 buildings with heights varying between 8- to 19-stories over six sub-sections of the 410,052.28 sf site. There would be a total of 2,200 residential units. There would be 1,927 units for incomes up to 80% AMI including 201 units of Affordable Independent Residences for Seniors (AIRS) and 273 units of moderate income housing. A 20,000 sf supermarket would be built in Phase I of the project as part of the 72,000 sf of retail space proposed in the project. 77,000 sf of community space is proposed that would allow schools, day care or medical care facilities, or cultural space. Parking would be provided in 973 accessory parking spaces located in parking lots and garages distributed throughout the development area with 754 residential spaces, 144 retail spaces and 75 spaces for community facility/ medical office uses. A proposed privately owned, open internal roadway network with two 60 feet wide streets (north-south extension of Beach 52nd Street, and an east-west Peninsula Way between Beach 50th & Beach 53rd Streets) that would create four quadrants on the site. 24,000 sf of publicly accessible open space would be distributed throughout the site. Two major components would be an open space identified as Pedestrian Plaza (Beach 51st Street) and Highpoint Intersection (intersection of Beach 52nd Street & Peninsula Way). Both would be resilient gathering points for pedestrians with planters and seating. The Beach 51st Street open space would have a children's play area, rubber play surfaces, and a pedestrian plaza with publicly accessible amenities. The phased build out is projected to be 10 to 15 years;
- The project site is the former site of Peninsula Hospital. Since the closure of the hospital there has been no activity on the property. The immediate area surrounding the project site is predominantly developed with residential buildings. Generally to the east most of the development is in low-rise houses. The NYCHA Ocean Bay Apartments (7- and 9-story buildings) are located directly north and west of the project site. Further to the west there are a few taller residential buildings. But, the majority of buildings are also low-rise residential buildings. The elevated subway track is located to the south with the beach and waterfront directly beyond it. There are semi-industrial storage uses located along Rockaway Beach Boulevard but, not many retail or commercial uses in the area;

QUEENS BOROUGH PRESIDENT RECOMMENDATION ULURP #190251 MMQ

Page two

- Community Board 14 (CB 14) disapproved the application with conditions by a vote of twenty-eight (28) against with five (5) in favor and one (1) abstaining at a public hearing held on June 25, 2019. CB 14's conditions were as follows: reduction of total number of units by 30% to 50%; residential parking should be increased up to 75% of units; there is an existing need for school seats a new K-8 school should be built immediately for existing area students and future students; a prevailing wage agreement for all workers should be in place; in consultation with CB 14 a holistic transportation plan for the entire peninsula should be done to mitigate existing issues and impacts from the project; the final AMI mix should be discussed with CB 14 to determine the need and what is appropriate. Senior affordable housing has strong CB 14 support; there should be 5 month progress reports by Arker to CB 14. There should be a website that provides project updates and project related job opportunities; development should be capped at 12-stories; HPD property at B48th to B 51st Streets should be deeded to Department of Parks. Arker should fund planning, construction, and maintenance of a community park with a park house, multipurpose field and running track, children's playground with water features; there should be a signed Community Benefits Agreement including jobs for local residents; minimum 50% of units should be set aside for CB 14 residents;
- The applicant at the Borough President's public hearing explained that project is intended to create new housing opportunities at varying income levels for the neighborhood, bring new economic activity into the area that would provide employment, goods and services for everyone in Rockaway. Several of those testifying in favor of the project spoke about the need for housing, jobs, goods and services that the project would make available for everyone and in particular for returning recent college graduates. Many of the people who testified against the project said there was need for increased economic activity and jobs for the neighborhood. However, there were also concerns about the number of housing units proposed, the proposed height of the buildings, and how the area is now impacted by the overburdened infrastructure and limited roadways that would be exacerbated by the increased numbers of people and vehicles drawn to the area by the proposed project and especially during a major storm emergency. The imbalance between overutilized and underutilized schools in the project area was also mentioned by some of the speakers.

RECOMMENDATION

There is a broad agreement that the former Peninsula Hospital site should be redeveloped to bring new housing and economic development into the area that needs jobs, goods and services. The differences and objections voiced by CB 14 and others in the community are about the size, density and scale of the proposed project compared to the built character of the area as it currently exists and potential negative impacts. This project should be the beginning of the revitalization of this area. But, it must be implemented with sensitivity to the existing area residents and in a manner to assure success of the project and improvements for the community.

Based on the above consideration, I hereby recommend approval of the project with the following conditions:

- There is a need for housing in the area. However, the need for housing in the area should not be met all
 on one site. CB 14 wanted a 30% to 50% reduction of units. The number of units should be reduced from
 the proposed 2200 units to 1600 units of housing. That reduction would bring the project closer to the
 density of the nearby Ocean Bay Apartments;
- Provision of more moderate- and medium-income units for seniors, working families and individuals who
 wish to stay in the community. CB 14 has requested further consultation on the AMI mix of the project.
 Mixed income neighborhoods are important to sustain local businesses that bring jobs, goods and services
 to the neighborhood;
- As proposed the project's tallest buildings would be more than double the height of other buildings in the immediate area. CB 14 has requested a 12-story cap on building heights. The heights of the tallest buildings should be lowered to be more in character with the heights of area buildings;
- CB 14 had requested provision of 75% parking for proposed units. The applicant should stay in contact
 with CB 14 and community groups and within a year of approval there should be a reassessment to
 determine the adequacy of the proposed parking and whether there are any opportunities to find more
 spaces to meet any unanticipated need. In any case, even with a reduction of units the proposed 973
 parking spaces must be maintained;
- The developer should meet with CB 14 every five months for project updates and an opportunity to discuss the project as it moves forward;
- Continued efforts to bring Article 28 emergency medical facilities into the area. There was testimony at the
 public hearing that timeliness of treatment is a critical element to survive cardiac emergencies and that
 there were no emergency cardiac treatment centers on the Rockaway Peninsula. Cardiac and other Article
 28 critical emergency treatment facilities must be made available on the Rockaway Peninsula;
- Good schools are very important to building communities. Pre-K seats should be provided onsite. CB 14
 has requested a new K to 8 school to serve the project and surrounding area. The Department of
 Education (DOE) must dedicate the resources necessary to improve the local schools immediately to serve
 the students living in the district in the vicinity of the project. Going forward DOE must account for the
 forecasted number of students and have funding and plans in place to have seats ready in a new school to
 accommodate new students.

PRESIDENT, BOROUGH OF QUEENS

8/8/2019

Queens Borough President Recommendation

APPLICATION: ULURP #190352 ZMQ

COMMUNITY BOARD: Q14

DOCKET DESCRIPTION

IN THE MATTER OF an application submitted by Akerman LLP on behalf of the Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 20-1 of the NYC Charter for the amendment of the Zoning Map Section No. 30c:

- eliminating from within an existing R5 District a C1-2 District bounded by a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, and the easterly street line of former Beach 51st Street;
- changing from an R5 District to a C4-4 District property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach boulevard, and Beach 53rd Street; and
- changing from an C8-1 District to a C4-3A District property bounded by Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, and Beach 52nd Street:

Borough of Queens, Community District 14, as shown on a diagram (for illustrative purposes only) dated May 6, 2019, and subject to the conditions of CEQR Declaration E-532. (Related ULURPs # 190251 MMQ, N190364 ZRQ, 190366 ZSQ, 190375 ZSQ)

PUBLIC HEARING

A Public Hearing was held in the Borough President's Conference Room at 120-55 Queens Boulevard on Thursday, July 11, 2019 at 10:30 A.M. pursuant to Section 82(5) of the New York City Charter and was duly advertised in the manner specified in Section 197-c (i) of the New York City Charter. The applicant made a presentation. There were ten (10) speakers in favor with five (5) speakers against. The hearing was closed.

CONSIDERATION

Subsequent to a review of the application and consideration of testimony received at the public hearing, the following issues and impacts have been identified:

- The applicant is proposing to rezone the project site between Beach Channel Drive and Rockaway Beach Boulevard generally between Beach 53rd Street to Beach 51st Street and 50th Street from R5 with a C1-2 overlay to a C4-4 district, and a site south across Rockaway Beach Boulevard and Beach 52rd Street from C8-1 to a C4-3A district. The proposed rezoning would facilitate development of a Large Scale General Development (LSGD);
- The applicant has also concurrently filed other applications to facilitate development of the proposed LSGD. The other applications are a mapping action to reopen a portion of Beach 52nd Street, a zoning text amendment to Appendix F of the NYC Zoning Resolution establishing and designating the area to be rezoned as a Mandatory Inclusionary Housing Area, a Special Permit allowing modification of yard, height and setback requirements and to allow a Physical Culture Establishment within the LSGD, and a Special Permit to modify signage regulations for potential commercial uses in the LSGD;
- The proposed LSGD would total 2,371,000 gross square feet built in 11 buildings with heights varying between 8- to 19-stories over six sub-sections of the 410,052.28 sf site. There would be a total of 2,200 residential units. There would be 1,927 units for incomes up to 80% AMI including 201 units of Affordable Independent Residences for Seniors (AIRS) and 273 units of moderate income housing. A 20,000 sf supermarket would be built in Phase I of the project as part of the 72,000 sf of retail space proposed in the project. 77,000 sf of community space is proposed that would allow schools, day care or medical care facilities, or cultural space. Parking would be provided in 973 accessory parking spaces located in parking lots and garages distributed throughout the development area with 754 residential spaces, 144 retail spaces and 75 spaces for community facility/ medical office uses. A proposed privately owned, open internal roadway network with two 60 feet wide streets (north-south extension of Beach 52nd Street, and an east-west Peninsula Way between Beach 50th & Beach 53rd Streets) that would create four quadrants on the site. 24,000 sf of publicly accessible open space would be distributed throughout the site. Two major components would be an open space identified as Pedestrian Plaza (Beach 51st Street) and Highpoint Intersection (intersection of Beach 52nd Street & Peninsula Way). Both would be resilient gathering points for pedestrians with planters and seating. The Beach 51st Street open space would have a children's play area, rubber play surfaces, and a pedestrian plaza with publicly accessible amenities. The phased build out is projected to be 10 to 15 years;
- The project site is the former site of Peninsula Hospital. Since the closure of the hospital there has been no activity on the property. The immediate area surrounding the project site is predominantly developed with residential buildings. Generally to the east most of the development is in low-rise houses. The NYCHA Ocean Bay Apartments (7- and 9-story buildings) are located directly north and west of the project site. Further to the west there are a few taller residential buildings. But, the majority of buildings are also low-rise residential buildings. The elevated subway track is located to the south with the beach and waterfront directly beyond it. There are semi-industrial storage uses located along Rockaway Beach Boulevard but, not many retail or commercial uses in the area;

QUEENS BOROUGH PRESIDENT RECOMMENDATION ULURP #190352 ZMQ

Page two

- Community Board 14 (CB 14) disapproved the application with conditions by a vote of twenty-eight (28) against with five (5) in favor and one (1) abstaining at a public hearing held on June 25, 2019. CB 14's conditions were as follows: reduction of total number of units by 30% to 50%; residential parking should be increased up to 75% of units; there is an existing need for school seats a new K-8 school should be built immediately for existing area students and future students; a prevailing wage agreement for all workers should be in place; in consultation with CB 14 a holistic transportation plan for the entire peninsula should be done to mitigate existing issues and impacts from the project; the final AMI mix should be discussed with CB 14 to determine the need and what is appropriate. Senior affordable housing has strong CB 14 support; there should be 5 month progress reports by Arker to CB 14. There should be a website that provides project updates and project related job opportunities; development should be capped at 12-stories; HPD property at B 48th to B 51st Streets should be deeded to Department of Parks. Arker should fund planning, construction, and maintenance of a community park with a park house, multipurpose field and running track, children's playground with water features; there should be a signed Community Benefits Agreement including jobs for local residents; minimum 50% of units should be set aside for CB 14 residents;
- The applicant at the Borough President's public hearing explained that project is intended to create new housing opportunities at varying income levels for the neighborhood, bring new economic activity into the area that would provide employment, goods and services for everyone in Rockaway. Several of those testifying in favor of the project spoke about the need for housing, jobs, goods and services that the project would make available for everyone and in particular for returning recent college graduates. Many of the people who testified against the project said there was need for increased economic activity and jobs for the neighborhood. However, there were also concerns about the number of housing units proposed, the proposed height of the buildings, and how the area is now impacted by the overburdened infrastructure and limited roadways that would be exacerbated by the increased numbers of people and vehicles drawn to the area by the proposed project and especially during a major storm emergency. The imbalance between overutilized and underutilized schools in the project area was also mentioned by some of the speakers.

RECOMMENDATION

There is a broad agreement that the former Peninsula Hospital site should be redeveloped to bring new housing and economic development into the area that needs jobs, goods and services. The differences and objections voiced by CB 14 and others in the community are about the size, density and scale of the proposed project compared to the built character of the area as it currently exists and potential negative impacts. This project should be the beginning of the revitalization of this area. But, it must be implemented with sensitivity to the existing area residents and in a manner to assure success of the project and improvements for the community.

Based on the above consideration, I hereby recommend approval of the project with the following conditions:

- There is a need for housing in the area. However, the need for housing in the area should not be met all
 on one site. CB 14 wanted a 30% to 50% reduction of units. The number of units should be reduced from
 the proposed 2200 units to 1600 units of housing. That reduction would bring the project closer to the
 density of the nearby Ocean Bay Apartments;
- Provision of more moderate- and medium-income units for seniors, working families and individuals who
 wish to stay in the community. CB 14 has requested further consultation on the AMI mix of the project.
 Mixed income neighborhoods are important to sustain local businesses that bring jobs, goods and services
 to the neighborhood;
- As proposed the project's tallest buildings would be more than double the height of other buildings in the immediate area. CB 14 has requested a 12-story cap on building heights. The heights of the tallest buildings should be lowered to be more in character with the heights of area buildings;
- CB 14 had requested provision of 75% parking for proposed units. The applicant should stay in contact
 with CB 14 and community groups and within a year of approval there should be a reassessment to
 determine the adequacy of the proposed parking and whether there are any opportunities to find more
 spaces to meet any unanticipated need. In any case, even with a reduction of units the proposed 973
 parking spaces must be maintained;
- The developer should meet with CB 14 every five months for project updates and an opportunity to discuss the project as it moves forward;
- Continued efforts to bring Article 28 emergency medical facilities into the area. There was testimony at the
 public hearing that timeliness of treatment is a critical element to survive cardiac emergencies and that
 there were no emergency cardiac treatment centers on the Rockaway Peninsula. Cardiac and other Article
 28 critical emergency treatment facilities must be made available on the Rockaway Peninsula;
- Good schools are very important to building communities. Pre-K seats should be provided onsite. CB 14
 has requested a new K to 8 school to serve the project and surrounding area. The Department of
 Education (DOE) must dedicate the resources necessary to improve the local schools immediately to serve
 the students living in the district in the vicinity of the project. Going forward DOE must account for the
 forecasted number of students and have funding and plans in place to have seats ready in a new school to
 accommodate new students.

Melinde at 8 8 2019
PRESIDENT, BOROUGH OF QUEENS DATE

Queens Borough President Recommendation

APPLICATION: ULURP #190364 ZRQ

COMMUNITY BOARD: Q14

DOCKET DESCRIPTION

IN THE MATTER OF an application submitted by Akerman LLP on behalf of the Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the New York City Charter for zoning text amendments:

- modifying Appendix F of the Zoning Resolution designating the Project Area as a Mandatory Inclusionary Housing ("MIH") area, Borough of Queens, Community District 14, as shown on a diagram (for illustrative purposes only) dated May 6, 2019, and subject to the conditions of CEQR Declaration E-532; and,
- modifying Zoning Resolution Section 74-744(a) to allow a Physical Culture Establishment (PCE) as-ofright within the Large Scale General Development.

(Related ULURPs #190251 MMQ, 190352 ZMQ, 190366 ZSQ, 190375 ZSQ)

PUBLIC HEARING

A Public Hearing was held in the Borough President's Conference Room at 120-55 Queens Boulevard on Thursday, July 11, 2019 at 10:30 A.M. pursuant to Section 82(5) of the New York City Charter and was duly advertised in the manner specified in Section 197-c (i) of the New York City Charter. The applicant made a presentation. There were ten (10) speakers in favor with five (5) speakers against. The hearing was closed.

CONSIDERATION

Subsequent to a review of the application and consideration of testimony received at the public hearing, the following issues and impacts have been identified:

- The applicant is proposing these zoning text amendments to facilitate development of a Large Scale General Development (LSGD). The designation and establishment of the area to be rezoned as a Mandatory Inclusionary Housing Area would assure that it remains an affordable housing into perpetuity. The as-of-right allowance of a PCE;
- The applicant has also concurrently filed other applications to facilitate development of the proposed LSGD. The other applications are for a rezoning to C4-4 and C4-3A, a mapping action to reopen a portion of Beach 52nd Street, a Special Permit allowing modification of yard, height and setback requirements, and a Special Permit to modify signage regulations for potential commercial uses in the LSGD;
- The proposed LSGD would total 2,371,000 gross square feet built in 11 buildings with heights varying between 8- to 19-stories over six sub-sections of the 410,052.28 sf site. There would be a total of 2,200 residential units. There would be 1,927 units for incomes up to 80% AMI including 201 units of Affordable Independent Residences for Seniors (AIRS) and 273 units of moderate income housing. A 20,000 sf supermarket would be built in Phase I of the project as part of the 72,000 sf of retail space proposed in the project. 77,000 sf of community space is proposed that would allow schools, day care or medical care facilities, or cultural space. Parking would be provided in 973 accessory parking spaces located in parking lots and garages distributed throughout the development area with 754 residential spaces, 144 retail spaces and 75 spaces for community facility/ medical office uses. A proposed privately owned, open internal roadway network with two 60 feet wide streets (north-south extension of Beach 52nd Street, and an east-west Peninsula Way between Beach 50th & Beach 53rd Streets) that would create four quadrants on the site. 24,000 sf of publicly accessible open space would be distributed throughout the site. Two major components would be an open space identified as Pedestrian Plaza (Beach 51st Street) and Highpoint Intersection (intersection of Beach 52nd Street & Peninsula Way). Both would be resilient gathering points for pedestrians with planters and seating. The Beach 51st Street open space would have a children's play area, rubber play surfaces, and a pedestrian plaza with publicly accessible amenities. The phased build out is projected to be 10 to 15 years;
- The project site is the former site of Peninsula Hospital. Since the closure of the hospital there has been no activity on the property. The immediate area surrounding the project site is predominantly developed with residential buildings. Generally to the east most of the development is in low-rise houses. The NYCHA Ocean Bay Apartments (7- and 9-story buildings) are located directly north and west of the project site. Further to the west there are a few taller residential buildings. But, the majority of buildings are also low-rise residential buildings. The elevated subway track is located to the south with the beach and waterfront directly beyond it. There are semi-industrial storage uses located along Rockaway Beach Boulevard but, not many retail or commercial uses in the area;

- Community Board 14 (CB 14) disapproved the application with conditions by a vote of twenty-eight (28) against with five (5) in favor and one (1) abstaining at a public hearing held on June 25, 2019. CB 14's conditions were as follows: reduction of total number of units by 30% to 50%; residential parking should be increased up to 75% of units; there is an existing need for school seats a new K-8 school should be built immediately for existing area students and future students; a prevailing wage agreement for all workers should be in place; in consultation with CB 14 a holistic transportation plan for the entire peninsula should be done to mitigate existing issues and impacts from the project; the final AMI mix should be discussed with CB 14 to determine the need and what is appropriate. Senior affordable housing has strong CB 14 support; there should be 5 month progress reports by Arker to CB 14. There should be a website that provides project updates and project related job opportunities; development should be capped at 12-stories; HPD property at B48th to B 51st Streets should be deeded to Department of Parks. Arker should fund planning, construction, and maintenance of a community park with a park house, multipurpose field and running track, children's playground with water features; there should be set aside for CB 14 residents;
- The applicant at the Borough President's public hearing explained that project is intended to create new housing opportunities at varying income levels for the neighborhood, bring new economic activity into the area that would provide employment, goods and services for everyone in Rockaway. Several of those testifying in favor of the project spoke about the need for housing, jobs, goods and services that the project would make available for everyone and in particular for returning recent college graduates. Many of the people who testified against the project said there was need for increased economic activity and jobs for the neighborhood. However, there were also concerns about the number of housing units proposed, the proposed height of the buildings, and how the area is now impacted by the overburdened infrastructure and limited roadways that would be exacerbated by the increased numbers of people and vehicles drawn to the area by the proposed project and especially during a major storm emergency. The imbalance between overutilized and underutilized schools in the project area was also mentioned by some of the speakers.

RECOMMENDATION

There is a broad agreement that the former Peninsula Hospital site should be redeveloped to bring new housing and economic development into the area that needs jobs, goods and services. The differences and objections voiced by CB 14 and others in the community are about the size, density and scale of the proposed project compared to the built character of the area as it currently exists and potential negative impacts. This project should be the beginning of the revitalization of this area. But, it must be implemented with sensitivity to the existing area residents and in a manner to assure success of the project and improvements for the community.

Based on the above consideration, I hereby recommend approval of the project with the following conditions:

- There is a need for housing in the area. However, the need for housing in the area should not be met all
 on one site. CB 14 wanted a 30% to 50% reduction of units. The number of units should be reduced from
 the proposed 2200 units to 1600 units of housing. That reduction would bring the project closer to the
 density of the nearby Ocean Bay Apartments;
- Provision of more moderate- and medium-income units for seniors, working families and individuals who
 wish to stay in the community. CB 14 has requested further consultation on the AMI mix of the project.
 Mixed income neighborhoods are important to sustain local businesses that bring jobs, goods and services
 to the neighborhood;
- As proposed the project's tallest buildings would be more than double the height of other buildings in the immediate area. CB 14 has requested a 12-story cap on building heights. The heights of the tallest buildings should be lowered to be more in character with the heights of area buildings;
- CB 14 had requested provision of 75% parking for proposed units. The applicant should stay in contact
 with CB 14 and community groups and within a year of approval there should be a reassessment to
 determine the adequacy of the proposed parking and whether there are any opportunities to find more
 spaces to meet any unanticipated need. In any case, even with a reduction of units the proposed 973
 parking spaces must be maintained;
- The developer should meet with CB 14 every five months for project updates and an opportunity to discuss the project as it moves forward;
- Continued efforts to bring Article 28 emergency medical facilities into the area. There was testimony at the
 public hearing that timeliness of treatment is a critical element to survive cardiac emergencies and that
 there were no emergency cardiac treatment centers on the Rockaway Peninsula. Cardiac and other Article
 28 critical emergency treatment facilities must be made available on the Rockaway Peninsula;
- Good schools are very important to building communities. Pre-K seats should be provided onsite. CB 14 has requested a new K to 8 school to serve the project and surrounding area. The Department of Education (DOE) must dedicate the resources necessary to improve the local schools immediately to serve the students living in the district in the vicinity of the project. Going forward DOE must account for the forecasted number of students and have funding and plans in place to have seats ready in a new school to accommodate new students.

Melinde at 8 2019
PRESIDENT, BOROUGH OF QUEENS
DATE

Queens Borough President Recommendation

APPLICATION: ULURP #190366 ZSQ COMMUNITY BOARD: Q14

DOCKET DESCRIPTION

IN THE MATTER OF an application submitted by Akerman LLP on behalf of the Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the NYC Charter for the grant of a Special Permit pursuant to Section 74-743(a)(2) of the NYC Zoning Resolution to modify:

- the rear yard requirements of Section 23-533 (Required rear yard equivalents for Quality Housing Buildings) and Section 35-53 (Modification of Rear yard Requirements);
- 2. the side yard requirements of Section 35-54 (Special Provisions Applying to R1 through R5 Districts); and
- the height and setback requirements of Section 23-664 (Modified height and setback regulations for certain Inclusionary Housing buildings or affordable independent residences for seniors) and Section 35-654 (Modified height and setback regulations for certain Inclusionary housing buildings or affordable independent residences for seniors);

in connection with a proposed mixed use development, within a large-scale general development, on property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, Beach 52nd Street, Rockaway Beach Boulevard and Beach 53nd Street (Block 15842 Lot 1 & p/o Lot 100, Block 15843 Lot 1, and Block 15857 Lot 1 & p/o Lot 7), in a C4-4* and C4-3A* Districts, Borough of Queens, Community District 14.

*Note: The site is proposed to be rezoned by eliminating C1-2 District within an existing R5 district and by changing an existing R5 and C8-1 Districts to C4-4 and C4-3A Districts under a concurrent application for a Zoning Map change (C190352 ZMQ)

(Related ULURPs #190251 MMQ, 190352 ZMQ, N190364 ZRQ, 190375 ZSQ)

PUBLIC HEARING

A Public Hearing was held in the Borough President's Conference Room at 120-55 Queens Boulevard on Thursday, July 11, 2019 at 10:30 A.M. pursuant to Section 82(5) of the New York City Charter and was duly advertised in the manner specified in Section 197-c (i) of the New York City Charter. The applicant made a presentation. There were ten (10) speakers in favor with five (5) speakers against. The hearing was closed.

CONSIDERATION

Subsequent to a review of the application and consideration of testimony received at the public hearing, the following issues and impacts have been identified:

- The applicant is requesting a Special Permit to allow a portion of a parking facility to abut an area on a neighboring property used for parking and vehicle access, instead of a required rear yard equivalent a portion of a proposed private open space and street network, and waivers of maximum base height and setback requirements to facilitate development of a proposed Large Scale General Development (LSGD);
- The applicant has also concurrently filed other applications to facilitate development of the proposed LSGD. The other applications are for a rezoning to C4-4 and C4-3A, a mapping action to reopen a portion of Beach 52nd Street, a zoning text amendment to Appendix F of the NYC Zoning Resolution establishing and designating the area to be rezoned as a Mandatory Inclusionary Housing Area and to allow a Physical Culture Establishment within the LSGD, and a Special Permit to modify signage regulations for potential commercial uses in the LSGD;
- The proposed LSGD would total 2,371,000 gross square feet built in 11 buildings with heights varying between 8- to 19-stories over six sub-sections of the 410,052.28 sf site. There would be a total of 2,200 residential units. There would be 1,927 units for incomes up to 80% AMI including 201 units of Affordable Independent Residences for Seniors (AIRS) and 273 units of moderate income housing. A 20,000 sf supermarket would be built in Phase I of the project as part of the 72,000 sf of retail space proposed in the project. 77,000 sf of community space is proposed that would allow schools, day care or medical care facilities, or cultural space. Parking would be provided in 973 accessory parking spaces located in parking lots and garages distributed throughout the development area with 754 residential spaces, 144 retail spaces and 75 spaces for community facility/ medical office uses. A proposed privately owned, open internal roadway network with two 60 feet wide streets (north-south extension of Beach 52nd Street, and an east-west Peninsula Way between Beach 50th & Beach 53rd Streets) that would create four quadrants on the site. 24,000 sf of publicly accessible open space would be distributed throughout the site. Two major components would be an open space identified as Pedestrian Plaza (Beach 51st Street) and Highpoint Intersection (intersection of Beach 52nd Street & Peninsula Way). Both would be resilient gathering points for pedestrians with planters and seating. The Beach 51st Street open space would have a children's play area, rubber play surfaces, and a pedestrian plaza with publicly accessible amenities. The phased build out is projected to be 10 to 15 years;
- The project site is the former site of Peninsula Hospital. Since the closure of the hospital there has been no activity on the property. The immediate area surrounding the project site is predominantly developed with residential buildings. Generally to the east most of the development is in low-rise houses. The NYCHA Ocean Bay Apartments (7- and 9-story buildings) are located directly north and west of the project site. Further to the west there are a few taller residential buildings. But, the majority of buildings are also low-rise residential buildings. The elevated subway track is located to the south with the beach and

QUEENS BOROUGH PRESIDENT RECOMMENDATION ULURP #190366 ZSQ

Page two

waterfront directly beyond it. There are semi-industrial storage uses located along Rockaway Beach Boulevard but, not many retail or commercial uses in the area;

- Ocommunity Board 14 (CB 14) disapproved the application with conditions by a vote of twenty-eight (28) against with five (5) in favor and one (1) abstaining at a public hearing held on June 25, 2019. CB 14's conditions were as follows: reduction of total number of units by 30% to 50%; residential parking should be increased up to 75% of units; there is an existing need for school seats a new K-8 school should be built immediately for existing area students and future students; a prevailing wage agreement for all workers should be in place; in consultation with CB 14 a holistic transportation plan for the entire peninsula should be done to mitigate existing issues and impacts from the project; the final AMI mix should be discussed with CB 14 to determine the need and what is appropriate. Senior affordable housing has strong CB 14 support; there should be 5 month progress reports by Arker to CB 14. There should be a website that provides project updates and project related job opportunities; development should be capped at 12-stories; HPD property at B48th to B 51st Streets should be deeded to Department of Parks. Arker should fund planning, construction, and maintenance of a community park with a park house, multipurpose field and running track, children's playground with water features; there should be set aside for CB 14 residents;
- The applicant at the Borough President's public hearing explained that project is intended to create new housing opportunities at varying income levels for the neighborhood, bring new economic activity into the area that would provide employment, goods and services for everyone in Rockaway. Several of those testifying in favor of the project spoke about the need for housing, jobs, goods and services that the project would make available for everyone and in particular for returning recent college graduates. Many of the people who testified against the project said there was need for increased economic activity and jobs for the neighborhood. However, there were also concerns about the number of housing units proposed, the proposed height of the buildings, and how the area is now impacted by the overburdened infrastructure and limited roadways that would be exacerbated by the increased numbers of people and vehicles drawn to the area by the proposed project and especially during a major storm emergency. The imbalance between overutilized and underutilized schools in the project area was also mentioned by some of the speakers.

RECOMMENDATION

There is a broad agreement that the former Peninsula Hospital site should be redeveloped to bring new housing and economic development into the area that needs jobs, goods and services. The differences and objections voiced by CB 14 and others in the community are about the size, density and scale of the proposed project compared to the built character of the area as it currently exists and potential negative impacts. This project should be the beginning of the revitalization of this area. But, it must be implemented with sensitivity to the existing area residents and in a manner to assure success of the project and improvements for the community.

Based on the above consideration, I hereby recommend approval of the project with the following conditions:

- There is a need for housing in the area. However, the need for housing in the area should not be met all
 on one site. CB 14 wanted a 30% to 50% reduction of units. The number of units should be reduced from
 the proposed 2200 units to 1600 units of housing. That reduction would bring the project closer to the
 density of the nearby Ocean Bay Apartments;
- Provision of more moderate- and medium-income units for seniors, working families and individuals who
 wish to stay in the community. CB 14 has requested further consultation on the AMI mix of the project.
 Mixed income neighborhoods are important to sustain local businesses that bring jobs, goods and services
 to the neighborhood:
- As proposed the project's tallest buildings would be more than double the height of other buildings in the immediate area. CB 14 has requested a 12-story cap on building heights. The heights of the tallest buildings should be lowered to be more in character with the heights of area buildings;
- CB 14 had requested provision of 75% parking for proposed units. The applicant should stay in contact
 with CB 14 and community groups and within a year of approval there should be a reassessment to
 determine the adequacy of the proposed parking and whether there are any opportunities to find more
 spaces to meet any unanticipated need. In any case, even with a reduction of units the proposed 973
 parking spaces must be maintained;
- The developer should meet with CB 14 every five months for project updates and an opportunity to discuss the project as it moves forward;
- Continued efforts to bring Article 28 emergency medical facilities into the area. There was testimony at the
 public hearing that timeliness of treatment is a critical element to survive cardiac emergencies and that
 there were no emergency cardiac treatment centers on the Rockaway Peninsula. Cardiac and other Article
 28 critical emergency treatment facilities must be made available on the Rockaway Peninsula;
- Good schools are very important to building communities. Pre-K seats should be provided onsite. CB 14 has requested a new K to 8 school to serve the project and surrounding area. The Department of Education (DOE) must dedicate the resources necessary to improve the local schools immediately to serve the students living in the district in the vicinity of the project. Going forward DOE must account for the forecasted number of students and have funding and plans in place to have seats ready in a new school to accommodate new students.

Melinde la 2019
PRESIDENT, BOROUGH OF QUEENS
DATE

Queens Borough President Recommendation

APPLICATION: ULURP #190375 ZSQ

COMMUNITY BOARD: Q14

DOCKET DESCRIPTION

IN THE MATTER OF an application submitted by Akerman LLP on behalf of the Peninsula Rockaway Limited Partnership pursuant to Sections 197-c and 201 of the NYC Charter for the grant of a Special Permit pursuant to Section 74-744(c)(1) of the NYC Zoning Resolution to modify the surface area requirements of Section 32-64 (Surface Area and Illumination Provisions), in connection with a proposed mixed use development, within a large-scale general development, on property bounded by Beach Channel Drive, the westerly street line of former Beach 51st Street, a line 420 feet southerly of Beach Channel Drive, Beach 50th Street, Rockaway Beach Boulevard, a line 100 feet easterly of Beach 52nd Street, a line 85 feet northerly of Shore Front Parkway, Beach 52nd Street, Rockaway Beach Boulevard and Beach 53nd Street (Block 15842 Lot 1 & p/o Lot 100, Block 15843 Lot 1, and Block 15857 Lot 1 & p/o Lot 7), in a C4-4* and C4-3A* Districts, Borough of Queens, Community District 14.

*Note: The site is proposed to be rezoned by eliminating C1-2 District within an existing R5 district and by changing an existing R5 and C8-1 Districts to C4-4 and C4-3A Districts under a concurrent application for a Zoning Map change (C190352 ZMQ)

(Related ULURPs #190251 MMQ, 190352 ZMQ, N190364 ZRQ, 190366 ZSQ)

PUBLIC HEARING

A Public Hearing was held in the Borough President's Conference Room at 120-55 Queens Boulevard on Thursday, July 11, 2019 at 10:30 A.M. pursuant to Section 82(5) of the New York City Charter and was duly advertised in the manner specified in Section 197-c (i) of the New York City Charter. The applicant made a presentation. There were ten (10) speakers in favor with five (5) speakers against. The hearing was closed.

CONSIDERATION

Subsequent to a review of the application and consideration of testimony received at the public hearing, the following issues and impacts have been identified:

- The applicant is requesting a Special Permit that allow modification of signage regulations for potential commercial uses in the LSGD The proposed Special Permit would facilitate development of a Large Scale General Development (LSGD);
- The applicant has also concurrently filed other applications to facilitate development of the proposed LSGD. The other applications are for a rezoning to C4-4 and C4-3A, a mapping action to reopen a portion of Beach 52nd Street, a zoning text amendment to Appendix F of the NYC Zoning Resolution establishing and designating the area to be rezoned as a Mandatory Inclusionary Housing Area and to allow a Physical Culture Establishment within the LSGD, and a Special Permit allowing modification of yard, height and setback requirements;
- The proposed LSGD would total 2,371,000 gross square feet built in 11 buildings with heights varying between 8- to 19-stories over six sub-sections of the 410,052.28 sf site. There would be a total of 2,200 residential units. There would be 1,927 units for incomes up to 80% AMI including 201 units of Affordable Independent Residences for Seniors (AIRS) and 273 units of moderate income housing. A 20,000 sf supermarket would be built in Phase I of the project as part of the 72,000 sf of retail space proposed in the project. 77,000 sf of community space is proposed that would allow schools, day care or medical care facilities, or cultural space. Parking would be provided in 973 accessory parking spaces located in parking lots and garages distributed throughout the development area with 754 residential spaces, 144 retail spaces and 75 spaces for community facility/ medical office uses. A proposed privately owned, open internal roadway network with two 60 feet wide streets (north-south extension of Beach 52nd Street, and an east-west Peninsula Way between Beach 50th & Beach 53rd Streets) that would create four quadrants on the site. 24,000 sf of publicly accessible open space would be distributed throughout the site. Two major components would be an open space identified as Pedestrian Plaza (Beach 51st Street) and Highpoint Intersection (intersection of Beach 52nd Street & Peninsula Way). Both would be resilient gathering points for pedestrians with planters and seating. The Beach 51st Street open space would have a children's play area, rubber play surfaces, and a pedestrian plaza with publicly accessible amenities. The phased build out is projected to be 10 to 15 years;
- The project site is the former site of Peninsula Hospital. Since the closure of the hospital there has been no activity on the property. The immediate area surrounding the project site is predominantly developed with residential buildings. Generally to the east most of the development is in low-rise houses. The NYCHA Ocean Bay Apartments (7- and 9-story buildings) are located directly north and west of the project site. Further to the west there are a few taller residential buildings. But, the majority of buildings are also low-rise residential buildings. The elevated subway track is located to the south with the beach and waterfront directly beyond it. There are semi-industrial storage uses located along Rockaway Beach Boulevard but, not many retail or commercial uses in the area;

QUEENS BOROUGH PRESIDENT RECOMMENDATION ULURP #190375 ZSQ

Page two

- Community Board 14 (CB 14) disapproved the application with conditions by a vote of twenty-eight (28) against with five (5) in favor and one (1) abstaining at a public hearing held on June 25, 2019. CB 14's conditions were as follows: reduction of total number of units by 30% to 50%; residential parking should be increased up to 75% of units; there is an existing need for school seats a new K-8 school should be built immediately for existing area students and future students; a prevailing wage agreement for all workers should be in place; in consultation with CB 14 a holistic transportation plan for the entire peninsula should be done to mitigate existing issues and impacts from the project; the final AMI mix should be discussed with CB 14 to determine the need and what is appropriate. Senior affordable housing has strong CB 14 support; there should be 5 month progress reports by Arker to CB 14. There should be a website that provides project updates and project related job opportunities; development should be capped at 12-stories; HPD property at B48th to B 51st Streets should be deeded to Department of Parks. Arker should fund planning, construction, and maintenance of a community park with a park house, multipurpose field and running track, children's playground with water features; there should be set aside for CB 14 residents;
- The applicant at the Borough President's public hearing explained that project is intended to create new housing opportunities at varying income levels for the neighborhood, bring new economic activity into the area that would provide employment, goods and services for everyone in Rockaway. Several of those testifying in favor of the project spoke about the need for housing, jobs, goods and services that the project would make available for everyone and in particular for returning recent college graduates. Many of the people who testified against the project said there was need for increased economic activity and jobs for the neighborhood. However, there were also concerns about the number of housing units proposed, the proposed height of the buildings, and how the area is now impacted by the overburdened infrastructure and limited roadways that would be exacerbated by the increased numbers of people and vehicles drawn to the area by the proposed project and especially during a major storm emergency. The imbalance between overutilized and underutilized schools in the project area was also mentioned by some of the speakers.

RECOMMENDATION

There is a broad agreement that the former Peninsula Hospital site should be redeveloped to bring new housing and economic development into the area that needs jobs, goods and services. The differences and objections voiced by CB 14 and others in the community are about the size, density and scale of the proposed project compared to the built character of the area as it currently exists and potential negative impacts. This project should be the beginning of the revitalization of this area. But, it must be implemented with sensitivity to the existing area residents and in a manner to assure success of the project and improvements for the community.

Based on the above consideration, I hereby recommend approval of the project with the following conditions:

- There is a need for housing in the area. However, the need for housing in the area should not be met all
 on one site. CB 14 wanted a 30% to 50% reduction of units. The number of units should be reduced from
 the proposed 2200 units to 1600 units of housing. That reduction would bring the project closer to the
 density of the nearby Ocean Bay Apartments;
- Provision of more moderate- and medium-income units for seniors, working families and individuals who
 wish to stay in the community. CB 14 has requested further consultation on the AMI mix of the project.
 Mixed income neighborhoods are important to sustain local businesses that bring jobs, goods and services
 to the neighborhood;
- As proposed the project's tallest buildings would be more than double the height of other buildings in the immediate area. CB 14 has requested a 12-story cap on building heights. The heights of the tallest buildings should be lowered to be more in character with the heights of area buildings;
- CB 14 had requested provision of 75% parking for proposed units. The applicant should stay in contact
 with CB 14 and community groups and within a year of approval there should be a reassessment to
 determine the adequacy of the proposed parking and whether there are any opportunities to find more
 spaces to meet any unanticipated need. In any case, even with a reduction of units the proposed 973
 parking spaces must be maintained;
- The developer should meet with CB 14 every five months for project updates and an opportunity to discuss the project as it moves forward;
- Continued efforts to bring Article 28 emergency medical facilities into the area. There was testimony at the
 public hearing that timeliness of treatment is a critical element to survive cardiac emergencies and that
 there were no emergency cardiac treatment centers on the Rockaway Peninsula. Cardiac and other Article
 28 critical emergency treatment facilities must be made available on the Rockaway Peninsula;
- Good schools are very important to building communities. Pre-K seats should be provided onsite. CB 14
 has requested a new K to 8 school to serve the project and surrounding area. The Department of
 Education (DOE) must dedicate the resources necessary to improve the local schools immediately to serve
 the students living in the district in the vicinity of the project. Going forward DOE must account for the
 forecasted number of students and have funding and plans in place to have seats ready in a new school to
 accommodate new students.

Melinde 18/8/2019
PRESIDENT, BOROUGH OF QUEENS

DATE